

Check for updates

ADOPTED: 30 June 2022 doi: 10.2903/j.efsa.2022.7445

Welfare of pigs during transport

EFSA Panel on Animal Health and Welfare (AHAW), Søren Saxmose Nielsen, Julio Alvarez, Dominique Joseph Bicout, Paolo Calistri, Elisabetta Canali, Julian Ashley Drewe, Bruno Garin-Bastuji, Jose Luis Gonzales Rojas, Christian Gortázar Schmidt, Virginie Michel, Miguel Ángel Miranda Chueca, Barbara Padalino, Paolo Pasquali, Helen Clare Roberts, Hans Spoolder, Karl Stahl, Antonio Velarde, Arvo Viltrop, Christoph Winckler, Bernadette Earley, Sandra Edwards, Luigi Faucitano, Sonia Marti, Genaro C Miranda de La Lama, Leonardo Nanni Costa, Peter T Thomsen, Sean Ashe, Lina Mur, Yves Van der Stede and Mette Herskin

Abstract

In the framework of its Farm to Fork Strategy, the Commission is undertaking a comprehensive evaluation of the animal welfare legislation. The present Opinion deals with protection of pigs during transport. The welfare of pigs during transport by road is the main focus, but other means of transport are also covered. Current practices related to transport of pigs during the different stages (preparation, loading/unloading, transit and journey breaks) are described. Overall, 10 welfare consequences were identified as highly relevant for the welfare of pigs during transport based on the severity, duration and frequency of occurrence: group stress, handling stress, heat stress, injuries, motion stress, prolonged hunger, prolonged thirst, restriction of movement, resting problems and sensory overstimulation. These welfare consequences and their animal-based measures are described. A variety of hazards were identified, mainly relating to factors such as mixing of unfamiliar pigs, inappropriate handling methods and devices, the use of pick-up pens, inexperienced/untrained handlers, structural deficiencies of vehicles and facilities, poor driving conditions, unfavourable microclimatic and environmental conditions and poor husbandry practices leading to these welfare consequences. The Opinion contains general and specific conclusions relating to the different stages of transport of pigs. Recommendations to prevent hazards and to correct or mitigate welfare consequences are made. Recommendations were also developed to define quantitative thresholds for microclimatic conditions and minimum space allowance within means of transport. The development of the welfare consequences over time was assessed in relation to maximum journey duration. The Opinion covers specific animal transport scenarios identified by the European Commission relating to transport of cull sows and 'special health status animals', and lists welfare concerns associated with these.

© 2022 European Food Safety Authority. *EFSA Journal* published by Wiley-VCH GmbH on behalf of European Food Safety Authority.

Keywords: pig, sow, animal welfare assessment, Farm to Fork Strategy, welfare consequences, animal-based measures, hazards, quantitative thresholds

Requestor: European Commission Question number: EFSA-Q-2021-00434 Correspondence: ahaw@efsa.europa.eu



Panel members: Søren Saxmose Nielsen, Julio Alvarez, Dominique Joseph Bicout, Paolo Calistri, Elisabetta Canali, Julian Ashley Drewe, Bruno Garin-Bastuji, Jose Luis Gonzales Rojas, Christian Gortázar Schmidt, Mette Herskin, Miguel Ángel Miranda Chueca, Virginie Michel, Barbara Padalino, Paolo Pasquali, Helen Clare Roberts, Hans Spoolder, Karl Stahl, Antonio Velarde, Arvo Viltrop and Christoph Winckler.

Waiver: In accordance with Article 21 of the Decision of the Executive Director on Competing Interest Management a waiver was granted to Sandra Edwards, Genaro C Miranda de La Lama and Luigi Faucitano, experts of the Working Group WG/P/AHAW/2020/05 -AHAW Welfare Farm to Fork. Pursuant to Article 21(6) of the aforementioned decision, the concerned experts were allowed to take part in the drafting and in the discussion of the scientific output but were not allowed to take up the role of, or act as, chairman, vice-chairman or rapporteur of the Working Group. Any competing interests are recorded in the respective minutes of the meetings of the Working Group WG/P/AHAW/2020/05 - AHAW Welfare Farm to Fork.

Declarations of interest: If you wish to access the declaration of interests of any expert contributing to an EFSA scientific assessment, please contact interestmanagement@efsa.europa.eu.

Acknowledgements: The Panel wishes to thank the following for the support provided to this scientific output: hearing experts Nancy De Briyne, Michael Cockram, Yolande Seddon and Clive Phillips. EFSA would like to thank Mariana Geffroy, Mimi Kalcheva and Maria Veggeland from EFSA, for all the support provided in this Opinion. EFSA wishes to acknowledge all European competent institutions, Member State bodies and other organisations that provided data for this scientific output.

Suggested citation: EFSA AHAW Panel (EFSA Panel on Animal Health and Welfare), Nielsen SS, Alvarez J, Bicout DJ, Calistri P, Canali E, Drewe JA, Garin-Bastuji B, Gonzales Rojas JL, Gortazar Schmidt C, Michel V, Miranda Chueca MA, Padalino B, Pasquali P, Roberts HC, Spoolder H, Stahl K, Velarde A, Viltrop A, Winckler C, Earley B, Edwards S, Faucitano L, Marti S, Miranda de La Lama GC, Nanni Costa L, Thomsen PT, Ashe S, Mur L, Van der Stede Y and Herskin M, 2022. Scientific Opinion on the welfare of pigs during transport. EFSA Journal 2022;20(9):7445, 108 pp. https://doi.org/10.2903/j.efsa.2022.7445

ISSN: 1831-4732

© 2022 European Food Safety Authority. *EFSA Journal* published by Wiley-VCH GmbH on behalf of European Food Safety Authority.

This is an open access article under the terms of the Creative Commons Attribution-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited and no modifications or adaptations are made.

Reproduction of the images listed below is prohibited and permission must be sought directly from the copyright holder:

Figure 3: © Oxford University Press; Figure 8: © Elsevier; Figure 10, 11 and 13: British Society of Animal Science.



The EFSA Journal is a publication of the European Food Safety Authority, a European agency funded by the European Union.



18314732, 2022, 9. Downoaded from https://efsa.onlinelibrary.wiley.com/doi/10.2903/j.sa.2022.7445 by Universidad De Zaragoza, Wiley Online Library on [3001/2025]. See the Terms and Conditions (https://onlinelibrary.wiley.com/terms-and-conditions) on Wiley Online Library for rules of use; OA articles are governed by the applicable Creative Commons License



Summary

In the framework of its Farm to Fork Strategy, the Commission is undertaking a comprehensive evaluation of the animal welfare legislation, including Council Regulation (EC) No 1/2005¹. The current EU legislation on the protection of animals during transport is based on a Scientific Opinion adopted in 2002. Against this background, the European Commission requested the European Food Safety Authority (EFSA) to give an independent view on the protection of animals during transport for different groups and categories of farmed animals. It also requested EFSA to propose detailed measures to prevent hazards and mitigate the welfare consequences for seven specific scenarios. This Opinion deals with the protection of pigs during transport.

The scientific assessment was carried out by breaking down the transport of pigs into four distinct stages, namely preparation, loading/unloading, transit and journey breaks. For road transport, which is the most common transport practice, each stage was described in terms of current practice and assessed in terms of welfare consequences, animal-based measures (ABMs), and hazards leading to the welfare consequences. In addition, recommendations to prevent hazards and to correct or mitigate welfare consequences were developed. Recommendations were also developed in relation to quantitative thresholds for microclimatic conditions and for spatial thresholds (minimal space allowance) within the means of transport. In addition, the development of welfare consequences over time were assessed in relation to maximum journey duration.

While the Opinion focuses primarily on the road transport of pigs, there are specific sections dealing with air and rail transport. Welfare concerns (defined as an area or a topic to which special attention should be given in order to potentially avoid welfare consequences) related to the transport of cull sows and 'special health status animals' are covered in a separate section of the Opinion.

According to TRACES, an average of ~ 31 million pigs were transported between Member States per year in the period from 2019 to 2021, across all means of transport. This means that pigs are the species, among animals transported as free-moving, that are transported in the highest numbers in the EU by far. Road transport constituted 99% of total pig transport reported in this period.

In total, ten welfare consequences were identified as being highly relevant for the welfare of pigs during transport based on the severity, duration and frequency of occurrence. These were (i) group stress, (ii) handling stress, (iii) heat stress, (iv) injuries, (v) motion stress, (vi) prolonged hunger, (vii) prolonged thirst, (viii) resting problems, (ix) restriction of movement, and (x) sensory overstimulation. The occurrence of each type of welfare consequence varied depending on the stage and means of transport. Pigs may experience one or more negative affective states associated with these welfare consequences, including fear, pain, discomfort, frustration, fatigue and distress. Specific ABMs were identified for each of the highly relevant welfare consequences, including behavioural, clinical and physiological ABMs. A definition and interpretation for each ABM is provided in the Opinion. Some ABMs are relevant to more than one welfare consequence.

A wide variety of hazards were identified for the different welfare consequences and transport stages. These were related to factors such as mixing unfamiliar pigs, inappropriate handling methods and devices, the use of pick-up pens, inexperienced/untrained handlers, structural deficiencies of vehicles and facilities, poor driving and road conditions, unfavourable microclimatic and environmental conditions and poor husbandry practices.

Throughout the scientific literature, it is agreed that ensuring that animals are fit for transport before departure is of utmost importance. However, currently no agreed scientific definition of the concept of fitness for transport exists. In order to avoid doubt and misclassification of animals in relation to fitness for transport, the concept should be properly defined, professional groups (including farmers, stockpersons, drivers, haulers, inspectors and veterinarians) should be well-educated and trained, and questions on responsibility between the groups should be clarified. Also, there are only very few conditions leading animals to be unfit for transport, for which ABMs have been established and validated, including the establishment of thresholds. The main conditions rendering pigs unfit for transport, and methods for assessing fitness for transport, are provided in the Opinion. Guidelines based on ABMs for conditions leading to animals being unfit, including thresholds, should be established and validated.

Periods of pre-transport fasting should be adjusted to the category of pigs and be appropriate for the planned journey duration, and in addition take into consideration whether pigs are transported for slaughter or for further fattening/breeding. Until evidence-based thresholds are established, it is

¹ Regulation (EC) No 1/2005 of the European Council of 22 December 2004 on the protection of animals during transport and related operations and amending Directives 64/432/EEC and 93/119/EC and Regulation (EC) No 1255/9.



recommended to expose finishers being transported to a pre-transit fasting of less than 10 h. The time taken to load should be included within this 10 h period. For other pig categories pre-transit fasting should likely be shorter.

The highly relevant welfare consequences during loading/unloading of pigs are: handling stress, heat stress, and sensory overstimulation. Across the highly relevant welfare consequences, the major hazards are inappropriate handling, unsuitable facilities and high temperatures. A delay in loading and/ or unloading is a hazard for a deterioration of these welfare consequences. The main preventive measures are establishment and maintenance of proper facilities, avoiding loading during hot hours and education and training of handlers.

During the transit stage, pigs will be exposed to a number of hazards, either on their own or in combination, leading to welfare consequences.

As regards microclimatic conditions during road transport of pigs, the upper threshold of the thermal comfort zone and the upper critical temperature of sows were estimated to be 20°C and 22°C dry temperature, respectively. The comparable figures were 22°C and 25°C for finishing pigs, and 25°C and 30°C for weaners of ~ 30 kg. Thus, between pig categories the sensitivity to heat stress differs. In vehicles transporting pigs, the temperature near the pigs should not exceed the upper critical temperature.

In relation to the horizontal space allowance for pigs during road transport, the available evidence suggests that a k-value of at least 0.027 in the allometric equation relating space to liveweight, is required for all categories of pigs (this equates to 0.62 m^2 for a 110-kg finishing pig).

The vertical space in a means of transport is also important for the welfare of pigs. Low vertical space is associated with reduced ventilation, lack of ability to move around and lack of space for natural movements. A proper deck height should allow the prevention of these, in order to avoid welfare consequences such as heat stress and restriction of movement. Research is needed to establish evidence-based thresholds.

The amount of time the animals are exposed to the hazards is dependent on the journey duration. The number and the severity of hazards that animals are exposed to during transport influence the resultant welfare consequences (continuous or semi-continuous, progressive and sporadic). On the basis of evidence on continuous welfare consequences involving stress and negative affective states, for the benefit of animal welfare, the journey duration and frequency, should be kept to a minimum.

To limit the impact of transport on animal welfare, in an effort to reduce the exposure to hazards and related welfare consequences, it is recommended to consider that: motion stress and sensory overstimulation starts as soon as a vehicle starts moving, and continues while the vehicle is moving, potentially leading to fatigue and negative affective states, such as fear and distress; Group stress and resultant injuries (skin lesions) are the result of post-mixing fighting and may start as soon as pigs are mixed and continues during the journey with the severity increasing potentially leading to fatigue; pain and/or discomfort from health conditions or injuries might be relatively rare, but for the affected animals the consequences might be severe, and will worsen over time during transport and may lead to suffering; resting problems severity is expected to increase with increasing duration, as the lack of resting becomes more problematic for the animals and may lead to fatigue; even when a transport vehicle is fitted with water drinkers prolonged thirst may lead to dehydration and associated negative affective states, and physiological and behavioural changes that are likely to be associated with thirst have been identified after 8 h of transport; and due to practical difficulties in feeding animals on a transport vehicle, based on behavioural and physiological indicators, prolonged hunger is likely present after 12 h of feed deprivation. Depending on the pre-transport fasting this may correspond with the early hours of a journey.

Per definition, breaks in journeys function to remove the animals from the hazards that they are exposed to during transit, and to allow them to recover from the associated welfare consequences. The available evidence suggests that, with the transport conditions currently practiced, the provision of water on the vehicle while in motion may be ineffective, in that some animals will not, or are not able to drink. Insufficient research is available to assess the effectiveness and optimal manner in which to provide water on a vehicle. No studies have documented successful feeding of pigs during the transit stage. Thus, if pigs are to recover from the welfare consequences experienced during transit they need to be unloaded from the vehicle.

A journey break at a control post allows animals to have access to resting, watering and feeding areas. However, at control posts, along with the mitigation of welfare consequences, there is also potential for exposure to hazards resulting in welfare consequences or interfering with the intended mitigation of other welfare consequences. In addition, control posts involve biosecurity risks as animals



can be exposed to infectious diseases through direct or indirect contact with other animals and opportunistic pathogens. Across the categories of pigs typically transported on journeys involving journey breaks, the scientific focus on control posts has been limited. This means that whether control posts in their current state fulfil their intended function is not known.

The specific scenarios relevant to pigs that EFSA was asked by the Commission to consider were the transport of cull sows to slaughterhouses, and issues relating to 'special health status animals', i.e. animals whose unloading before the final destination might jeopardise their health status.

Health issues are major causes for culling sows, so these animals present additional challenges when it comes to transporting them to a slaughterhouse. Therefore, the key animal welfare issue for cull sows transported to slaughter is to determine the fitness of the sows to be transported. In addition, heat stress is a particular concern, especially in lactating sows, because of their high sensitivity to heat. Another concern during transport is that sows are often mixed with unfamiliar individuals, leading to group stress and resultant injuries. The Opinion contains a number of recommendations on how welfare consequences from transport can be prevented or mitigated for cull sows during transport to slaughter.

The failure to unload pigs in order to protect their 'special health status' presents a concern with respect to the welfare of the animals in question. If, for biosecurity reasons, pigs are not unloaded to provide required rest, feed and water, facilities must be available on the vehicle to provide the necessary resting, feeding and drinking, and suitable microclimatic conditions. However, no scientific studies have been found that confirm that this is currently possible.



Table of contents

Abstract.	
1	Introduction
1.1	Background and Terms of Peference as provided by the requestor
1 1 1	Background and Terms of Reference as provided by the requestor
1 1 2	Torms of Deference
1.1.2.	Assocsment of common transport practices
1 1 2 2	Assessment of common transport practices
1.1.2.2.	Interpretation of the Terms of Deference
2	Data and mothodologies
2.	Data and methodologies
2.1.	Data from literature
2.1.1.	Data from Public Consultation
2.1.2.	Methodologies
2.2.	Experts' oninion
2.2.1.	Literature searches
3	Accecement
3.1	The transport of pigs in the EU
3.1.	Welfare consequences associated with transport of pigs
321	Negative affective states
322	Definition and interpretation of ABMs for highly relevant welfare consequences during hig transport
33	Preparation of nins for transport
331	Current practice
332	Highly relevant welfare consequences
333	Fitness for transport
3331	Introduction
3332	Assessment of fitness for transport in pigs
3333	Transport of animals with reduced fitness
3.4.	Loading/unloading
3.4.1.	Current practice
342	Highly relevant welfare consequences
3.5.	Transit stage
3.5.1.	Current practice
3.5.2.	Highly relevant welfare consequences
3.5.3.	Quantitative examination of thresholds to protect the welfare of pigs during the transit stage:
	microclimatic conditions, space allowance and journey time
3.5.3.1.	Threshold of microclimatic conditions
3.5.3.2.	Threshold of space requirements during journeys.
3.5.3.3.	Thresholds for journey times
3.6.	Journey breaks
3.6.1.	Provision of water and/or feed on the vehicle while stationary
3.6.2.	Unloading of the pigs into a pen where water and/or feed is provided for a period of time before
	reloaded to continue the journey
3.6.3.	Control posts
3.6.3.1.	Current practice
3.6.3.2.	Highly relevant welfare consequences
3.6.4.	Journey break – summarising considerations
3.7.	Specific scenarios: Transport of cull sows to slaughter
3.8.	Specific scenarios: 'Special health status pigs'
3.9.	Specific scenario: The transport of pigs by air
3.10.	Specific scenario: The transport of pigs by train
3.11.	Uncertainty analysis
4.	Conclusions
4.1.	General conclusions on transport of pigs
4.2.	Conclusions on the preparation of pigs before transport
4.3.	Conclusions for loading/unloading of pigs during road transport
4.4.	Conclusions for the transit stage during road transport of pigs
4.5.	Conclusions on journey breaks and control posts
4.6.	Welfare of pigs when transported by air and rail
4.7.	Conclusions on specific scenarios
5.	Recommendations

5.1.	General recommendations on the transport of pigs by road	91	
5.2.	Recommendations for preparation of pigs before transport	91	
5.3.	Recommendations for loading/unloading of pigs during road transport	92	
5.4.	Recommendations for the transit of pigs during road transport	92	
5.5.	Recommendations for journey breaks and control posts	94	
5.6.	Recommendations for the transport of pigs by air and rail	94	
5.7.	Recommendations on specific scenarios	94	
Reference	25	95	
Abbreviati	ions	106	
Appendix	Appendix A – Template used during the selection of the highly relevant welfare consequences 10		



1. Introduction

1.1. Background and Terms of Reference as provided by the requestor

1.1.1. Background

In the framework of its Farm to Fork strategy, the Commission will start a comprehensive evaluation of the animal welfare legislation. This will include the following acts:

- Council Directive 98/58/EC of 20 July 1998 concerning the protection of animals kept for farming purposes.
- Council Directive 1999/74/EC of 19 July 1999 laying down minimum standards for the protection of laying hens.
- Council Directive 2008/119/EC of 18 December 2008 laying down minimum standards for the protection of calves.
- Council Directive 2008/120/EC of 18 December 2008 laying down minimum standards for the protection of pigs.
- Council Directive 2007/43/EC of 28 June 2007 laying down minimum rules for the protection of chickens kept for meat production.
- Council Regulation (EC) No 1/2005 of 22 December 2004 on the protection of animals during transport and related operations and amending Directives 64/432/EEC and 93/119/EC and Regulation (EC) No 1255/976.
- Council Regulation (EC) No 1099/2009 of 24 September 2009 on the protection of animals at the time of killing.

These acts are based on scientific opinions that are outdated. The current EU legislation on the protection of animals during transport is based on a scientific opinion adopted in 2002. Since then, the EFSA adopted opinions in 2004 (two opinions) and 2011.

In the context of possible drafting of legislative proposals, the Commission needs new opinions that reflect the most recent scientific knowledge.

Against this background, the Commission would like to request the EFSA to review the available scientific publications and possibly other sources to provide a sound scientific basis for future legislative proposals.

This request is about the protection of terrestrial animals during transport.

1.1.2. Terms of Reference

The Commission therefore considers opportune to request EFSA to give an independent view on the protection of animals during transport for the following groups and categories of farmed animals: Free-moving animals (group 1):

- 1) Equids (horses, donkeys and their crossings),
- 2) Bovine animals (cattle and calves),
- 3) Small ruminants (sheep and goats),
- 4) Pigs,

Animals in containers (group 2):

- 5) Domestic birds (chickens for meat, laying hens, turkeys, ducks, geese, quails, etc.)
- 6) Rabbits.

The request refers to any journey, i.e., journeys of less than 8 h ("short journeys"), journeys of more than 8 h ("long journeys") and long journeys that need unloading and/or feeding ("very long journeys").

1.1.2.1. Assessment of common transport practices

For each category of animals (1–6), the EFSA will describe, based on existing literature and reports, the current practices regarding:

a) the preparation for transport (including catching and crating of poultry and rabbits), loading, unloading and handling of animals at all stages of the journey, including at destination,



- b) the means of transport by road, roll-on-roll-off vessels, livestock vessels, the means of transport by rail and by air,
- c) the conditions within the means of transport: space, microclimatic conditions, watering and feeding,
- d) the journey duration and its circumstances as well as the resting of animals in the vehicle being stationary or being unloaded,
- e) the conditions for areas where animals are unloaded and/or grouped as part of the journey (assembly centres, livestock markets, control posts, EU ports).

Additionally, for each of the above practices, the EFSA will:

- Describe the relevant welfare consequences for each category of animals during each step of the process. Relevance will not need to be based on a comprehensive risk assessment, but on EFSA's expert opinion regarding the severity, duration and occurrence of each welfare consequence,
- Define qualitative or quantitative measures to assess the welfare consequences during transport (animal based measures),
- Identify the hazards leading to these welfare consequences,
- Provide recommendations to prevent, mitigate or correct the welfare consequences (resource and management based measures).

1.1.2.2. Assessment of seven specific transport practices

For the following scenarios, the Commission has identified practical difficulties or insufficient information in ensuring the welfare of animals. At least for them, the EFSA is asked to propose detailed animal-based measures and preventive and corrective measures with, where possible, either qualitative (yes/no question) or quantitative (minimum/maximum) criteria (i.e., requirements to prevent and/or mitigate the welfare consequences):

- 1) "Export by livestock vessels" Transport of adult cattle, weaned calves and sheep over long journeys involving the combination road/livestock vessels;
- 2) "Export by road" Transport of adult cattle, weaned calves and sheep over long journeys by road involving the use of facilities where animals are unloaded and reloaded (control posts, livestock markets) or when animals are kept in stationary vehicles for hours (exit points) including in third countries;
- 3) "Roll-on-roll off" Transport of adult cattle, calves and sheep over long journeys involving the combination road/roll-on-roll-off vessels;
- 4) "End-of-career animals" Transport of end of career animals to slaughterhouses of dairy cows, breeding sows, and laying hens;
- 5) "Unweaned calves" Transport of unweaned calves over long journeys; this scenario will particularly consider the risks regarding fitness for transport, watering, feeding and thermal comfort under Section c of the current practices associated with inappropriate drinkers and liquid feed for unweaned calves;
- 6) "Horses" Transport of horses on long journeys to slaughterhouses;
- 7) "Special health status animals" Transport of ruminants and pigs where unloading them before the final destination might jeopardize their health status.

For all scenarios, the EFSA will consider the risks regarding microclimatic conditions under Section c of the current practices associated with extremely high or low temperatures including the difficulty of measuring of temperature, humidity and gas concentration within animals' compartment.

1.2. Interpretation of the Terms of Reference

This Scientific Opinion concerns the protection of pigs during transport. The fundamental premise of the work underlying this Scientific Opinion is that it is an accepted practice that humans breed animals for food, sports and leisure. The assessment does not go into details with breed differences or the consequences of the housing system or production system, from which the animals to be transported are coming, even though it cannot be excluded that welfare consequences (WCs) of transport to some extent differ depending on for example the previous husbandry conditions.

The Opinion deals with the preparation, loading and unloading, transit and journey breaks. For the purpose of this Opinion, the preparation phase involves all types of actions and animal management that take place during the interval from the decision to transport pigs until the initiation of loading of



the animals onto a vehicle or other means of transport. In effect, in this Opinion, the preparation of pigs for transport essentially involves the gathering of the animals to holding facilities and the keeping of them there prior to transport. Loading starts when the first animal is moved from the holding pen into the means of transport and ends when the last animal is loaded and up until the ramp is closing. Unloading starts when the ramp is open, and the first animal exits the means of transport, and ends when the last animal exits the means of transport, and ends when the last animal exits. Loading and unloading will be dealt with together due to the similarities of the processes. The transit starts when the ramp has been closed and ends when the first animal unloads. Journey breaks conceptually apply to periods when the vehicle is stopped on the side of a road, or when animals are offloaded to other facilities for feeding, watering and resting, including control posts (CPs). Legislation regarding drivers of animal transport vehicles affect animal transport, especially on journeys with only one driver, as drivers have to have rest breaks, where vehicles will be stationary (Table 1). As these breaks are not aimed to rest, feed and water the animals, for the purpose of the current assessment, they are not included in the 'journey break' stage.

Table 1: A summary of the EU drivers' hours rules and sector specific working time rules (Department for transport, 2014)

Drivers' hours rules Regulation (EC) 561/2006	Working time rules Directive 2002/15/EC		
 Driving 9 h daily driving limit (can be increased to 10 h twice a week) Max 56 h weekly driving limit Max 90 h fortnight driving limit 	 Working time (including driving) Working time must not exceed average of 48 h a week (no opt out)¹ Max working time of 60 h in one week (provided average not exceeded) Max working time of 10 h if night work performed.² 		
 Breaks 45 min break after 4.5 h driving A break can be split into two periods, the first being at least 15 min and the second at least 30 min (which must be completed after 4.5 h driving) 	 Breaks³ A driver cannot work for more than 6 h without a break. A break should be at least 15 min long. 30 min break if working between 6 and 9 h in total.⁴ 45 min break if working more than 9 h in total. 		
 Rest 11 h regular daily rest⁵; which can be reduced to 9 h no more than three times a week. 45 h weekly rest, which can be reduced to 24 h, provided at least one full rest is taken in any fortnight. There should be no more than six consecutive 24 h periods between weekly rests. 	 Rest Same rest requirements as EU' drivers. 		

(1): Normally calculated over a rolling 17-week period, but can be extended to 26 weeks under a collective or workforce agreement.

(2): Can be extended under a collective or workforce agreement.

(3): EC Regulation 561/2006 is directly effective and takes precedence over EC Directive 2002/15 - Article 2.4 Directive 2002/15. Therefore, EU drivers' hours break requirements take precedence when driving.

(4): After working for 6 h a mobile worker must take a break of at least 15 min. However, if working more than 6 and up to 9 h in a shift a mobile worker needs to take a break totalling at least 30 min - this could be two breaks of 15 min. Where a shift will contain more than 9 h of working time, a total of 45 min of break is needed.

(5): Alternatively, this regular daily rest period may be taken in two periods, the first of which must be an uninterrupted period of at least 3 h and the second an uninterrupted period of at least 9 h.

This Opinion does not focus on the different types of premises defined in the transport Regulation (e.g. markets, assembly centres) but refers to them where appropriate. Destination is not covered, but only specified when important considerations are found. In case of animals arriving for slaughter, additional information can be found in the EFSA Opinion on Welfare of Pigs at Slaughter (EFSA AHAW Panel, 2020).

Within pigs, different animal categories exist, such as gilts and weaners. Across this Scientific Opinion, focus is given to three often transported categories of pigs: weaners transported for further fattening, finishers transported to slaughter, and cull sows and boars. When specific studies are

referred to, the average body weights of the animals involved are mentioned (when available) as well as the animal category involved in the study (e.g. gilts).

The scientific assessment carried out by EFSA takes two forms. First, for road transport practices, which is the most common transport practice, the transport stages are described and assessed in terms of WCs, animal-based measures (ABMs) and hazards leading to the WCs. In addition, recommendations to prevent hazards, and mitigate/correct WC are provided. The preventive measures (PRE) relate to the hazards, and the corrective/mitigative measures refer to the WCs. Where possible, the assessment leads to the establishment of recommendations on quantitative thresholds for microclimatic conditions within means of transport (maximum temperature), and to spatial thresholds (minimum space allowance). In addition, the development of WCs over time is assessed in relation to maximum journey duration.

Second, for the specific industry practices (specific scenarios) listed in the mandate that relate to pigs, EFSA examines selected welfare concerns (defined as an area or a topic to which special attention should be given in order to potentially avoid WCs), and, where possible, suggests recommendations. A similar approach will be taken for transport by air and rail. In relation to the specific transport practice of 'end-of-career animals' in this Scientific Opinion, this animal category will be denoted 'cull animals', which is the term used in scientific literature.

A list of WCs is selected among those reported in the guidance protocol published by EFSA (EFSA AHAW Panel, 2022a). These WCs can lead to negative affective states such as fear, pain and/or distress. For each transport stage, the highly relevant WCs are selected based on literature and expert opinion considering the severity, duration and frequency of occurrence of the WC. When possible, each WC is linked to one or more ABMs that are indicative of it.

During preparation for transport, animals might present health conditions (including WCs, such as injuries) that may increase in severity during transport. Certain physiological conditions, while not being a WC as such (e.g. pregnancy or certain age categories), are conditions that predispose the animal to experience WCs if transported. Rather than assessing all WCs that might occur at any given stage of transport due to animals being unfit for transport, a separate section in the Scientific Opinion, focusing on fitness for transport is developed as part of the assessment of the preparation stage.

For the purposes of this scientific assessment, failure to implement or non-compliance with the current rules as specified in the transport regulation are not considered. This is outside the remit of EFSA as a risk assessor. During the work, the EFSA Experts may include scientific information from practices currently prohibited in the EU.

The assessment is not split according to the current legislation, for example specifying that 8 h as journey duration is the threshold between short and long journeys (each with specific legislative requirements). Alternatively, the assessment is performed on a journey carried out in the EU of an unspecified length and duration.

2. Data and methodologies

2.1. Data

2.1.1. Data from literature

The information contained in the scientific papers and reports identified as relevant during the literature search was used as a basis for the text of this Scientific Opinion. Additional sources were added by the WG experts when dealing with specific sections.

2.1.2. Data from Public Consultation

To consult interested parties and gain feedback on EFSAs interpretation of the transport mandate, a Public Consultation was launched in the period 15 April–10 June 2021. In particular, EFSA called on interested parties to:

- identify current transport practices of particular concern not already identified by EFSA in the interpretation of mandate;
- describe the practical difficulties or insufficient information in ensuring the welfare of animals, for the specific transport practices listed in the request from the European Commission and for any other additional practices of concern that might be identified;



• provide any available recorded data from road or sea transport, for example from a data logger, related to the microclimatic environment (temperature, humidity and ammonia levels). The data should demonstrate a link between the microclimatic conditions and any adverse WCs that are experienced by the animals during transport.

The information received in the Public Consultation were considered by the EFSA experts as part of their work on this Opinion (See Annex A: Report of the Public Consultation on the Protection of Animals during Transport, published under 'Supporting Information' in the Opinion on transport of small ruminants).

2.2. Methodologies

This Scientific Opinion follows the guidance protocol that was developed by the AHAW Panel to deal with all the mandates in the context of the Farm to Fork strategy revision (EFSA AHAW Panel, 2022a).

To address the terms of reference of the mandate, EFSA translated the assessment questions into more specific sub-questions. These are interrelated, meaning that the outcome of each sub-question is necessary to proceed to the next sub-question. The approach to develop the sub-questions was based on evidence from the scientific literature and expert opinion. The translation of the assessment questions into sub-questions is mapped in Table 2.



18314732, 2022.

Assessment Questions		Sub-questions		
i.	Describe the current transport practices	1. Identify and select relevant transport scenarios (common animal transport practices per species and animal category)	2. Describe the transport practices	
		Aim: Animal transport practices to be considered in the assessment are identified and selected to be common (representative of the current practice) in the EU.	Aim: All the animal transport practices per animal category identified and selected from Sub-question 1 are described narratively	
		Approach: expert opinion via group discussion.	Approach: Literature review.	
		Relationship with assessment question: This sub-question is necessary for the overall assessment question requiring the description of the practices.	Relationship with assessment question: this corresponds to the assessment question and is necessary for the next assessment question.	
ii.	Describe the relevant welfare consequences that may occur due to the practices	3. Identify the welfare consequences common for all mandates and provide their definitions	4. Select the most relevant welfare consequences for the selected animal transport practices	
		Aim: to identify the welfare consequences and provide a definition for them. EFSA generates a list of welfare consequences common for all mandates.	Aim: To select the most relevant welfare consequences for each of the previously defined animal transport scenarios per species or animal category.	
		Approach: Expert opinion via group discussion (see focus and full resulting list in Section 3.2).	Approach: expert opinion via EKE (see Section 2.2.1). Relationship with assessment question: this corresponds to	
		Relationship with assessment question: the list of all possible welfare consequences is necessary for the next assessment question asking to identify the most relevant ones per each system	which relevant welfare consequences are identified only for current transport scenarios.	
iii.	Define qualitative or quantitative animal-based measures (ABMs) to assess these welfare consequences	5. Identify the feasible ABMs for the assessment of the most relevant welfare consequences	6. Describe the feasible ABMs for the assessment of the most relevant welfare consequences	
		Aim: The ABMs for the assessment of the welfare consequences previously identified as relevant are selected (only for feasible ABMs).	Aim: The ABMs for the assessment of the welfare consequences previously identified as the most relevant are described.	
		Approach: Expert opinion via group discussion.	Approach: Literature review.	
		Relationship with assessment question: This corresponds to the assessment question and is related to Sub-question 4 in which ABMs are identified only for the most relevant welfare consequences	Relationship with assessment question: related to Sub- question 5.	

Table 2: Specific assessment questions and sub-questions of the mandate



18314732, 2022, 9, Dov

[30/01/2025]. Se

Creative Co

icense

Assessment Questions		Sub-questions		
iv. Identify the hazards leading to these welfare consequences		7. Identify the hazards leading to the most relevant welfare consequences	8. Describe the hazards leading to the most relevant welfare consequences	
		Aim: The hazards leading to the most relevant welfare consequences are identified.Approach: Expert opinion via group discussion.Relationship with assessment question: This corresponds to the assessment question and is related to Sub-question 4 in which hazards are identified only for the most relevant welfare consequences	Aim: The hazards are described. Approach: Literature review. Relationship with assessment question: related to Sub- question 6.	
v.	Provide recommendations to prevent, mitigate or correct the hazards	9. Identify the preventive and corrective measures for the most relevant welfare consequences	10. Describe the preventive and corrective measures for the most relevant welfare consequences	
		Aim: preventive and corrective measures for the most relevant welfare consequences for the previously defined transport scenarios per animal category are identified. Approach: Expert opinion via group discussion.	Aim: Preventive and corrective measures are described Approach: Literature review. Relationship with assessment question: related to Sub- question 8	
		Relationship with assessment question: this corresponds to the assessment question and is related to Sub-question 4 in which preventive and corrective measures are identified only for the most relevant welfare consequences.		



2.2.1. Experts' opinion

The data obtained from the literature and public consultation were complemented by the opinions of the EFSA experts. As described in Table 2, expert opinion was mainly used for the sub-questions requiring the identification of transport practices, WCs, ABMs, hazards, preventive and corrective or mitigative measures. Expert opinion was mainly elicited via discussion among EFSA experts. However, for the identification of the highly relevant WCs, an informal, structured Expert Knowledge Elicitation (EKE) was carried out.

As explained above (Sub-question 4), the mandate requested the identification of the highly relevant WCs for each of the defined animal transport practices.

The starting point was the list of 33 specific WCs identified under Sub-question 3 (for details see Section 3.1.1.3 of the protocol, EFSA AHAW Panel, 2022a). The exercise was carried out separately for each of the animal transport stages per species or animal category resulting from Sub-question 1.

The exercise consisted of selecting the highly relevant WCs out of these 33 per each of these combinations (species/animal category \times transport stage).

For each combination, EFSA experts classified, based on an estimate of their magnitude, the 33 WCs into four categories of relevance: (i) non-applicable, (ii) slightly relevant, (iii) moderately relevant and (iv) highly relevant. Appendix A contains an example of this process. The magnitude of a WC was defined as the product of three parameters: severity, duration and frequency of occurrence (EFSA AHAW Panel, 2012). Owing to the lack of published data on these three parameters, the experts expressed their qualitative expert opinion on the magnitude of WCs.

Expert opinion is elicited in three phases:

- 1) First phase: the experts go individually through the list of welfare consequences and identify those that would fall in the 'non-applicable' or 'slightly relevant' categories. Their individual judgements are then be collated, and those WCs unanimously identified as belonging to these two categories are removed and not considered for further assessment. Those WCs for which there is no consensus whether they are considered 'non-applicable' or 'slightly relevant' remain for further assessment and require an open group discussion to find a consensus.
- 2) Second phase: the experts individually went through the list of remaining WCs and identified those that fell in the category of 'highly relevant". These were kept for further assessment. Similarly, as during the first phase in case discrepant opinions emerged, consensus was sought through group discussion.
- 3) Third phase: the experts are asked to rank individually all the remaining WCs in the list that are not already identified as highly relevant (and thus kept) or non-applicable or slightly relevant (and thus removed) from the highest to the least relevant. Their individual rankings are then discussed again in an open group discussion with the aim to assign the remaining WCs into the category 'highly relevant' or in the category 'moderately relevant'.

The Scientific Opinion only reports, for each of the defined animal transport stages, those WCs that were selected to be highly relevant from this exercise (since the mandate asks for the 'most relevant' WCs in each identified transport practice).

Expert opinion was also part of the syntheses involved in the development of the quantitative recommendations for specific conditions within the means of transport (space allowance, microclimatic conditions, development over time) relevant to the assessment.

2.2.2. Literature searches

As described in Table 2, literature searches were carried out for the sub-questions requiring the description of transport stages, WCs, ABMs, hazards, preventive and corrective or mitigative measures.

First broad searches of literature providing information on current practices on transport of the animal categories and species included in the 'free-moving' mandate were carried out. Restrictions were applied in relation to the date of publication, considering only records published after a previous EFSA Scientific Opinion on the topic (EFSA AHAW Panel, 2011).

Following the broad searches, more specific searches were carried out focusing on welfare consequences, ABMs, hazards, preventive and corrective or mitigative measures.

For pigs, the search results (general + specific) yielded a total of 1,910 (235 + 1,675) records that were exported to an EndNote library together with the relevant metadata (e.g. title, authors, abstract). Titles and abstracts were screened to remove irrelevant publications (e.g. related to species, processes

18314732, 2022, 9, Downoaded from https://efsa.onlinelibrary.wiley.com/doi/10.2903/j.sa.2022.7445 by Universidad De Zaragoza, Wiley Online Library on [3001/2025]. See the Terms and Conditions (https://onlinelibrary.wiley.com/terms-and-conditions) on Wiley Online Library for nales of use; OA articles are governed by the applicable Creative Commons License

and research purposes that were out of scope of this Opinion) and duplicates, and successively to identify their relevance to the topic. The screening led to 397 relevant records. Experts screened these papers and selected 223 for further assessment. Full texts were retrieved and made available to the experts.

The search terms were saved in Web of Science and rerun with any results (records) subsequent to 2021 screened and added to the pool of papers available to the experts. In addition, the experts selected relevant references starting from scientific papers, including review papers, book chapters, non-peer-reviewed papers known by the experts themselves, or retrieved through non-systematic searches, until the information of the topic was considered sufficient to undertake the assessment by the EFSA experts. If needed, relevant publications from before 2011 were considered.

3. Assessment

3.1. The transport of pigs in the EU

Transport of animals between MS and exports from the EU are recorded in the TRAde Control and Expert System (TRACES), which is the European Commission's multilingual online platform for sanitary and phytosanitary certification required for intra-EU trade and importation of animals, semen, embryos, food, feed, and plants (https://ec.europa.eu/food/animals/live_animals_en). However, movements inside MS (i.e. to slaughterhouses or between farms) are not recorded in this database (Rojek, 2021).

According to TRACES, around 35 million pigs were transported between MS in 2019, 33 million in 2020 and 25 million in 2021, across all means of transport. Road transport constituted the vast majority of total pig transport in these years.

3.2. Welfare consequences associated with transport of pigs

During the last decades, several scientific reviews (e.g. Cockram, 2007; Nielsen et al., 2011), textbooks (e.g. Grandin, 2019) and international organisations (e.g. WOAH, 2011) have described and discussed the consequences of animal transport in terms of animal welfare. In general, it is agreed that animal transport can lead to severe negative animal welfare consequences. Transport of animals is known as a complex stressor involving many aspects (related to the condition of the animals, their general biological characteristics, as well as the conditions under which the transport takes place, including journey duration), the majority of which to some extent may influence animal welfare. Thus, when analysed in detail, a highly complex picture emerges and transport stress must be considered as a multifactorial stressor.

Across the different stages of the transport of pigs, the following WCs were selected as highly relevant: group stress, handling stress, heat stress, injuries, motion stress, prolonged hunger, prolonged thirst, resting problems, restriction of movement and sensory overstimulation (Table 3). It is clear from this table that most transport stages involve several WCs.

For the purpose of this Scientific Opinion, the WC injuries, was created as a combination of the WCs 'Soft tissue lesions and integument damage' and 'Bone lesions' (see Section 3.1.1.3 of the protocol, EFSA AHAW Panel, 2022a).

			Transport s	tages	
Welfare consequences and definitions		Preparation	Loading/ unloading	Transit	Journey break
Group stress	The animal experiences stress and/ or negative affective states such as pain, fear and/or frustration resulting from a high incidence of aggressive and other types of negative social interactions, often due to hierarchy formation and competition for resources or mates.	X		X	Х

Table 3: Welfare consequences selected as highly relevant per each of the transport stages involved in this Opinion



		Transport stages			
Welfare consequ	ences and definitions	Preparation	Loading/ unloading	Transit	Journey break
Handling stress	The animal experiences stress and/ or negative affective states such as pain and/or fear resulting from human or mechanical handling (e.g. loading/unloading).	X	Х		Х
Heat stress	The animal experiences stress and/ or negative affective states such as discomfort and/or distress when exposed to high effective temperature.	X	Х	X	Х
Injuries	The animal experiences negative affective states such as pain, discomfort or distress due to physical damage to somatic tissue types (bones, joints, skin muscles). This can be due to injuries or pathological changes.	X		X	х
Motion stress	The animal experiences motion sickness, stress and/or fatigue due to the forces exerted as a result of acceleration, braking, stopping, cornering, gear changing, vibrations and uneven road surfaces during transport.			X	
Prolonged hunger	The animal experiences craving or urgent need for food or a specific nutrient, accompanied by a negative affective state, and eventually leading to a weakened condition as metabolic requirements are not met.	X		X	х
Prolonged thirst	The animal experiences craving or urgent need for water, accompanied by an uneasy sensation (a negative affective state), and eventually leading to dehydration as metabolic requirements are not met.			X	x
Resting problems	The animal experiences stress and/ or negative affective states such as discomfort, and/or frustration due to the inability to lie/rest comfortably or sleep. (e.g. due to hard flooring or vibration during transport). This may eventually lead to fatigue.	X		X	x
Restriction of movements	The animal experiences stress and/ or negative affective states such as pain, fear, discomfort and/or frustration due to the fact that it is unable to move freely, or is unable to walk comfortably (e.g. due to overcrowding, unsuitable floors, gates, barriers).			X	X



Welfare consequences and definitions		Transport stages			
		Preparation	Loading/ unloading	Transit	Journey break
Sensory overstimulation	The animal experiences stress and/ or negative affective states such as fear or discomfort due to visual, auditory or olfactory under/ overstimulation by the physical environment.		Х	Х	

3.2.1. Negative affective states

The description of each WC reported in Table 3 refers to one or more negative states involving affective components (e.g. pain, fear, fatigue). These are the major negative affective states that derive from the occurrence of the WC, and that may potentially lead to animal suffering. A list and description of the negative states as derived from literature, and also described in the guidance protocol document (EFSA AHAW Panel, 2022a) is reported in Table 4.

Table 4: List and description of the negative states, that animals may experience, when exposed to at least one of the welfare consequences listed above

Negative affective state	Description
Boredom	Boredom is an unpleasant emotion including suboptimal arousal levels and a thwarted motivation to experience almost anything different or more arousing than the behaviours and sensations currently possible (adapted from Mason and Burn, 2011).
Discomfort	Discomfort can be physical or psychological and is characterised by an unpleasant feeling resulting in a natural response of avoidance or reduction of the source of the discomfort. Pain is one of the causes for discomfort, but not every discomfort can be attributed to pain. Discomfort in non-communicative patients is assessed and measured via behavioural expression, also used to describe pain and agitation, leading to discomfort being interpreted as pain in some conditions (Ashkenazy and Ganz, 2019).
Stress ¹ & Distress	STRESS ¹ : Stressors are events, internal or external to the body involving real or potential threats to the maintenance of homeostasis. When stressors are present, the body will show stress responses (biological defence to re-establish homeostasis – for example behavioural, physiological, immunological, cognitive, and emotional). Stress is a state of the body when stress responses are present (Sapolsky, 2002).
	DISTRESS: Distress is a conscious, negatively valenced, intensified affective motivational state that occurs in response to a perception that current coping mechanisms (involving physiological stress responses) are at risk of failing to alleviate the aversiveness of the current situation in a sufficient and timely manner (McMillan, 2020).
Fatigue	Physiological state representing extreme tiredness and exhaustion of an animal (EFSA AHAW Panel, 2020).
Fear	The animal experiences an unpleasant emotional affective state induced by the perception of a danger or a potential danger that threaten the integrity of the animal (Boissy, 1995).
Frustration	Negatively valenced emotional state consecutive to the impossibility to obtain what is expected or needed. Frustration is very often triggered by restriction of natural behaviours thus resulting in thwarted motivation to perform these behaviours.
Pain	An unpleasant sensory and emotional experience associated with, or resembling that associated with, actual or potential tissue damage (Raja et al., 2020).

(1): The term stress does not describe a negative affective state in itself, but it is mentioned and defined in the table as it is a prerequisite of distress.

3.2.2. Definition and interpretation of ABMs for highly relevant welfare consequences during pig transport

Only a few studies have evaluated systems for the assessment of welfare during transport for livestock species, and – so far – no validated protocol for the assessment of pig welfare before, during



18314732, 2022, 9, Downoaded from https://efsa.onlinelibrary.wiley.com/doi/10.2903/j.sa.2022.7445 by Universidad De Zaragoza, Wiley Online Library on [3001/2025]. See the Terms and Conditions (https://onlinelibrary.wiley.com/terms-and-conditions) on Wiley Online Library for nales of use; OA articles are governed by the applicable Creative Commons License

and after transport has been described. Consequently, no benchmark values documenting optimal animal welfare status during or after transport exist to inform this Scientific Opinion. In addition, benchmark values documenting good animal welfare do not exist for any of the ABMs listed in this Scientific Opinion in relation to transport.

The feasibility of an ABM may be defined as the practicality to carry out an assessment of the ABM under field conditions. Feasibility does not relate to the sensitivity, specificity or repeatability of recording of an ABM. A feasible ABM for use during transport should be able to be recorded quickly, without using any specialised equipment or laboratory test, at a low cost, and with no (or only minimal) interference with normal operation procedures (Llonch et al., 2015). Llonch et al. (2015) divided feasibility into three categories: high (easy and quick recording without any special needs/tools), medium (extra time and/or space needed for recording), and low (not able to record under field conditions).

Some ABMs may have acceptable feasibility when it comes to recordings as part of research projects, but not when it comes to recordings during routine transport, especially in the transit stage. No studies have evaluated whether an ABM is of low, medium or high feasibility during animal transport. Feasibility is therefore not further addressed in this Scientific Opinion. A similar lack of knowledge exists with reference to sensitivity and specificity of ABMs in the context of animal transport and so these characteristics will not be dealt with in this Scientific Opinion.

One of the main feasibility challenges in relation to ABMs during transport is the access to, and visibility of, the animals during particular stages, especially the transit stage. It is very difficult, if not impossible to see all animals when in a livestock truck, for example. This problem can be partially overcome by the use of cameras and/or other types of sensors. However, the mere presence of cameras or sensors does not overcome all of these challenges, as data generated by such sensors need to be analysed in some way leading to an interpretation of the practice in question. Currently, technological tools within this area are under development outside the transport context (as described in Larsen et al., 2021), but not yet applicable in transport practice. This constitutes a gap in knowledge. Until sensors and associated interpretive or alarm systems are available at a practical level, the stage of transport appears to be the biggest influence on the feasibility of any ABM during animal transport. Another aspect of transport stress that may be monitored and mitigated in the future by use of sensor technology is motion stress caused by movements of the vehicle during the transit stage (as described by Morris et al., 2021).

During preparation, loading, unloading and journey breaks off the vehicle, the animals can be properly inspected, and ABMs in principle utilised. Among these could be visually recognised indicators, but potentially also auditory indicators or physiological biomarkers that can be obtained from, e.g. saliva, or even behavioural tests of, e.g. latency to drink from a bucket. At present, however, these potential indicators need further development and validation for use in animal transport. There is also a subset of ABMs that are not feasible during transport even when the animals can be inspected (Llonch et al., 2015). Examples of these are physiological indicators requiring invasive procedures. Tables 5–13 below contain information on the definition and interpretation of ABMs, including ABMs considered potential candidates for future inspection, as well as ABMs that so far have only been used in scientific studies underlying the conclusions of the Scientific Opinion.

i) ABMs for the assessment of the welfare consequence group stress

Table 5:	ABMs selected for the assessment of group stress in pigs during preparation, the transit	
	stage and journey breaks	

ABM	Definition and interpretation of the ABM
Skin lesions and wounds Definition: Tissue damage such as bruises, scratches, and wounds (EFSA AHA Panel, 2012).	
	Interpretation: Aggressive interactions between pigs lead to skin lesions.
Fighting	Definition: Negative social interactions such as biting, head-knocking and threatening, pushing at the side or chasing each other.
	Interpretation : Fighting is an aggressive behaviour occasioned by mixing of pigs from different social groups, competition for food, and lack of adequate space to respond submissively to a threat and to achieve full avoidance action and move away.

ABM: animal-based measure.



- ii) ABMs for the assessment of the welfare consequence handling stress
- **Table 6:** ABMs selected for the assessment of handling stress in pigs during preparation and loading/unloading

ABM	Definition and interpretation of the ABM
High-level vocalisations (screams)	Definition: Squealing or screaming, at group level when pigs are moved or manipulated (adapted from Welfare Quality [®] , 2009; Dalmau et al., 2009).
	Interpretation: Pigs are prone to scream when scared, forced to do something they do not want or exposed to pain. Electric goad use is associated with such vocalisations (Correa et al., 2010).
Falling	Definition: Loss of balance, in which part(s) of the body (beside legs) are in touch with the floor (Dalmau et al., 2009; Welfare Quality [®] , 2009).
	Interpretation: Pigs may fall on the ramp or in barn/truck as a result of forceful handling, behaviour of other pigs, slippery flooring, steepness or obstacles.
Slipping	Definition: Loss of balance, without (a part of) the body being in touch with the floor (Dalmau et al., 2009; Welfare Quality [®] , 2009).
	Interpretation: Slipping occurs when pigs are forced to move rapidly on an inadequate surface.
Refusal to move forward	Definition: An animal that stops for at least two seconds not moving the body or head (freezing) or that refuses to move when coerced by the operator (Dalmau et al., 2009; Welfare Quality [®] , 2009).
	Interpretation: Pigs refusal to move forward when exposed to handling stress or loading/unloading area is not well designed (angle, depth of slope, flooring, lack of foot battens or lateral protection, etc.).
Mounting/overlapping	Definition : Pig mounts another pig, with its front legs on the back of the other pig (Dalmau et al., 2009; Welfare Quality [®] , 2009).
	Interpretation: Insufficient space on the ramp during loading and unloading due to inadequate group size in tight/narrow spaces and rough handling may lead to mounting and/or overlapping.
Post-mortem carcass lesions	Definition: Skin damages on carcases that are assessed in the slaughterhouse subjectively along the dressing line by using different photographic scales. The assessment can be conducted as a whole or separately in different parts of the carcass, such as head/shoulder, middle/loin, and ham (Faucitano, 2001).
	Interpretation: Post-mortem carcass lesions are considered as evidence of poor handling and inadequate facilities at farm and during all stages of loading/ unloading.

ABM: animal-based measure.

iii) ABMs for the assessment of the welfare consequence heat stress

Table 7: ABMs selected for the assessment of heat stress in pigs during preparation, loading/ unloading, transit and journey breaks

ABM	Definition and interpretation of the ABM
Respiratory rate	Definition: Breathing rate. Usually measured by counting the movements of the flank from direct observation and converting this into breaths per minute (BPM).
	Interpretation: Heat stress leads pigs to show increased respiration rates until reaching panting. Respiration rate increases with body temperature in order to maintain homeostasis (Brown-Brandl et al., 2001; Bjerg et al., 2020).
Rectal temperature	Definition: A proxy for the body temperature of the pig as measured by a thermometer inserted in the rectum (Yundong, 2012).
	Interpretation: Body temperature rises during heat stress when the physiological and behavioural mechanisms for the dissipation of heat can no longer maintain equilibrium because of heat gained from excessive environment heat combined with metabolic heat production.

18314732, 2022, 9, Downloaded from https://efsa.onlinelibrary.wiley.com/doi/10.2903/j.efsa.2022.7445 by Universidad De Zaragoza, Wiley Online Library on [3001/2025]. See the Terms and Conditions (https://onlinelibrary.wiley.com/terms-and-conditions) on Wiley Online Library for rules of use; OA articles are governed by the applicable Creative Commons License

ABM	Definition and interpretation of the ABM
Panting	Definition: Short breathing carried with the mouth open (Bracke et al., 2020). The first phase of panting is characterised by rapid, shallow breathing called thermal polypnea, associated with an increase in respiratory rate. This shifts to a second phase characterised by slower and deeper breathing, named thermal hyperpnea characterised by an increase of alveolar ventilation rate.
	Interpretation : Evaporation of water by panting is the primary form for pigs to loose heat at high temperatures (Huynh et al., 2005). Panting is considered a physical sign of stress and it is associated with dissipation of excess heat and/or fatigue.
Skin discoloration	Definition: Changes from light to redder colour of the skin with a transient blotchy appearance (Kephart et al., 2010).
	Interpretation : It is considered a physical sign of stress due to the effort related to physical activities such as loading/unloading (Pilcher et al., 2011).

ABM: animal-based measure.

iv) ABMs for the assessment of the welfare consequence injuries

Table 8: ABMs selected for the assessment of injuries in pigs during preparation, transit and journey breaks

ABM	Definition and interpretation of the ABM
Skin lesions and wounds	Definition: Tissue damage such as bruises, scratches, and wounds (EFSA AHAW Panel, 2012).
	Interpretation: Damage may be caused by biting among pigs and sharp or abrasive structures in handling facilities or vehicles.
Lameness	Definition : Clinical manifestation of locomotory changes characterised by alterations from normal gait or posture and potential reduced mobility (Nalon et al., 2013)
	Interpretation: Most lameness is a consequence of pain but it can also occur from mechanical defects resulting in physical incapacity. Behavioural changes associated with lameness indicate attempts by the animal to protect the affected limb from pain and further injury. Lameness is relevant for the welfare of the pigs, not only as indicator of pain, but also reduces the capacity of the animal to cope with the environment (Dalmau et al., 2010).
Post-mortem carcass lesions	Definition: Skin damages on carcases are assessed in the slaughterhouse subjectively along the dressing line by using different photographic scales. The assessment can be conducted as a whole or separately in different parts of the carcass, such as head/ shoulder, middle/loin, and ham (Faucitano, 2001).
	Interpretation: Post-mortem carcass lesions are considered as evidence of poor handling and inadequate facilities at farm and during all stages of loading/unloading.

ABM: animal-based measure.

- v) ABMs for the assessment of motion stress and sensory overstimulation
- **Table 9:** ABMs selected for the assessment of motion stress and sensory overstimulation in pigs during loading/unloading and in transit

АВМ	Definition and interpretation of the ABM
Foaming at the mouth	Definition: An excess of saliva mixed with air or gasses to create a foam.
	Interpretation : Hypersalivation at the mouth is a symptom of travel sickness and is often associated with chomping and retching (Randall and Bradshaw, 1998).
Vomiting	Definition: Vomiting is a protective response and coordinated reflex where upper gastrointestinal tract contents are forcefully ejected from the mouth (Santurtun and Phillips, 2015).
	Interpretation : During journey, pigs may vomit as a sign of motion sickness (Santurtun and Phillips, 2015).



ABM	Definition and interpretation of the ABM
Falling	Definition: Loss of balance, in which part(s) of the body (beside legs) are in touch with the floor (Dalmau et al., 2009; Welfare Quality [®] , 2009).
	Interpretation: Pigs may fall on the ramp or in barn/truck as a result of forceful handling, behaviour of other pigs, slippery flooring, steepness or obstacles.
Heart rate (HR)	Definition: The number of heart beats per unit time, usually per minute.
	Interpretation : The response to stressors leads to a secretion of catecholamines that prepare the body to respond, by for example, increasing heart rate (Von Borell, 2007). HR is affected by factors other than acute stress such as the exercise associated with movement in response to handling and loading. HR reflects the activity of the sympathetic branch of the autonomic nervous system and therefore is an indicator of the stress response and of the animal's emotional reactivity.
Plasma cortisol	Definition: When a threat is perceived by an animal, the hypothalamic–pituitary– adrenal (HPA) axis gets activated resulting in release of several hormones that lead to the synthesis of glucocorticoids such as cortisol in the adrenal cortex and its consequent increase into the bloodstream (Turner and Tilbrook, 2006).
	Interpretation: Plasma cortisol concentration increases from the moment of loading an indirect measure of the stress experienced by an animal when exposed to adverse conditions during the initial part of a journey (Miranda-De La Lama et al., 2009). With increased journey duration feedback mechanisms can reduce its concentration and the plasma cortisol concentration may no longer reflect the continued perception of pigs. Continued hypothalamic stimulation results in the continual secretion of corticotropin-releasing hormone even though plasma cortisol concentration may have fallen (Smith et al., 2003).
Post-mortem carcass lesions	Definition: Skin damages on carcases are assessed in the slaughterhouse subjectively along the dressing line by using different photographic scales. The assessment can be conducted as a whole or separately in different parts of the carcass, such as head/ shoulder, middle/loin, and ham (Faucitano, 2001).
	Interpretation: Pigs with lesions might have hurt themselves during the journey.

ABM: animal-based measure.

vi) ABMs for the assessment of the welfare consequence prolonged hunger

Table 10:	ABMs selected for the assessment of prolonged hunger in pigs during preparation, wh	nile
	in transit and during journey breaks	

ABM	Definition and interpretation of the ABM
Aggressive behaviour	Definition: Negative social interactions such as fighting, head-knocking, threatening, and biting (Welfare Quality [®] , 2009).
	Interpretation : Negative social interactions occur during most transport stages due to mixing of unfamiliar pigs, and are generally increased in pigs that have been food deprived.
Latency to feed after unloading	Definition: A quantification of the time interval from unloading and until the animal is observed eating for the first time
	Interpretation: A short latency to eat is a reflection of a high motivation to eat. Can be affected by other factors such as fear.
Competition for feed	Definition: Occurs when more than one pig tries to access a food source or one pig monopolise the feed. Usually, aggression is seen.
	Interpretation: When pigs start to compete for access to feed, it is a sign of increased motivation to eat.
Plasma glucose	Definition : Glucose is a monosaccharide, which can be stored as a multi-branched polysaccharide, called glycogen, in skeletal muscle and the liver (Terlouw and Bourguet, 2022). In monogastric animals, like pigs, blood glucose levels reflect the nutritional status of the animal (Shaw and Tume, 1992).

18314732, 2022, 9, Downloaded from https://efs.onlinelibury.wiley.com/doi/10.2903j.efs.a.2022.7445 by Universidad De Zaragoza, Wiley Online Library on [3001/2025]. See the Terms and Conditions (https://onlinelibrary.wiley.com/terms-and-conditions) on Wiley Online Library for ules of use; OA articles are governed by the applicable Creative Commons License

ABM	Definition and interpretation of the ABM
	Interpretation : Food deprivation diminishes plasma glucose levels and glycogen content in the liver and muscle (Knowles, 1999). The effects on muscle energy depletion at slaughter and meat quality are only visible if glycogen levels reach a certain, low threshold (55 µmol glucose equivalents/g fresh muscle tissue; Terlouw et al., 2021).
Plasma free fatty acids (FFA)	Definition : Free fatty acids (FFA) are elongated hydrocarbon chains with a terminal carboxylate group and are stored in body fat. Jointly with glucose, FFA are degraded via specific pathways to yield energy (Terlouw and Bourguet, 2022).
	Interpretation : Increased concentrations of FFA may indicate increased breakdown of fat in response to undernutrition (Shaw and Tume, 1992). When glucose levels are relatively low, plasma FFA levels, and their uptake and oxidation by the muscles increase (Romijn et al., 1995). Food deprivation increases FFA in blood (Brown et al., 1999).
Muscle glycolytic potential	Definition : Muscle glycogen can be measured alone or within the glycolytic potential (GP), which is a measure of all compounds present in the muscle that can be converted into lactic acid. The GP is an indicator of the muscle's capacity for <i>post-mortem</i> glycolysis, and, therefore, of the potential extent of muscle pH decline after slaughter (Monin and Sellier, 1985).
	Interpretation : Lower muscle GP values are indicative of muscle energy use and depletion after long-term stress due to, among others, longer period without feed, e.g. longer post-transport lairage (Rocha et al., 2015).

ABM: animal-based measure.

vii) ABMs for the assessment of the welfare consequence prolonged thirst

Table 11:	ABMs selected for the assessment of prolonged thirst in pigs during preparation, tran	sit
	and journey breaks	

ABM	Definition and interpretation of the ABM
Plasma osmolality	Definition : Measurement of the concentration of solutes in blood, used as indicator of dehydration (Clemens et al., 1986).
	Interpretation : Lack of access to water coupled with physical exercise and heat during travel causes hypertonic dehydration which is expressed in an elevation of plasma osmolarity (Parrott et al., 1987). This indicates hypertonic dehydration which occurs when water loss is proportionally greater than electrolyte loss, with water passing from the cell to the extracellular space as a compensatory mechanism (Walz et al., 2012).
Packed cell volume	Definition: The ratio of the volume occupied by packed red blood cells to the volume of the whole blood as measured by a haematocrit
	Interpretation: Increases in response to dehydration but changes are confounded by potential splenic contraction due to stress, and loss of red blood cells (Turner and Hodgetts, 1959).
Plasma total protein concentration	Definition : Concentration of plasma proteins such as albumin and globulins in blood plasma volume.
	Interpretation : A journey in hot conditions, high densities and without water intake will cause animals to lose fluids and consequently have a marked elevation of this indicator.
Intake of water after unloading	Definition: Intake of water during the initial hours after unloading, including latency to drink.
	Interpretation: Shows the transport-induced thirst threshold as a function of drinking time (Pascual-Alonso et al., 2017).

ABM: animal-based measure.



- viii) ABMs for the assessment of the welfare consequence resting problems
- **Table 12:** ABMs selected for the assessment of resting problems in pigs during preparation, in
transit and journey breaks

ABM	Definition and interpretation of the ABM
Time spent standing	Definition: Time that pigs maintaining an upright position during transport.
	Interpretation: Pigs tend to spend more time standing during a transport stage in comparison with housing conditions because of the impossibility of lying down due to lack of space and/or excessive vibration-motion of the truck.
Falling	Definition: Loss of balance, in which part(s) of the body (beside legs) are in touch with the floor (Dalmau et al., 2009; Welfare Quality [®] , 2009).
	Interpretation: Falling during transport occurs when there is a slippery floor surface, the driving style is not adequate, or animals are fatigued.
Sitting	Definition: Pigs have their hind quarters in contact with the trailer floor, and are propped up with weight on the fore legs.
	Interpretation: When space is limited, more sitting behaviour may be observed, as there is insufficient space for all pigs to lie. Sitting is a postural adjustment and space saving behaviour that has been associated with reduced space provision during transport (Urrea et al., 2021).
Lying behaviour after unloading	Definition: Pigs adopting a lying position, removing weight from their legs. Pigs may adopt a lying posture on their sternum, their lateral side, or be semi-recumbent.
	Interpretation: An increased number of pigs lying down and additionally, adopting a lying posture rapidly after unloading is associated with greater fatigue during the journey (Urrea et al., 2021). There may also be a reduction in other behaviours (drinking, fighting), with pigs prioritising lying down (Dalla Costa et al., 2016; Urrea et al., 2021).
Blood creatine kinase	Definition: Creatine kinase (CK) measured in blood (serum or plasma) is an enzyme that is increased with muscle activity and injury (Hody et al., 2011).
	Interpretation : CK has been used as a measure to indicate stress and fatigue in pigs during transport. Pigs transported with lower space allowances show raised levels of CK, suggesting increased fatigue associated with a reduction in resting and increased posture adjustments/requirements to stabilise themselves as opposed to rest (Gerritzen et al., 2013; Urrea et al., 2021).

ABM: animal-based measure.

- *ix)* ABMs for the assessment of the welfare consequence restriction of movements
- **Table 13:** ABMs selected for the assessment of restriction of movements in pigs during transit and journey breaks

ABM	Definition and interpretation of the ABM
Mounting/ overlapping	Definition : Pig mounts another pig, with its front legs on the back of the other pig (Dalmau et al., 2009; Welfare Quality [®] , 2009).
	Interpretation: Insufficient space on the ramp during loading and unloading lead to mounting and/or overlapping. Mounting is considered an aggressive behaviour showed by both male and female often related to the restriction of movement. Overlapping during transport occurs when the pigs are not able to walk and reach drinkers, and when are not able to be lying in full lateral recumbency

ABM: animal-based measure.

3.3. Preparation of pigs for transport

Across pig categories as well as journey types and journey durations, the preparation of pigs for transport may differ substantially. The fact that the place of departure may be a farm of origin, assembly centre, market or auction, depending on the type of pigs and MS, adds significantly to this variability. Regardless of the variability, preparation is an important phase in terms of animal welfare, as careful preparation of pigs for transport can substantially improve the welfare impact of transport.

For the purpose of this Opinion, the preparation phase involves all types of actions and animal management that take place during the interval from the decision to transport pigs until the initiation of loading of the pigs onto a vehicle or other means of transport. In effect, in this Opinion, the preparation of pigs for transport essentially involves the gathering of the animals to holding facilities and the keeping of them there prior to transport itself. Assessment of fitness for transport is included in the preparation stage. The loading of animals into the transport vehicle is covered in Section 3.4. For matters related to logistics, paperwork and planning, such as for example route planning, readers are recommended to check recent recommendations from the EU Transport Guides (Consortium of the Animal Transport Guides Project (CATGP, 2017) (2017-rev1), 2018). Recommendations for facilities at the various types of premises in question can also be found in the Transport Guides. These issues will not be covered in the Scientific Opinion.

3.3.1. Current practice

Below, the current practice when pigs are prepared for transport is described, including aspects relevant for weaners transported for further fattening and transport of finisher pigs to slaughter. When relevant, reference is made to cull sows and boars. The animal category cull sows is also covered in Section 3.7.

Depending on the category of pigs, farmers typically send full loads (weaners shipped to other MS, for export, or farm-to-farm journeys within the same MS, and finishers destined for slaughter) or only fractions of a load (cull sows and/or boars). This structural difference has significant effects on the preparation of the pigs for transport in terms of, for example, biosecurity. In addition, whether pigs are sent for further fattening or to slaughter has significant effects on the preparation in terms of pre-transport fasting. Another significant factor affecting preparation, and potential WCs of this stage, is whether the pigs are supposed to depart from an assembly barn, a CP, auction or similar, or whether the journey is directly from farm-of-origin to destination. In this Opinion, the use of CPs as a journey break is treated in Section 3.6.3. Irrespective of the category of pigs, or types or duration of journeys, a central part in the preparation for transport is the check of fitness for transport.

As mentioned, pigs can be collected from farms-of-origin as full loads or as smaller groups. When collected as full loads, an empty, clean vehicle can approach the building of the farm and collect the pigs from their home pens or from designated 'pick-up' pens in separate sections of the building or separate buildings housing the pigs selected for transport. When pigs are collected in smaller groups, typically vehicles will visit more than one farm on the route to the slaughterhouse. In this case, for biosecurity reasons, farmers may place the pigs in some type of vehicle or portable pen, outside the farm buildings holding the production herd, and leave the pigs there to be picked up (Figure 1).



Figure 1: Sows kept in a 'pick-up vehicle' outside farm buildings, near a public road, awaiting the pick-up by the truck taking them to the slaughterhouse. Photos: Louisa Gould, Aarhus University

3.3.2. Highly relevant welfare consequences

The highly relevant WCs during the preparation stage are group stress, handling stress, heat stress, injuries, resting problems and prolonged hunger. The latter will not occur during the preparation stage, but actions there predispose pigs to this WC at later transport stages. Below, hazards (in bold) and preventive measures (PRE), corrective and mitigating measures are described.

i) Group stress and injuries

Group stress is probably the most relevant WC for pigs during the preparation stage. Injuries are often the physical outcome of group stress in pigs, and therefore these two WCs are combined here.

Mixing unfamiliar pigs: Irrespective of subsequent journey type, as described by the EFSA AHAW Panel (2020), when pigs are sorted for transport, and mixed with non-penmates, there is a risk of increased aggression as the animals start to establish a new hierarchy. Goumon and Faucitano (2017) reviewed pre-transport on-farm procedures in terms of animal welfare of finishers and stated that the initial departure from the finishing pen is stressful as it involves alteration of the social environment through separation from the familiar group, which is one of the factors that influences aggressive behaviour.

• PRE: To prevent this hazard, pigs that are not familiar with each other should not be mixed.

The use of 'pick-up' pens (also termed shipping pens): Across the EU pick-up pens are used in some countries. Only very few studies have dealt with this type of pen, and with the pre-transport welfare of pigs in pick-up pens (sometimes called shipping pens as in Goumon and Faucitano (2017)), 'temporary lairage pens' or 'pick-up vehicles', but some studies are available on related issues such as the use of pre-sorting before pigs are sent to slaughter (Gesing et al., 2010). The use of pick-up pens implies mixing of pigs from different finishing pens which may be one of the major sources of fightingtype lesions observed on carcasses (as reviewed by Brandt and Aaslyng, 2015). Similar findings have been reported after mixing of cull sows in pre-transport pick-up pens (Herskin et al., 2020a), and during stays in stationary pick-up vehicles, considerable levels of aggression were reported (Herskin et al., 2017). These findings suggest that the use of pick-up pens may be of welfare concern. It is, however, not known whether resources can be provided to the pens, thereby reducing the level of aggression. In addition, it is not known whether the use of pick-up pens changes the total level of post-mixing aggression shown in pick-up pens plus during the journey as such. It is likely that all categories of pigs, will, at least to some extent, experience aggression when kept in pick-up pens, and thus risk being challenged in terms of animal welfare and fitness for transport in the hours before being picked up by the transport vehicle.

• PRE: To prevent the hazard of post-mixing fighting, the pick-up pen size should be adapted to the compartment size in a transport vehicle (Goumon and Faucitano, 2017), so that pigs will not need to be re-mixed when entering the means of transport. In addition, the pick-up area should be provided with proper ventilation.

Corrective/mitigating measure of group stress and injuries

If pigs are mixed, the consequences in terms of aggression and injury can be mitigated by ensuring enough space for subordinate individuals to move away from aggressive ones. This is further improved, if the place where the pigs are kept are provided with hiding places, such as half walls. In addition, the WC can be mitigated by the provision of ample resources, such as rooting materials and water, and measures to keep the microclimatic conditions within the thermal comfort zone (TCZ) of the animals. In the Scientific Opinion concerning the protection of pigs (EFSA AHAW Panel, 2022b), a pen concept including many of such mitigating measures are introduced and called a mixing pen (Figure 2).





Figure 2: Example of a mixing pen, where pigs are offered mitigating measures of group stress induced by mixing (EFSA AHAW Panel, 2022b)

ii) Handling stress

Depending on how the pigs are prepared for transport, handling stress may be a highly relevant WC, the hazards of which are:

Inexperienced, untrained or aggressive handlers: Insufficient knowledge of handling issues by the handler is regarded as the main risk for impaired animal welfare during handling of pigs.

• PRE: A crucial preventive measure is the training of handlers. Education and training leads to the necessary knowledge on the impact of a person's work on animal stress, fear, and related injuries, and on the main clinical signs to judge the state of the animals before loading. In order to move pigs, it is advantageous if staff understands the behaviour of pigs, and e.g. use the flight zone and point of balance principle. The use of flags, paddles or pig boards to move the pigs in small groups using their natural 'following' behaviour (Grandin, 2016a,b) will help to avoid the use of electric goads, which lead to fear and pain, and may lead to distress if used repeatedly.

Time pressure: The consequences of time pressure on handlers mean a risk of handling getting rougher, and pigs will be more prone to the risks of slipping, falling and skin lesions (Wilhelmsson, 2022).

• PRE: Staff have to avoid time pressure in order to move pigs in a quiet and calm way.

Inappropriate handling methods and devices: When devices to stimulate animals to move, i.e. boards and rattles, are used roughly (with excessive force, causing damage) handling stress (and potentially injuries, either as a consequence of animals falling or trying to escape) is induced. Another consequence of the rough handling is lesions that are recognisable post-mortem (Faucitano, 2001), sometimes called human-inflicted lesions (Nielsen et al., 2014).

• PRE: Handlers should be trained in non-aversive handling techniques.

Hours fasted before loading: When fasting is prolonged, pigs may be more difficult to handle at loading (Dalla Costa et al., 2016; Acevedo-Giraldo et al., 2020), and the increased frustration and agitation caused by hunger are the likely causes for these behavioural changes as an extended fasting period may cause hunger and aggressiveness (as reviewed by Faucitano, 2018). There are, however, also data suggesting that pigs fasted for 18 h were less aggressive during post-transport lairage than non-fasted pigs (Dalla Costa et al., 2016) and also studies reporting no physiological stress responses after handling of pigs with full stomachs versus fasted pigs (Faucitano et al., 2006). Importantly, the available data on effects of pre-transport fasting on behaviour and welfare of pigs, are all from studies of finishers. Whether there are effects of fasting on, e.g. aggression in cull sows – where a significant proportion is lactating and used to a high feeding level (Thodberg et al., 2019) - is not known.

• PRE: Pre-loading fasting duration should not be too long. See recommendations on fasting time in Section 3.3.2 [prolonged hunger] below.



Corrective/mitigating measures of handling stress

If aggressive or rough handling is observed, the handler should be removed from the place or receive training in situ.

iii) Heat stress

The main hazards that can lead to heat stress during this stage of transport are listed below, together with preventive measures, and corrective or mitigating measures for the WC. More detailed information on the influence of microclimatic conditions on pigs can be found in Section 3.5.3.1 dealing with heat stress during the transit stage.

High effective temperature: Heat stress in pigs is mainly due to high air temperature, but is intensified by high humidity, solar radiation (lack of shade), low air movement, and by metabolic heat. Pigs may become sunburned if placed in direct sunlight for hours in, for example, a pick-up pen or similar facility. High ambient temperature accompanied by high humidity causes discomfort and enhances the stress level.

• PRE: To prevent high temperatures, avoid putting pigs into pick-up facilities without a roof. The facilities should be designed to allow for a breeze to pass through, and the warmest hours of the day should be avoided. When stationary, pick-up pens and vehicles should be placed in shaded areas.

Insufficient pre-transport water provision: If animals have not been provided with water during the preparation stage, there is an increased risk of heat stress.

• PRE: Animals should be given *ad libitum* access to water until the moment of loading.

High stocking density: Keeping pigs in pick-up facilities at high stocking density can increase the risk of heat stress during the preparation stage.

• PRE: Under hot conditions, lowering stocking density during preparation will help prevent heat stress.

Corrective/mitigating measure of heat stress

Animals spotted before loading showing signs of heat stress, should not be loaded. Affected animals should be moved to a place with shade, and ventilation (fan) and water should be provided with no restrictions. Animals can be cooled down using water sprinklers, showers or equivalent. Animals should be inspected before re-starting the loading to ensure they are fit for transport.

iv) Prolonged hunger

Unless pigs are fasted extensively before initiation of transport, it is unlikely that they will experience prolonged hunger during this stage of transport. However, as the feeding level during the preparation stage is highly influential for the level of hunger during the later transport stages, and thus can predispose pigs for prolonged hunger there, it is also mentioned here. Corrective and mitigative measures are described in Section 3.5 covering the transit stage.

Hours fasted before loading: Regarding pre-transport feeding, quite large differences exist between categories of pigs being transported for slaughter or for further fattening. Before being sent to slaughter, pigs will typically be fasted for hours while still on-farm. As recently described by EFSA AHAW Panel (2020), fasting is done to reduce gut content and prevent release and spread of bacterial contamination through faeces during transport and lairage, as well as through the spillage of gut contents during carcass evisceration. Fasting before slaughter, within reasonable limits, can also be beneficial to animal welfare as it may prevent pigs from vomiting in transit (due to motion sickness) and from developing hyperthermia (Averós et al., 2008). Survey studies have also found increased risk of in-transit mortality in pigs that were not fasted before transport (Guàrdia et al., 1996; Averós et al., 2008), but, as reviewed by Zurbrigg et al. (2017), the use of survey type studies means that the fasting per se may not be the causal factor of the mortality as this management tool is most probably confounded by other pre-slaughter management, such as pig sorting method. The physiological mechanism as to why fasting can reduce transport mortality has not been well studied (as discussed by Zurbrigg et al., 2017).

Kephart and Mills (2005) reported that feed takes 4–8 h to be absorbed by the small intestine after ingestion, where most nutrients enter the blood 9 h post-ingestion, thereby feed eaten within the last 10 h will remain as gut-fill. In his review of pig welfare during transport, Warriss et al. (1998)

18314722, 2022, 9, Downloaded from https://efsa.onlinelibtrary.wiley.com/doi/10.2903/j.fsa.2022.7445 by Universidad De Zaragoza, Wiley Online Library on [3001/2025]. See the Terms and Conditions (https://onlinelibtrary.wiley.com/terms-and-conditions) on Wiley Online Library for nules of use; OA articles are governed by the applicable Creative Commons License

18314732, 2022, 9, Downoaded from https://efsa.onlinelibrary.wiley.com/doi/10.2903/j.sa.2022.7445 by Universidad De Zaragoza, Wiley Online Library on [3001/2025]. See the Terms and Conditions (https://onlinelibrary.wiley.com/terms-and-conditions) on Wiley Online Library for nales of use; OA articles are governed by the applicable Creative Commons License

estimated a rate of loss in body weight of ~ 0.2% of liveweight per hour. Part of this is urine and faeces, but carcass and liver weight reduction, reflecting loss of body reserves, begins 9–18 h after the last meal. Associated with the liver weight loss is a reduction in liver glycogen content. This follows a logarithmic pattern (Warriss and Bevis, 1987), so that very little remains after 24 h, over 2/3 having been lost in the initial 12 h. According to Warriss et al. (1998), it is likely that pigs are feeling very hungry at this time.

- PRE: Irrespective of whether pigs are transported for slaughter or for further fattening, preloading fasting duration should not be too long and should be appropriate for the journey duration, the pig category as well as take into consideration whether pigs are transported for slaughter or for further fattening/production. See recommendations on fasting time in Section 3.5.3.3.
- v) Resting problems

Depending on the duration of preparation, resting problems might and might not develop during this stage. However, if pigs lack the opportunity to rest during the preparation stage, they will be predisposed to develop resting problems during later transport stages (Faucitano and Goumon, 2018). Therefore, the hazards for the development of resting problems connected to the preparation stage are assessed here, whereas corrective/mitigating measures are assessed in Section 3.5 covering the transit stage.

Insufficient space: When stocking density in pick-up facilities is too high, the animals do not have the space to lie down undisturbed, increasing the risk of resting problems during preparation.

• PRE: Depending on the duration of the preparation stage, pigs should be provided with enough space to allow them to rest, as described in detailed in quantitative examination of the space requirements (Section 3.5.3.2).

3.3.3. Fitness for transport

3.3.3.1. Introduction

Throughout the scientific literature, it is agreed that – in terms of animal welfare – making sure that animals are fit for transport before departure is of utmost importance (Grandin, 2001; Cockram, 2019). However, currently no agreed scientific definition of the concept of fitness for transport exists (as discussed by Herskin et al., 2021).

Animals sent for slaughter with pre-existing conditions are more likely to die in transit, become non-ambulatory, or be condemned as unfit for human consumption upon arrival at the slaughterhouse (Lambooij, 2014; Cockram, 2019; Dalla Costa et al., 2019). Unfit pigs, i.e. pigs unable to move or keep balance, showing respiratory distress, prolapsed uterus or rectum, profuse and continuous bleeding, in late pregnancy, or having just given birth, should not be loaded for transport as these poor health conditions make them weak and with less chance to cope with extended feed deprivation and to move away from aggression. In addition, the animals will be more likely to lose balance due to sudden stops or accelerations or changes of direction of the transport vehicle (Faucitano and Raj, 2022). Based on data collected in a survey of more than 300 loads transported to three Brazilian slaughterhouses, Dalla Costa et al. (2019) reported that of the 0.38% pigs identified as non-ambulatory after transport, half of them were already in this condition at the farm before loading. The authors also calculated a 1.32 times greater risk of DOA (Dead-On-Arrival) in pigs that were non-ambulatory or presenting hernias or arthritis at loading on the farm.

Thus, if animals are not properly inspected, and unfit animals are allowed to enter the logistic chain, it is a hazard for their welfare, predisposing them to different WCs during the journey, and potentially leading to negative affective states such as discomfort, pain and suffering. Typical characteristics leading to animals being unfit for transport are related to health impairment, but some characteristics rendering animals unfit for transport do not directly relate to health, but to certain age groups or certain stages of the production cycle.

The criteria and circumstances leading to a decision of 'unfit for transport' probably vary considerably across the different categories of pigs, thereby posing extra challenges for the professionals involved in decision-making: farmers, livestock drivers, veterinarians, and competent authorities, as well as posing challenges to the welfare of pigs that may be transported despite being unfit for the intended journey.



Based on the above, it is clear that making an assessment of fitness for transport is not simple. Studies in pigs have reported doubt in the decision-making of involved professionals (Thodberg et al., 2020; Herskin et al., 2020b). In cattle, a comparison between and within three different professional groups (livestock drivers, veterinarians and farmers) regarding fitness for transport of dairy cows, showed at best moderate agreement (Dahl-Pedersen et al., 2018). Hence, in order to prevent the significant welfare hazard of allowing unfit animals to enter a means of transport, it is highly important that the inspection is carried out correctly. If the concept of fitness for transport is not well-defined, if guidelines for fitness of transport are not comprehensive and broadly available, if professionals involved are not properly educated, and if questions about responsibility are present, the risk of animals entering a means of transport as unfit will be higher.

From a scientific point of view, the concept of fitness for transport as such has received limited attention, and at present, thresholds for ABMs as indicators of animals being unfit for transport have most often not been established or validated. If pigs are to be fully protected from the consequences of being transported, while in reality unfit for transport, knowledge about the risk associated with transport of animals with a number of conditions potentially leading to negative affective states (e.g. lameness, open wounds, mastitis in sows), as well as the establishment of ABMs useful to identify these and their thresholds (suitable for use across professional groups), are needed.

3.3.3.2. Assessment of fitness for transport in pigs

To assess the fitness of pigs for transport, they should be observed to detect potential clinical signs of illness and evaluate their behaviour, posture, gait, respiration and discharges. When necessary, pigs should be caught and examined in detail. As described by the EU Reference Center for Animal Welfare of Pigs (EURCAW, 2021), the diseases that limit the coping capacity of pigs during transport are, for example, cardiac issues (pericarditis, endocarditis), respiratory diseases (pneumonia, pleurisy), locomotory disorders and infections (abscesses, metritis, nephritis, etc.) (Cockram, 2019; Ritter et al., 2020). In addition, transport- or stress-related myopathies observed in fast-growing pig breeds or lines (Zurbrigg et al., 2017; Semenova et al., 2019) may also challenge the fitness for transport of pigs.

• Conditions leading to pigs being unfit for transport

As reviewed by Cockram (2019), if, before transport, a pig has a **clinical condition that is painful**, transport will almost certainly aggravate the pain and may lead to suffering. Movement of, or pressure on, a painful area of inflammation, such as an arthritic joint, causes additional pain. Therefore, movement of body parts during loading, unloading, in response to vehicular movements or behaviour of other animals, and during postural changes, is likely to cause movement of the sensitive tissue and result in additional pain. Animals should not be transported with a non-stabilised fracture as this will cause additional pain and may lead to suffering. Bone fractures are painful; mechanical pressure applied to the fracture site, or movement and mechanical distortion of fractured bone causes pain (Herskin and Di Giminiani, 2018).

Lameness is common in pigs (Kilbride et al., 2009) and can be detected using gait scoring (as done by Fogsgaard et al. (2018) in sows prior to transport). When a pig appears lame or is reluctant to walk, it is most likely experiencing pain. Prolonged standing and efforts to maintain stability when the vehicle or other animals move, are likely to cause the condition of a lame animal to deteriorate during a journey (as documented by Thodberg et al. (2019) after transporting sows to slaughter). To be fit for transport, a pig must be able to stand, bear weight on all of its legs, and be able to adjust footing to maintain balance during the journey. The pig must also be able to walk up and down ramps at the start and the end of the journey (Faucitano and Raj, 2022).

One clinical condition that is relatively prevalent in weaners and finishers is **umbilical outpouchings** (Hovmand-Hansen et al., 2021a), often termed hernias, which is the popular name used to describe outpouchings on the ventral aspect of the abdominal wall of pigs. The outpouchings may cover several types of underlying pathology, e.g. cysts, abscesses and fibrosis (Hovmand-Hansen et al., 2021b), and true hernias (when abdominal viscera pass through weak areas in the inguinal ring or the navel without breaking through the skin) are only one type (Andersen et al., 2014). As reviewed by the EURCAW (2021), the general condition of the pigs with umbilical outpouchings vary from completely normal to un-thriving and depressed, with or without signs of pain. The mobility of the pigs might be limited by the outpouchings depending on their size and location and the underlying pathology. In addition, contact between the outpouching and the flooring (when lying or, for very big outpouchings, also when moving) may result in skin wearing off, and lead to the formation of sores. In



the worst case, the outpouching may rupture. Under transport conditions, it is likely that pigs with umbilical outpouchings are in an increased risk of injury and rupture, as compared to healthy pigs, if piling or some degree of overlying occurs in the vehicle. However, the effects of umbilical outpouchings on behaviour and welfare of pigs are not fully understood (Schild et al., 2015a,b). Hernias are described in the practical guidelines to assess fitness for transport of pigs, published as part of the European research project 'TransportGuides' (Consortium of the Animal Transport Guides Project (CATGP, 2017) (2017-rev1), 2018). According to these guidelines, outpouchings should be considered serious, regarding fitness for transport, if they are more than 15–20 cm wide and present sores. Some authors reported a greater risk of death (1.32 times more) during transport in pigs presenting hernias before loading (Dalla Costa et al., 2019). However, it is not specified whether the 15–20 cm threshold covers the span of pig live weights typically transported (from weaners to breeding boars) and no studies have successfully validated this threshold in terms of fitness for transport of pigs.

Even though several sources mention physiological weakness as a reason for not transporting animals, it is not specified how the term should be operationalised. Some characteristics rendering animals unfit for transport do not directly relate to health issues, but to certain age groups or certain stages of the production cycle (EURCAW, 2021). One such potential candidate is lactating females. Thodberg et al. (2019) focused on the proportion of cull sows sent for slaughter with engorged udders due to ongoing milk production, caused by a very recent weaning of their litters. In the days before weaning, modern prolific sows produce a substantial amount of milk (Williams et al., 2013), and this process is not stopped until days after weaning. However, the accumulation of milk in the porcine udder, and potential consequences of this, in terms of animal welfare, have not received the same scientific attention, as, for example udder engorgement post dry-off in dairy cows (e.g. Franchi et al., 2022). In addition to the possible pain and/or discomfort from the engorged udders, these sows may also be at a higher risk of developing mastitis during preparation for transport or during the journey. This was confirmed by the results of the survey carried out by Thodberg et al. (2020) in Denmark to collect opinions and experiences of livestock drivers transporting cull sows to slaughter. Some of the respondents of this survey expressed their concerns over the fitness for transport in sows with engorged udders. Whether being transported shortly after weaning is challenging for cull sows is, to the best of our knowledge, not known, but cannot be excluded. This constitutes a gap in knowledge.

Some conditions can make a pig unfit for transport because of reduced ability to perform an **important physiological function**, e.g. respiratory disorders can reduce exercise tolerance and capacity to deal with heat. Emaciated or weak animals may have reduced ability to obtain feed and water, are more susceptible to the combined effects of fasting and cold exposure and less able to respond to other animals and events affecting their stability (Faucitano and Raj, 2022). In addition, during transport, a diseased animal may feel ill (e.g. inappetence, thirst and fever), be more susceptible to extremes in the microclimatic conditions and might also be at increased risk of experiencing other negative affective states, such as fear (because of disorientation or reduced ability to respond to perceived danger) and distress (as discussed by Cockram and Hughes (2018) for sheep).

Although pregnancy should not in itself make a pig unfit for transport, **<u>in late pregnancy</u>**, there are physical and physiological changes that increase the challenges experienced during transport.

The concern for animal welfare in relation to transport of pregnant females is two-fold, and includes the pregnant female as well as the fetus/newborn:

Concerns for the welfare of the pregnant female relates to:

- i) The stress and WCs associated with the different transport stages when carrying fetuses;
- ii) The risk of going into labour or giving birth during transport; and
- iii) The risk of abortion and health consequences thereof.

Concerns for the welfare of the fetus/newborn relates to:

- i) Prenatal stress associated with being transported if the pregnant female is not slaughtered prior to farrowing;
- ii) The risk of being born during transport.

Female pigs may be transported during pregnancy to move to a different barn, or transported to a farrowing paddock (for outdoor bred females). Additionally, animals may be sent to slaughter while pregnant, be it for health, welfare or management reasons, or because the farmer is unaware that the animal is pregnant (EFSA, 2017).

Across livestock species, the biology of the species in question is likely to influence the fitness for transport of pregnant females. Sows are pregnant for an average of 114 days, and give birth to a litter of semi-precocial young (Nowak et al., 2000), and modern sow breeds selected for increased litter size and productivity traits (Rutherford et al., 2013), carry between 10 and 24 piglets per gestation.

Pregnancy is associated with changes in the cardiovascular system with an increase in heart rate and important alterations of the heart rate variability as well as behavioural changes with a decreased activity level, probably due to increased metabolic demands on the sows (Marchant-Forde and Marchant-Forde, 2004). In addition, heat stress in sows can reduce the gestation length as well as the weight of the litter (Lucy and Safranski, 2017). There is no evidence from existing literature that a relationship between stage of gestation and fitness for transport has been researched in sows or gilts. The rate of fetal growth *in utero* increases cubically (Figure 3); with a period of rapid growth from mid-gestation (McPherson et al., 2004). Fetal growth leads to an increase in heat production (Zhao et al., 2020), and consequently, sows in late gestation (measured at day 99 \pm 5) are more susceptible to heat stress than sows that are not pregnant or in mid-gestation (day 60 \pm 10) (McConn et al., 2021), which has implications for transport in the different stages of gestation. The weight of fetuses measured by McPherson et al. (2004) were 50% greater than those studied in the 1970s by Knight et al. (1977), reflecting the selection for increased sow prolificacy and high lean growth (McPherson et al., 2004).



Figure 3: Increase in fetal weight over the course of gestation in sows representing cubic growth: ($y = 0.00108 x^3 - 62.922$, $R^2 = 0.943$, where y = fetal weight (g) and x = day of gestation) (McPherson et al., 2004)

Sows transported to slaughter during gestation may be fasted before and during transport. A sow off feed may be at a greater risk for ketosis during late gestation and lactation (Alsop et al., 1994), but reporting of this condition in sows is rare. The activity level of sows decreases as gestation progresses, with a marked increase in lying within the second and third trimester (Marchant-Forde and Marchant-Forde, 2004). Similarly, sow heart rate progressively increases over the course of gestation, with a pronounced increase within the second and third trimester (Marchant-Forde and Marchant-Forde, 2004), reflecting changes to cardiac output as fetal growth progresses. Consequently, sows transported in late gestation may have lower tolerance for exercise and exertion. This lower exercise tolerance, combined with increased lethargy in late gestation may make sows more resistant to moving for loading, including walking up ramps. Experience from the pig production industry practice is that handling stress and the incline of ramps should be minimised and the use of steps requiring sows to jump into, or down from trailers, avoided (Bates, 2016). There is no available data on whether transport stress can increase the risk of abortion in sows, but industry experience suggests transporting sows in early gestation (from insemination to day 45 of pregnancy) can increase the risk of pregnancy loss (Bates, 2016; Y. Seddon, personal communication).

Several existing guidelines suggest that sows should not be transported beyond 90% of pregnancy (equivalent to 103 days of gestation) (Table 14). However, no evidence to support these guidelines have been found, as, for example illustrated from Figure 3, showing the fetal growth curve of piglets.

Fetuses may be exposed to prenatal stress, that has the potential to affect them later in life (Braastad, 1998). In sows, the evidence for the effects of prenatal stress on offspring has been reviewed by Miguel-Pacheco and Seddon (2021), and there is evidence that stress experienced in the preconception as well as during gestation can influence the behaviour and physiological responses of

porcine offspring. However, there is no published literature exploring the role of transport as a prenatal stressor on piglet welfare. Additionally, many of the studies exploring prenatal stress have provided stressors for a longer period of time than a single transport event may last. Therefore, it is currently unknown if and how maternal transport stress influences the welfare of piglets.

The EFSA AHAW Panel (2017) concluded that livestock fetuses in the last third of gestation have the anatomical and neurophysiological structures required to experience negative affect. However, regarding welfare of fetuses during transport, the EFSA AHAW Panel (2017) concluded, with 66–99% likelihood, that the neurophysiological situation of the livestock fetuses throughout pregnancy (e.g. inhibitory and excitatory systems) does not allow for perception of pain or other negative affect as long as the fetus is *in utero*. If this is correct, it means that welfare concerns for the fetus during transport most likely are of minor relevance. However, if this is incorrect, the fetuses might experience negative affective states while in utero. If this lesser possibility should not be precluded, pregnant females should not be transported within the last trimester.

Young pigs, particularly in the period immediately after weaning, have a reduced capacity to withstand the challenges of transport. They are at increased risk of hypothermia and hypoglycaemia. Young pigs may also be at greater risk of an infectious disease due to their underdeveloped immunity. These factors make them more susceptible to extreme conditions such as cold temperatures and long journeys without access to feed, water and rest.

Conditions affecting fitness for transport are listed in several regulations (Government of Canada, 2022; Council Regulation (EC) No 1/2005¹), and there are several guidance documents and decision trees available to assist in the assessment of the fitness of pigs (Consortium of the Animal Transport Guides Project, 2017; WOAH, 2011, DVFA (The Danish Veterinary and Food Administration), 2019, and Factsheets from the EU Reference Center for Animal Welfare (EURCAW, 2021)). These are summarised in Table 14.

General condition	Specific condition	References
Sickness/illness	Not specified further	WOAH
	Cardiovascular or respiratory disorders/laboured breathing	CATGP
	Dehydration risk and poor general status (profuse diarrhoea)	CATGP
Pathophysiological state	Weakness	WOAH
	Anaemia (pale skin, breathing quickly)	EURCAW
	Emanciation	EURCAW
	Circulatory weakness	CATGP
	Breathing difficulties	CATGP
	Fatigued/exhausted	WOAH
Injured	Not specified further	WOAH
	Serious wound, open or bleeding	CATGP, DVFA
	Disabled/infirmity	WOAH
	Severe, profuse nose bleeding	CATGP
	Profuse bleeding from the vulva	CATGP
	Significant pus discharge from the vulva	CATGP
	Swelling. Multiple abscesses and/or deformation and/or arthritis	CATGP
	Unhealed wounds after recent surgery	WOAH
	Severe blood loss	CATGP
Tail lesions	Severe tail lesions: Evidence of chewing or puncture wounds with swelling and signs of infection. Pigs appearing depressed.	EURCAW, CATGP, DVFA
	Partial or total loss of the tail with possible necrosis	EURCAW, CATGP
Ear lesions	 Severe ear lesions: Wounds involving a large part of the ear. Exposed ear cartilage. Inflammation/ infection deeper in the ear. 	EURCAW, DVFA

Table 14: List of conditions that can make pigs unfit for transport



General condition	Specific condition	References
	 Severe elephantiasis - the ear is so large/heavy that it limits the pig from moving normally, or from maintaining balance due to the titled head. Acute accumulation of blood ('blood-ear'). 	
Shoulder sores	Open shoulder sores, if more than 5 cm in diameter	EURCAW, DVFA
Umbilical outpouchings (including true hernias)	Severe hernias: The hernia is larger than 15–20 cm (weight depending) with sores. Irrespectively of the size of the hernia sac, the pig is not fit for transport if ability to move freely is limited or the pigs appears depressed.	Eurcaw, Catgp, Dvfa
Skin lesions	Severe lesions risk of aggravation and major haemorrhage	CATGP
	Widespread red patches (Erysipelas)	CATGP
Eye lesion	Blind in both eyes	WOAH
	Blind	CATGP
In pain	Cannot be moved/transported without causing additional suffering	WOAH
Non-ambulatory	Animal not able to walk/move	CATGP, WOAH
Lameness	Severely impaired mobility: No weight in the affected limb(s)	EURCAW, CATGP, WOAH, DVFA
	Impaired mobility: Minimum weight on affected limb(s), severely lame	EURCAW, CATGP, DVFA
	Decidedly walking-impaired due to diagnosed or supposed fracture, dislocated joints, or other painful inflammatory conditions of the limbs	DVFA
	'Tipping' quick change on weight between affected and non- affected limb(s)	EURCAW
Prolapse	Uterine prolapse	EURCAW, CATGP
	Irreversible rectal prolapse	EURCAW, CATGP
Pregnancy	Final 10% of gestation period	Eurcaw, Catgp, Woah
Recent birth	Given birth in the previous week	EURCAW, CATGP
	Given birth in previous 48 h	WOAH
New born	Unhealed navel	EURCAW, WOAH
	Less than 3 weeks old (unless < 100 km)	EURCAW

Wording used in source material has where appropriate been modified for consistency and clarity

- CATGP: Consortium of the Animal Transport Guides Project (2017).
- WOAH: WOAH (2011).
- EURCAW: EU Reference Centre for Animal Welfare Pigs factsheets: Fitness for transport of pigs.
- DVFA: The Danish Veterinary and Food Administration guide (2019).

3.3.3.3. Transport of animals with reduced fitness

Some guidelines and regulations list conditions that would consider pigs vulnerable, but fit for transport, as long as mitigation measures are used, e.g.

- a) the pig is isolated;
- b) the pig is individually loaded and unloaded without having to negotiate any ramps inside the conveyance;
- c) measures are taken that are necessary to prevent the animal's suffering, injury or death during loading, confinement, transport and unloading; and
- d) the animal is transported directly to the nearest place, other than an assembly centre, where care or euthanasia can be applied.

Examples of additional measures that could be used to reduce the risk of suffering by animals of reduced fitness during transport include: increased contingency planning, reducing journey duration, adjusting ventilation, increased bedding, avoiding extreme weather conditions, avoiding loading via



steep ramps, loading last and unloading first, providing space to lie down, increasing monitoring frequency or providing feed, water and rest more frequently. However, the effectiveness of mitigation measures to avoid the additional suffering likely to be associated with the transport of an animal with reduced fitness is questionable. There are also differing opinions on the types of conditions that would make vulnerable pigs fit for transport, even when additional mitigation measures are used. If animals of reduced fitness are transported, they are likely to continue to experience pain and discomfort, there is a risk of deterioration of the animal during the journey, and pre-existing conditions are likely to be aggravated by transport (as described for cull sows by Thodberg et al., 2019). This constitutes a gap in knowledge.

3.4. Loading/unloading

3.4.1. Current practice

Due to the considerable physical and physiological efforts required, loading and unloading have been considered stressful steps along the pork production chain (Goumon and Faucitano, 2017). In particular, loading is preceded by potential stressful experiences, such as disruption of the social group, mixing with unfamiliar conspecifics and handling by unknown staff (Faucitano and Lambooij, 2019). Unloading follows the arrival to another farm, CP or to the slaughterhouse. The condition of pigs at this stage depends on several factors related to the previous journey, i.e. duration, stocking density, temperature and humidity inside and outside the vehicle, road and driving conditions, and depend on the category of pigs as well. Unloading should be carried out as soon as possible to avoid heat and humidity increasing in the stationary vehicle, and exposure to noxious gases (CO_2 and NH_3), unless vehicles are mechanically ventilated. Delay might expose pigs to new hazards leading to additional WCs (e.g. heat stress).

The most common device for loading and unloading pigs at the farms and CPs is the ramp, even if some trucks are equipped with a loading tail-gate lift or semi/fully hydraulic decks (McGlone et al., 2014). At the slaughterhouse, a ramp or bridge is the common device used for unloading, with the latter being more present in large slaughterhouses to avoid any difference in level from the vehicle to the dock.

The common tools for moving pigs at loading and unloading are plastic paddles, boards and flags. Codes of practice such as the 'Guide to good practices for the transport of pigs' (Consortium of the Animal Transport Guides Project, 2017) recommends avoiding electric goads for pig handling, suggesting that the use of paddles and boards are more efficient (Correa et al., 2010).

To minimise the biosecurity risks represented by the vehicle, often the ramps are located out of piggery's perimeters or outside the farm area, just at the connection with the main road. In the latter cases, pigs are moved by internal vehicles (Herskin et al., 2020b) and external trucks arrive from the outside. Irrespectively of the ramp location, the physical effort required to climb the ramp should be reduced as much as possible. Ramps no steeper than an angle of 20° , that is 36% to the horizontal, have been suggested (Council Regulation (EC) No $1/2005^{1}$). Moreover, where the slope is steeper than 10° , that is 18% to the horizontal, fitting systems, such as foot battens, will ensure that the animals climb or go down without risks or difficulties, effectively reducing the sharpness of the ramp. According to Dalla Costa et al. (2019), the risk of DOA increases by 4 times when the ramp slope is higher than 20° compared to a ramp slope lower than 20° .

3.4.2. Highly relevant welfare consequences

The highly relevant WCs during the loading and unloading of pigs are: handling stress, heat stress and sensory overstimulation. The selected ABMs for the assessment of these WCs are shown above in Section 3.2.2. Below, hazards, preventive and corrective/mitigating measures are described.

i) Handling stress

Most of the hazards contributing to the risk of handling stress during the loading and unloading of the animals are identical to those present during the preparation stage (i.e. inexperienced handlers, time pressure, inappropriate handling methods and devices) (Section ii). In addition, inappropriate loading facilities and inadequate loading ramps are specific hazards during the loading and unloading.

www.efsa.europa.eu/efsajournal



Inappropriate facilities: Poor design of loading and unloading facilities increases the difficulty of the animals to access the vehicle, increasing the risk of slipping and falling. Slippery floor surfaces may result in loss of balance. The consequent slips or falls aggravate the physical effort and lead to an increased risk of skin lesions and injuries.

PRE: Loading and unloading areas have to be well designed with solid walls, appropriate non slippery flooring, presence of foot battens and lateral protection so that they facilitate calm and efficient handling. To ease handling, pigs should not have blind corners to negotiate, either when entering/exiting the vehicle or on the unloading ramp (Faucitano and Pedernera, 2016). As reviewed by Dalla Costa et al. (2016), A greater heart rate, balking behaviour and unloading time have been reported in pigs dealing with angles or bends of 45° to 90° compared with 0° and 30° (Warriss et al., 1991; Goumon et al., 2013; Herskin et al., 2020b).

Inadequate loading ramps: Inadequately designed ramps increase the risk of slipping, falling, bruises and additional physical stress to the animals. Keeping in mind that pigs have difficulties in ascending and descending a slope (Brown et al., 2005), ramps steeper than 20° are not recommended, as they result in greater increases in pigs' heart rate, vocalisation and backing up behaviour, increased intervention by the handler (eventually shouting and use of electric goads) and increase the time required for unloading by a factor of 3–4 (Warriss et al., 1991; Ritter et al., 2008; Goumon et al., 2013; Torrey et al., 2013a,b; Faucitano and Pedernera, 2016).

PRE: Faucitano and Pedernera (2016)) explained that, when compared to ramps, the use of hydraulic lifts or decks reduces handling stress, increases the easiness of handling and shortens off-load time. Current guidelines do not recommend ramps steeper than 20° to be used for fixed ramps (Warriss et al., 1991) or than 25° for adjustable ramps. However, as finishers have become heavier and more difficult to move (Bertol et al., 2005; Bertol et al., 2011; Rocha et al., 2016), recent works have recommended the maximum ramp slope to be reduced to 15° (Grandin, 2012; Faucitano and Goumon, 2018).

Corrective/mitigating measures of handling stress

Corrective measures include removal of the specific person employing inappropriate handling or provision of assistance to him/her on the spot.

If an animal refuses to move, the following procedure should be applied: behave calmly, and let the animal calm down, check the animal is not sick, wounded or unfit for transport. If it is, remove it from the group and decide on the course of action (Consortium on the Animal Transport Guidelines Project, 2017).

ii) Heat stress

The hazards that can lead to heat stress during the loading and unloading of pigs are described below, together with preventive measures, and corrective or mitigating measures for the WC. More detailed information on the influence of high effective temperature on pigs can be found in Section 3.5.3.1, assessing microclimatic conditions during the journey.

High effective temperature and solar radiation: High ambient temperature and humidity may exacerbate the efforts required to climb or descend the ramp, and also the effort required to stay in the vehicle until departure. Direct solar radiation leads to an increase of temperature, worsening the environmental conditions. The application of water sprinkling in a stationary truck under warm ambient conditions resulted in the reduction of drinking behaviour in lairage, physical fatigue at slaughter (lower blood lactate levels) and drip loss in meat (Fox et al., 2014; Nannoni et al., 2014). However, water sprinkling combined with insufficient ventilation can also result in increased difference in humidity levels (up to + 7.5%) between the vehicle interior and the external environment (Fox et al., 2014), preventing efficient evaporative cooling in pigs. Further research is needed in this area before practices can be recommended.

• PRE: Avoid loading/unloading pigs during the hottest hours of the day to avoid extreme temperatures, especially when vehicles are not equipped with mechanical ventilation. When possible, the vehicle should be parked in an area that provides shade and allows for a breeze to pass through the sides, and the loading ramp should be opened (Consortium on the Animal Transport Guidelines Project, 2017). Pigs should be protected from direct sunlight during loading/unloading. If ambient temperature is above 23°C, it has been shown that water sprinkling of pigs for 5 min in the vehicle before leaving the farm and before unloading at the slaughterhouse reduced heat stress.


Delays to loading and unloading: When the temperature and humidity are high, any delay in loading or in unloading can aggravate heat stress. Moreover, in pig transport managed under US industry guidelines, and transporting finishing pigs at ~ 275–300 kg/m², internal temperature (quantified at pig level) in passively ventilated stationary vehicles was reported to increase up to 3–4°C in 5 min (Xiong et al., 2015). Each 1°C and 1% increase in the external ambient temperature and relative humidity was reported to increase internal vehicle temperature by 0.99°C and 0.11°C, respectively (Dewey et al., 2009). In certain compartments of a passively-ventilated truck kept stationary for 0.5 h, the temperature was found to be between at least 6 and 10°C higher than the external ambient temperature (Weschenfelder et al., 2012; Faucitano and Goumon, 2018). This has only been shown in few truck types, with certain space allowances and for certain pig categories, of which the majority were not European, and it is thus possible that the difference may be higher or lower under European conditions. Research is needed to clarify changes in temperature in stationary vehicles under European conditions.

• PRE: The stage of loading has to be organised to ensure that the vehicle starts moving as soon as possible, and, if possible, loading should be carried out at a time of day to avoid heat stress.

Physical effort: The forced climbing to the vehicle on inclined ramp demands substantial muscular exercise which most pigs are not used to. This exercise leads to an increment of respiration, body temperature, and heart rates.

• PRE: Ramps should be properly designed with ideally no greater than 15° slope.

Stocking density: Large groups of pigs collected and forced to move in a restricted passageway, like the ramp, or forced to move out from the vehicle, are very difficult to handle because of fear from narrow passageways or corridors as well as fear of novelty.

• PRE: The pigs should be moved in small groups using their natural 'following' behaviour (Grandin, 2016a,b).

Time off water: If pigs do not have access to water (or for some reason fail to obtain water) during preparation, dehydration will increase the risk of heat stress during loading.

• PRE: The provision of water before loading and after unloading can prevent the development of heat stress. Thus, pigs need to have access to water on farm until being loaded on the vehicle.

Corrective/mitigating measures for heat stress

In case of heat stress in any pig during loading/unloading, the identified animal should be removed from the area, moved to a place with shade and ventilation (fan) and provided with water. Animals can be cooled down using water sprinklers, showers, or equivalent, if the ventilation permits exchange of humidity.

iii) Sensory overstimulation

The main hazards that can lead to the sensory overstimulation at this stage of the journey are listed below.

Sound/noise: Unfamiliar noises from the vehicle, or shouts from the handlers can increase the risk of sensory overstimulation.

• PRE: Any unnecessary noise should be avoided and, a calm loading area should be selected.

Lighting: Sudden change or inadequate lighting (high contrast with bright and shaded areas) during loading/unloading may induce fear and slow down handling.

• PRE: A strong contrast of light should be avoided as well as direct sunlight in the eyes of the animals.

Corrective/mitigating measures of sensory overstimulation

If pigs experience sensory overstimulation during loading/unloading, the environment should be checked to identify the sources and remove them (i.e. switch off the machinery creating the unfamiliar noise, modifying the lights).



3.5. Transit stage

3.5.1. Current practice

After loading onto the vehicle, the animals are transported to another farm or to the slaughterhouse, sometimes passing CPs or assembly centres on the way. Often the term journey is used to encompass all management and actions taken from the place of departure to the destination including any unloading, accommodation and loading that occurs at intermediate points in the journey. However, in order to be able to separate the assessments of animal welfare during the part of the transport practice where the animals are being moved from breaks in the movement (e.g. during stays in CPs), in this Scientific Opinion, the transit excludes loading/unloading and journey breaks, and starts when the ramp has been closed and ends when the first animal unloads or the vehicle parks for a break in the journey aimed to rest, feed or water the animals (as defined in Section 1.2).

There is a diverse range of vehicles used to transport pigs, of which many are trucks with multiple decks (see Figure 4).





Figure 4: (A) Three-deck pig truck collecting cull sows from a pick-up vehicle in Denmark. Source: Thodberg et al., 2019; (B) Four-deck truck with 5-deck trailer transporting weaners. Photo: Aarhus University

3.5.2. Highly relevant welfare consequences

Welfare of pigs during the transit stage of transport depends on several factors such as journey duration, space allowance, ambient conditions (temperature, humidity, vibrations, and noise) vehicle design and driving conditions (Schwartzkopf-Genswein et al., 2012; Faucitano, 2018).

The WCs selected as highly relevant during the transit stage of pig transport are: group stress, heat stress, injuries, motion stress and sensory overstimulation, prolonged hunger, prolonged thirst, resting problems and restriction of movement. The ABMs used to assess each WC have been defined above. The hazards leading to the WCs are identified below as well as preventive measures and mitigating or corrective measures.

i) Group stress and injuries

Group stress is a highly relevant WC for pigs during the transit stage. Injuries are often the physical outcome of group stress in pigs, and therefore these two WCs are combined here. The main hazard contributing to group stress and injuries is the mixing of pigs that are not familiar to each other.

Mixing unfamiliar pigs: It takes several days after mixing for pigs to establish a new social hierarchy. This means that any pre-journey mixing increases the risk of aggression and injuries as a result of fighting taking place during the transit stage.

18314722, 2022, 9, Downloaded from https://efsa.onlinelibtrary.wiley.com/doi/10.2903/j.fsa.2022.7445 by Universidad De Zaragoza, Wiley Online Library on [3001/2025]. See the Terms and Conditions (https://onlinelibtrary.wiley.com/terms-and-conditions) on Wiley Online Library for nules of use; OA articles are governed by the applicable Creative Commons License

• PRE: To prevent this hazard, pigs that are not familiar with each other should not be mixed at any stage during transport. In order to allow this to happen, compartment size should be adjusted to on-farm group size.

Corrective/mitigating measure of group stress

If pigs are mixed, the consequences during the transit stage are difficult to mitigate, as it is natural behaviour of newly mixed pigs to fight. However, the level of group stress during the transit stage may possibly be mitigated by avoiding competition for resources such as water and lying places, and to avoid exposing the pigs to other hazards that may induce fighting.

ii) Heat stress

The hazards contributing to the risk of heat stress during the transit stage are broadly similar to those in the loading/unloading stages; namely high effective temperature and solar radiation, physical effort, stocking density and time off water. In addition, the ventilation rate and high temperatures inside vehicles contribute to increase the risk and severity of heat stress. The main hazards contributing to heat stress together with their preventive measures and mitigative and corrective measures of the WC are described below. More detailed information on the microclimatic conditions recommended for the transit stage is included in Section 3.5.3.1 assessing thresholds for microclimatic conditions.

High effective temperature: High effective temperature inside the vehicle (dry temperature combined with high humidity) is the major hazard for heat stress. In vehicles equipped with natural as well as mechanical ventilation, the major determinant of the microclimatic conditions inside, is the outdoor climatic conditions. The temperature and humidity outside the vehicle dictate the temperature and humidity of the air coming into the vehicle via air vents. In the absence of air conditioning, the effective temperature in the vehicle can never be lower than that outside. Solar radiation can also play a significant role by heating the vehicle over and above the effective temperatures seen outside.

• PRE: To prevent this hazard, animals should be transported in their TCZ, as explained below in Section 3.5.3.1, where the upper threshold of the comfort zone and the upper threshold of the thermoneutral zone (TNZ) (the upper critical temperature, UCT) are identified as the basis of a quantitative recommendation on microclimatic conditions during transport of different categories of pigs. If outside temperatures are higher than the TCZ, the hazard can to some extent be prevented by avoid transporting animals during the hottest hours of the day.

Solar radiation: Solar radiation on the roof and walls of the vehicle causes the interior of the vehicle to heat up.

• PRE: In construction of vehicles, radiation-reflecting material should be considered to reduce the risk of heat stress.

Low ventilation rate: Ventilation replaces the air in the vehicle with air from outside the vehicle.

• PRE: To reduce the risk of high effective temperatures inside the vehicle, appropriate ventilation should be ensured. This will remove metabolic heat from the animals and also help to avoid concentration of gases and fumes, thereby also reducing the risk of sensory overstimulation.

Stocking density: Lowering space allowance increases the number of pigs that can be loaded onto a vehicle and the amount of metabolic heat and moisture that they produce will increase.

PRE: Increasing space allowance can help to reduce the risk of heat stress, by allowing the animals room to lie on their side without touching each other. As described in detail in Section 3.5.3.2, whether increasing the space allowance as such can contribute significantly to the microclimatic conditions inside the vehicle is uncertain.

Time off water: Described in Section 3.4.2 [ii heat stress].

Corrective/mitigating measure of heat stress

When animals show signs of heat stress, they should be unloaded immediately, moved to a place with shade and ventilation (fan) and provided with water. Animals can be cooled down using water sprinklers, showers or equivalent, if the ventilation permits exchange of humidity.

If unloading cannot be done immediately, it is recommended to complete the journey as soon as possible, increasing ventilation if possible, to provide convective air cooling. Avoid stopping the vehicle when ventilation is restricted.

iii) Motion stress and sensory overstimulation

The two WCs, motion stress and sensory overstimulation share mediating biological mechanisms, ABMs and to some extent also hazards. They are, thus, treated in combination during the transit stage of transport. The main hazards leading to these WC in pigs during the transit stage are listed below, and the ABMs have been described in Section 3.2.2.

As reviewed by Valadez Noriega and Miranda de la Lama (2020), during transport, animals are exposed to vertical, lateral and horizontal vibrations. Unpaved roads or roads with strong wind currents transmit a more significant amount of vibrations. An animal's sensitivity increases after long standing periods (Gebresenbet et al., 2011), causing fatigue and displacement of their centre of gravity, which may lead to falls and injuries (Bulitta, 2015). Transport motion research has focused on road transport and there is, thus, limited research on for example air transport motion, even though it has potential to stimulate the vestibular system and produce body instability, both precursors of motion sickness (Santurtun and Phillips, 2015). The vestibular system detects motion and gravity and initiates movements to maintain balance and orientation. Further research is needed to investigate physiological and behavioural responses to determine responses to this condition, leading to a better understanding of how motion sickness could affect livestock during transport (Santurtun and Phillips, 2015).

During transport, animals experience stress and/or fatigue due to the forces exerted as a result of acceleration, braking, stopping, cornering, gear changing, vibrations and uneven road surface. Vibration is the movement of a body about its reference position, and occurs because of an excitation force that causes motion. Vibration has been shown to alter animal behaviour and induce physiological changes as well as to cause effects at the cellular and molecular levels. For these reasons, vibrations have a considerable potential to alter animal welfare status (Reynolds et al., 2019). Vibratory movement has a direction (generally in three planes) (Figure 5), a magnitude (how far) and a velocity (how quickly – what rate).





Rolling and pitching of vehicle: While the vehicle is moving, all pigs are to some extent exposed to motion stress, which may lead to motion sickness in this animal species. The stress is increased during driving events, such as acceleration, braking and turns. Rural roads are a mixture of unsurfaced, paved and surfaced roads. The former two increase the transmission of vibrations to the animals compared to larger roads and, when waterlogged, can cause the vehicle to lose stability, and the animals to lose their balance.

• PRE: Among the preventive measures for this hazard are planning journeys on motorways, with vehicles equipped with good suspension and driven by experienced and skilled drivers. In addition, vehicle vibration can be reduced through a suspension system which, in the case of freight trucks, can be either leaf spring or air suspension. Both suspension systems improve vehicle contact with the road surface and indirectly reduce vehicle vibration (Dalla Costa et al., 2017).

Rough driving: Rough driving, which includes high acceleration and forceful braking, may cause postural instability, leading to toppling, sliding and excessive corrective muscular action, resulting in bruising, muscular fatigue, fear and general injuries to animals during transport (Driessen



et al., 2020a,b). The proportion of pigs standing or lying associated with an increase or decrease in acceleration was examined by Peeters et al. (2008) who studied the combined effect of driver and driving style on the behaviour of pigs and other physiological indicators during a short journey. In addition, the effect of differing accelerations (longitudinal, lateral and vertical) of the trailer on these variables was studied. Increasing acceleration resulted in increased proportion of pigs standing during the journey and a decrease in the proportion of pigs lying down. Measurements of variability in heart rate revealed that lateral acceleration was an important stressor for pigs. Randall et al. (1996) recommended a good air suspension on all axles of the vehicle to reduce vibrations. Driessen et al. (2008) examined the effect of vibration on cardiac activity and lying behaviour in 90 pigs which were vibrated in the vertical direction in a transport simulator with a frequency of 8 Hz and an acceleration of 3 m/s^2 . The effect of vibration increased heart rate. More recently, Morris et al. (2021) collected and interpreted three-axis acceleration, temperature and relative humidity data from six locations within commercial transport trailers transporting finisher pigs. Based on the data collected, the greatest vibrations (both horizontal and vertical) occurred in the bottom rear compartment which may potentially expose pigs to uncomfortable transport experience that could result in discomfort or fatique.

• PRE: Adequate driving style should be ensured to limit acceleration, braking, and cornering which affect the ability of the animals to maintain postural stability. The route should be well planned with a view to avoiding rough and winding roads which may exacerbate WCs due to motion stress.

Excessive stimuli (lights, odours, noise, etc.): During a journey, animals are exposed to numerous stimuli, including noise originated from traffic or from mechanical ventilators. According to Talling et al. (1998), the sound level experienced by pigs during transport (88 to 96 dB) is almost as high as what they are exposed to in the peri-mortem area at the slaughterhouse (89–97 dB). Sounds higher than 70 dB are deleterious to pig welfare since they induce an increase in heart rate, interpreted as fear, making the animals more fearful and difficult to move (Goumon and Faucitano, 2017). In addition, improper design of the vehicle or of compartments holding animals may lead to the risk of introducing exhaust fumes into the space destined for the animals. Improper ventilation without the removal of noxious gas (such as NH₃) produced by the animals and their excreta, leads to overexposure followed by clear signs such as watering eyes, nasal discharge and coughing, retching and ocular/vision disorders. In addition, sudden or inadequate lighting (high contrast with bright and shaded areas) during the journey will have a negative effect on pigs and may lead to stress and fear.

• PRE: The use of an adequate vehicle should reduce hazards such as excessive noise as well as the possible introduction of exhaust fumes into the space destined to the animals.

Space allowance and conditions inside the vehicle: The maintenance of balance in transit involves great physical effort (Terlouw et al., 2008). Therefore, during the journey, pigs should have enough space, and of adequate quality in terms of surface texture, dryness and hygiene, to lie down, stand up, and turn around without hindrance (Velarde et al., 2021).

• PRE: Pigs should have enough space and of adequate quality in terms of surface texture, dryness and hygiene to lie down and get up again.

Pigs that have eaten a large meal before start of a journey: Motion sickness has been described in several species, including pigs (Randall and Bradshaw, 1998; Santurtun and Phillips, 2015). Pre-transport fasting has been suggested to decrease the likelihood of motion sickness (Randall, 1992; Bradshaw et al., 1996a). However, the scientific evidence for an optimal fasting time regarding motion sickness is sparse. Warriss et al. (1998) reviewed the available literature, and concluded that the minimal interval is unclear, but suggested that, at least on vehicles showing poor vibrational characteristics or when transported on rough roads, on-farm fasting of ~ 12 h may be preferable for finishers in terms of in-transit mortality. The mortality rate of transported finisher pigs that have been fasted 8–18 h before loading has been reported to be significantly lower than that of pigs loaded on a full stomach (Warriss and Brown, 1994; Guàrdia et al., 1996). According to Averós et al. (2008), the mortality risk of non-fasted pigs increases with transport up to 8 h.

The studies mentioned above focused on in-transit mortality as the outcome. From a production point of view, pig mortality is unwanted, and it is likely that pigs dying in transit experience negative WCs during their last hours alive. However, even though periods of fasting may be beneficial in terms



of mortality and associated WCs, fasting pigs may experience other WCs due to the hours off-feed, such as hunger, group stress (as reviewed by Faucitano, 2018) and handling stress (Dalla Costa et al., 2016). According to Warriss et al. (1998), it is likely that pigs are feeling very hungry at 12 h after their last meal. Thus, the decision on pre-transport fasting is likely a compromise between several concerns related to animal welfare.

Importantly, the available data on effects of pre-transport fasting on behaviour and welfare of pigs are all from studies of finishers. The relationship between hours off feed and the risk of developing motion sickness (and other WCs such as prolonged hunger or group stress associated with the fasting), in cull sows (where a significant proportion is producing milk at the time of slaughter and thus used to a high feeding level (Thodberg et al., 2019)) is not known. This constitutes a gap in knowledge.

For pigs transported for further fattening or breeding, fasting can be beneficial to avoid motion sickness, but concerns about hygiene at slaughter are not relevant. For these categories of pigs no studies have identified pros and cons of pre-journey fasting, examined effects of fasting duration or established guidelines, for example taking into account different types of feed (wet feed vs dry feed fed from feed stations), or potential post-transport consequences of the fasting in terms of health impairment (e.g. the development of gastric ulcers). This constitutes a gap in knowledge.

• PRE: In order for pigs not to be at risk of developing motion sickness caused by full stomachs, pigs should not be able to eat a full meal shortly before journey initiation. See Sections 3.3.2 [iv prolonged hunger] in preparation and Section 3.5.2 [iv prolonged hunger] during the transit stage for further assessment of the use of pre-transport fasting in pigs.

Corrective/mitigating measures of motion stress and sensory overstimulation

If the severity of the motion stress and sensory stress is too high on the animals, the driver can stop the vehicle, unload the animals and give them some rest. Otherwise, the best option would be to finish the journey as soon as possible.

iv) Prolonged hunger

Prolonged hunger is regarded as a highly relevant WC during the transit stage. The prevalence is expected to be very high, as pigs – especially animals sent for slaughter – most often have been fasted before the initiation of the journey. The severity of hunger will develop further during the transit stage. The duration of hunger depends on journey duration and availability of feed in transit, and severity is expected to increase with increasing duration, as the need for feed becomes problematic for the animals. Prolonged hunger may lead to frustration, exhaustion and a weakened condition. The main hazard leading to prolonged hunger during the transit stage is time off-feed. Additional information on the effect of duration of food deprivation on hunger of pigs can be found in the assessment of journey duration (Section 3.5.3.3).

Time off feed: Feed deprivation can result in prolonged hunger.

- PRE: To prevent prolonged hunger, fasting duration at farm has to be appropriate for the journey duration. See Section 3.5.3.3 for recommendations on journey time for the different pig categories.
- Careful planning of the transport time, scheduling and prioritising the slaughter of the animals (if destined for slaughter) should be done to avoid prolonged periods with food withdrawn.

Corrective/mitigating measures of prolonged hunger

To mitigate the WCs of prolonged hunger, the journey should be terminated as soon as possible. Upon arrival at a slaughterhouse, pigs showing signs of prolonged hunger should be prioritised for slaughtering or fed in lairage. Feeding the pigs in the moving vehicle is not recommended as they will be at risk of developing motion sickness, and due to increased risk of aggression.

v) Prolonged thirst

www.efsa.europa.eu/efsaiournal

Prolonged thirst is regarded as a highly relevant WC during the transit stage. The prevalence may be high, if water is not provided to the animals, or if they for some reason (e.g. due to lack of familiarity with drinking devices, neophobia or fear of other animals) are not able to obtain enough water. Depending on factors such as time off water before journey start and/or microclimatic conditions, pigs may not be thirsty during the initial phase of the journey, but thirst will develop over time, if they are not drinking as much as they need. The duration of prolonged thirst depends on time



off water, and severity is expected to increase with increasing journey duration and heat, as the need for water becomes problematic for the animals. Prolonged thirst may lead to dehydration, discomfort and suffering. Main hazards for this WC are listed below, and additional information is included in the quantitative assessment of the journey duration (Section 3.5.3.3).

Time off water: If no water is accessible in the vehicle compartments, animals experience water deprivation, in the period from when they leave the last water source and until they again have access to water.

PRE: To prevent this hazard, pigs need to have access to water on farm until being loaded on the vehicle. Animals have to be prepared carefully to be transported, i.e. in reference to fluid balance. There is no demonstration of advantage in water deprivation prior to transport. As examined in Section 3.5.3.3, the evidence for the welfare benefits of providing pigs with water during the transit stage is currently limited, and further research is needed in this area to develop effective preventive measures.

Limited space allowance affecting possibility to thermoregulate: During the journey, insufficient space for all pigs to lay down in lateral recumbency will deny them the possibility for thermoregulation and might exacerbate heat stress and thirst.

• PRE: Sufficient space allowance should be provided for the pigs to lay down during transport. Additional quantitative information on space allowances can be found in Section 3.5.3.2.

Inadequate/not working drinkers: Even when transported under condition where water is provided to pigs during the transit stage, dehydration may occur if the drinking devices are not designed and positioned in an appropriate way considering the category and the size of pigs and if they are not available to all pigs held in the compartment. On a vehicle drinkers maintain the water cleaner than troughs or buckets.

• PRE: The driver must ensure to store (and/or replace) sufficient water in the vehicle's tanks and to check the function of the water supply system on the decks. Drinkers should be readily available at the appropriate angle and height for the animals.

Reduced intake of water: Only few studies have examined the intake of water during pig transport. If pigs drink less than they need during a journey, they might develop prolonged thirst even though drinkers in principle can be in place. See Section 3.5.3.3 for an examination of this hazard in relation to journey duration.

 PRE: If pigs drink less water than required during transit, this hazard cannot be fully prevented during the journey phase, not even by giving access to drinkers on vehicles. The only preventive measure, then, is to provide free access to water before loading, and to limit journey duration, so that prolonged thirst will not develop. If journey breaks are involved, pigs should be watered there.

Reduced accessibility of water: As discussed further in Section 3.5.3.3, reduced accessibility of the water may be one hazard for the development of prolonged thirst during the transit stage, even in vehicles equipped with drinking nipples.

• PRE: Only few studies have examined the intake of water during pig transport. Among factors of importance for this hazard are design of drinkers (height, position, angle, type), lack of familiarity, lack of space to manoeuvre inside vehicles and effects of the behaviour of the other animals. Currently, no studies have documented drinking systems and management thereof to prevent the hazard of reduced accessibility. This constitutes a gap in knowledge.

High effective temperature in vehicle: Too high effective temperature may have a cumulative effect on time for water deprivation.

• PRE: To prevent this hazard, temperatures inside the vehicle should be kept within the TCZ, as described in detail in quantitative examination of microclimatic conditions (Section 3.5.3.1).

Corrective/mitigating measures of prolonged thirst

In the case that signs of prolonged thirst are observed during the transit stage, the vehicle should be stopped, and all pigs given the opportunity to drink water, providing enough time for this, and



sufficient quantities of drinkers for the number of animals in the group. At CPs, pigs must be watered and fed, to recover from thirst and to have enough energy to continue the journey.

vi) Resting problems

Resting problems are regarded as a highly relevant WC during the transit stage. The prevalence is at least moderate, as resting problems may affect a large proportion of animals in a moving vehicle, depending on factors such as the driving conditions and the space allowance. Duration of resting problems depends on the journey duration, and severity is expected to increase with increasing duration, as the lack of resting becomes problematic for the animals. Resting problems may eventually lead to fatigue. Below, the main hazards are identified and preventive, corrective and mitigating measures suggested. A more thorough assessment of resting problems during the transit stage can be found in Section 3.5.3.3.

Vehicle motion and poor driving conditions: The movement of the vehicle, and especially when driving is not done in a calm and focused way, and when road conditions are of lower quality, pigs have to make continuous postural adjustment to acceleration changes, maintain their balance and to avoid falling (see above and in Section 3.5.3.3 for examination of motion stress).

• PRE: A series of preventive actions (see above for the examination of motion stress as well as in Section 3.5.3.3) can be initiated to lower vehicle motion and thereby to reduce resting problems.

Insufficient horizontal space (space allowance): When the stocking density is too high, the animals do not have the space to lie down and get up again, increasing the risk of resting problems during the transit stage.

• PRE: Pigs should be provided with enough space to allow them to rest, as described in detailed in quantitative examination of the space requirements (Section 3.5.3.2).

Unsuitable floor and inadequate bedding: Poorly maintained or poorly designed corrugated patterns of flooring do not allow animals to comfortably maintain balance during the transit stage, predisposing them to slipping, falling and constant changes of position. Anti-slip materials should be used. In addition, without adequate bedding (type and/or quality and/or quantity) animals will be less motivated to rest lying down during journeys (Goumon et al., 2013) and may be exposed to slips, falls, fatigue and weakness.

• PRE: Sufficient bedding (in terms of quality and quantity) should be provided for the journey, made of adequate materials such as sawdust.

Insufficient vertical space: A deck height forcing animals to adopt abnormal positions during the transit stage is problematic. In addition, low deck height can be associated with (1) reduced ventilation; (2) lack of ability to move around; and (3) lack of space for natural movements, and should be prevented in order to avoid WCs such as heat stress and restriction of movement. Information on the minimum height of the decks is provided in Section 3.5.3.2 where a quantitative assessment is done for the vertical space required by pigs during transport.

• PRE: In order to prevent resting problems induced by insufficient vertical space, the deck height should allow animals room to move around and for natural movements. See Section 3.5.3.2 for a quantitative assessment for the vertical space required by pigs during the journey.

Mixing unfamiliar animals: Mixing unfamiliar pigs exacerbates aggressiveness and can therefore increase the risk that pigs are prevented from resting during journeys.

• PRE: Making sure that all pigs in a vehicle compartment have been together for at least some days before transport, reduces the risk of aggressive interactions and fighting during the transit stage.

Corrective/mitigating measures of resting problems

In extreme situations where severe resting problems have been identified, pigs can be offloaded and allowed to rest. Otherwise, the journey should be finalised as soon as possible and upon arrival animals should be given enough space and calm environment to rest.



vii) Restriction of movement

The main hazard leading to the restriction of movement of pigs during the transit stage is the provision of insufficient space for the animals – horizontal as well as vertical space. This topic is covered in detail in Section 3.5.3.2, where detailed research is compiled to provide quantitative recommendations of space allowance during transport.

Corrective/mitigating measures of restriction of movement

In case of restriction of movement is suspected during the transit stage, the journey should be finalised as soon as possible. Upon arrival, animals should be unloaded with no delays and provided sufficient space to rest.

3.5.3. Quantitative examination of thresholds to protect the welfare of pigs during the transit stage: microclimatic conditions, space allowance and journey time

3.5.3.1. Threshold of microclimatic conditions

A) Introduction – pig biology and consequences of selection for productivity

Pigs are special among mammals as most of their sweat glands are non-functional. In addition, pigs have relatively small lungs, thereby reducing their ability to dissipate heat by panting (D'Allaire et al., 1996) and characteristically carry a thick subcutaneous adipose tissue layer, which impedes radiant heat loss. Taken together, this means that pigs have difficulty dissipating heat and may experience heat stress, especially at combinations of high ambient temperature and high humidity (Ross et al., 2015).

Heat stress is not only a problem for pigs during transport, but also a welfare concern under production conditions, leading to significant challenges to productivity, especially in tropical regions of the world but also during periods of hot weather in Europe (Baumgard and Rhoads, 2013).

Larger pigs, especially sows, are known to be more vulnerable to heat stress than smaller individuals. These differences are caused by their large body size, low surface area: body weight ratio, higher energy intake relative to maintenance requirement (in lactating sows), and higher subcutaneous fat tissues (Renaudeau et al., 2012). For example, body heat production increases with body weight (+2% for each 5 kg increment), while heat dissipation rate decreases (Brown-Brandl et al., 2004; Renaudeau et al., 2011). In addition, animals undergoing high metabolic activity, and especially in combination with high feeding level – as for example highly pregnant or lactating sows – are typically more susceptible to heat stress than animals of lower metabolic activity (Gourdine et al., 2021). This means that the reproductive status of sows affects their response to heat stress, and that sows that produce milk likely are more vulnerable than dry sows. In terms of transport, sows can be culled at all stages of the reproductive cycle, but considerable proportions of sows have been reported to be sent to slaughter just after weaning (when still producing milk) (Fogsgaard et al., 2018).

As reviewed by Gourdine et al. (2021) and Mayorga et al. (2019), there is growing evidence that genetic selection has reduced pigs' ability to cope with heat stress, due to an increase of metabolic heat production with the improvement in reproductive traits and lean tissue growth rate, at the expense of adaptive capacities. In a review of the potential for heat stress in future livestock production, Mutua et al. (2020) stated that the highly prolific modern genetic pig lines produce almost 20% more heat than the animals produced only a few decades ago.

B) Background

Thermoregulation is the physiological process allowing the balance between heat production and heat loss mechanisms. The approach taken in this Scientific Opinion to recommend microclimatic conditions during pig transport is based on the thermoregulatory concepts and model as described by EFSA (2004), and originally formulated by Mount (1974). The figure covers a range of environmental temperatures from cold to very hot. Since the selection of highly relevant WCs was not done per animal category (cold stress is likely more relevant for pigs close to weaning than for older categories), but per animal species, cold stress was not chosen as a highly relevant WC for the average pig. Thus, this assessment of thresholds for microclimatic conditions focusses on temperatures higher than B as indicated on Figure 6.





LCT/A: Lower critical temperature (LCT), UCT/D: Upper critical temperature; B: Lower limit of thermal comfort zone; C: Upper limit of thermal comfort zone.

Figure 6: Schematic representation of thermal zones as a function of the environmental temperature (Adapted from EFSA, 2004)

The following three concepts from the figure need to be introduced:

<u>Thermoneutral zone</u>: As reviewed by Bracke et al. (2020), the TNZ covers the range of environmental temperatures within which metabolic rate and heat production are constant and independent of the ambient temperature. The zone is limited by the lower critical temperature (LCT) (marked as A in Figure 6) and the UCT (marked as D in Figure 6). Many factors influence the TNZ of an individual animal including the size, body condition score, breed, level of nutrition, agitation level, and environmental factors such as humidity, radiation, heat loss to the floor, air velocity around the animal, but also motor activity (e.g. maintaining balance during transport) (Bracke et al., 2020).

<u>Thermal comfort zone</u>: According to Silanikove (2000) subdivision of the TNZ into a zone of thermal well-being is the most suitable way to describe the relation between an animal and its environment from the viewpoint of animal welfare. Based on studies in humans (e.g. Schlader et al., 2011), Kingma et al. (2014) described the TCZ as defined in terms of perception, qualifying as the state of mind that expresses satisfaction with the thermal environment. Translated into animal welfare, Silanikove described the TCZ (denoted comfort zone in Figure 6) as the environmental temperature interval, where the energetic and physiological efforts of thermoregulation are minimal, and the animal is in the preferred or chosen thermal environment. In the figure above, the upper limit of the TCZ is marked by C, where an animal will activate evaporative physiological thermoregulation processes, and may start to display thermoregulatory behaviour. The TCZ is sometimes called the safe zone as it is referred in EFSA Scientific Opinion to be published in 2022 on the welfare of animals in containers during transport (EFSA AHAW Panel, 2022c)

<u>Upper critical Temperature (UCT)</u>: As outlined in EFSA (2004), there are several definitions of UCT. The UCT describes the point above which an animal must significantly increase the use of physiological mechanisms to prevent a rise in body temperature above normal. For example, evaporative heat loss increases and metabolic rate increases (Silanikove, 2000). As described by Norris and Kunz (2012), heat is transferred by four mechanisms: radiation (from a hot object to a cooler object via electromagnetic waves), conduction (between two solid objects in contact with one another), convection (through the movement of a gas or liquid) and evaporation (conversion of water from the liquid to gas phase). In the TNZ, evaporation is per definition kept to a minimum, whereas increased evaporative heat loss through the skin and/or respiratory tract occurs when the organism is challenged with higher ambient temperatures. At high ambient temperatures, heat transfer by conductive, convective and radiant changes are less effective, because of the reduction of the required minimal thermal gradient between skin and air temperature (Renaudeau et al., 2012).

According to the definition of WCs published by EFSA AHAW Panel, (2022a), the term 'heat stress' is defined as: 'A situation, where an animal experiences stress and/or negative affective state(s) such as discomfort and/or distress when exposed to a high effective temperature'. This definition differs to some extent from other proposed definitions of heat stress, focusing on lack of ability to cope or on performance loss.

The scientific literature underpinning the model shown in Figure 6 is based on studies involving a certain level of feed intake under stable or resting conditions. As reviewed by Bracke et al. (2020), care should, thus, be taken when extrapolating findings obtained from experiments in conventional barns to transport conditions. During transport, pigs are often exposed to factors that may act as stressors and/or limit their possibility to thermoregulate, as they would have done in non-transported control conditions. In contrast to the conditions provided to the pigs under basic thermoregulatory studies, transport often includes deprivation of feed and water, exposure to vibration and motion forces, low space allowances and highly variable ventilation rates. Consequently, if a negative impact on animal welfare from the microclimatic conditions during journeys is to be fully prevented, animals should be transported in their TCZ. This means that the WC, heat stress, defined by the accompanying stress and/or negative affective states, may start when an animal is no longer in the TCZ, and the risk and severity of heat stress, is likely high when animals reach UCT. Once this point is reached, the rate of evaporative heat loss starts to increase exponentially, meaning that signs of heat stress, in pigs mainly seen as panting and attempts to lie laterally, increase steeply in an effort to stop the rise in core body temperature above normal.

The warm zone in Figure 6, also sometimes called the alert zone (EFSA AHAW Panel, 2022c), between temperatures C and D is not as such included in the TCZ. However, even though heat stress cannot be fully excluded when animas are exposed to conditions as between C–D, the risk and the severity of heat stress is likely not high in this interval. This approach is based on the definition of the animal WC heat stress, addressing a situation where an animal experiences stress and/or negative affective states such as discomfort and/or distress.

C) Heat stress and animal welfare during transport of pigs

As reviewed by Rashamol et al. (2019), Bjerg et al. (2020), and de Castro Júnior and Silva (2021), not only the ambient temperature, but also other environmental conditions influence heat load placed on animals. Examples of these are: relative humidity (RH), thermal radiation, solar radiation including long- and short-wave radiation, temperature of surrounding surfaces, the heat and moisture generated by the animals, the heat loss from the vehicle, vertical space, placement of compartment partitions along the longitudinal axis of the vehicle, the vehicle type, the type of ventilation shutters, wind speed, the opportunity for pigs to moisten their skin and many more. These will all influence the microclimatic conditions experienced by pigs and should, in theory, all be taken into account when microclimatic conditions of pigs during transport are evaluated. However, due to the complexity of such tasks, as well as the strong evidence for the effect of humidity on heat stress, at least the combined effects of temperature and humidity should be taken into account when animal welfare during transport is evaluated.

The water vapour content of air is important because it impacts the rate of evaporative heat loss through the skin and respiratory tract (Bohmanova et al., 2007). When the ambient temperature is above the animal's TCZ, a high level of humidity in the air will reduce evaporative heat loss and therefore result in increased risk of heat stress. In this case, the routes of conduction, convection and radiation for heat exchange are reduced, and the only remaining route of heat loss is through evaporative routes, which require a vapour pressure gradient and thus dictates that relative humidity is a major factor controlling rate of evaporative loss.

Generally, air water vapour content is assessed by RH, which is a measure of the percentage saturation of the air with water vapour at a specific temperature in relation to the maximum water vapour that the air could potentially contain at that temperature. However, RH is temperature dependent and thus the same RH at different temperatures may equate to very different water vapour contents. Therefore, although sensors recording temperature have been used in road transport of pigs in the past, it would be a significant refinement to use improved sensors that take into account humidity effects.

In the context of animal transport, ventilation functions to replace the metabolic heat and moisture produced by the animals inside the vehicle with air of a certain humidity and temperature from outside the vehicle. Ventilation also serves to mix and redistribute internal air to attempt to make the internal thermal micro-environment more homogeneous. In addition, concentrations of



different gases (O_2 , CO_2 , NH_3) can be modulated. The effect on individual animals depends on the rate of air change and the flow around the bodies of the animals. In this way, the temperature and humidity (and all other microclimatic conditions) in the vehicle can, in theory, be maintained only slightly elevated compared to the level of those outside the vehicle, but only if ventilation is very efficient.

However, in a passively ventilated vehicle, air flow over the surface of the moving vehicle results in a pressure gradient in which there is lower pressure towards the front sides of the vehicle than at the rear sides and tail. There may be higher pressure on the front (forward face or headboard). The net effect is that in a passively ventilated configuration, air movement will tend to involve entry of air towards the rear, forward movement of air towards the front of the vehicle and exit of air at the front sides of the structure (Figure 7).



Figure 7: Predominant patterns of air flow (black arrows) in a moving passively ventilated vehicle. Red areas indicate where air heated by the animals accumulates and blue areas show colder spots. Photo: Marianne Kaiser, Aarhus University

For all passively ventilated vehicles, when the vehicle is stationary (e.g. during mandatory driver breaks), there is no driving force for ventilation other than buoyancy or free convection or external factors, such as cross winds. The problems when stationary will be exacerbated in vehicles operating with restricted air inlets with minimal gaps for air to enter, exit and circulate within the load.

Even vehicles fitted with fans to aid ventilation often have the airflow dictated by two principles. The stack effect causes heated air to rise and colder air to descend and is the dominant means in stationary vehicles. When moving, the stack effect continues to operate, particularly in areas of the load with low ventilation, but is overlaid by the air flows which operate around and within a vehicle in motion. Both of these drivers of air flow may be influenced by external factors, such as wind. In the type of transport vehicle used in the EU, the pressure field around the vehicle drives the passive ventilation, and air will tend to enter towards the rear of the vehicle and exit towards the front side of the load. The presence of the sides on the vehicle limits air entry and exit along the sides of the vehicle and the route for air flow will be determined by the location of any openings in the structure. This results in uneven distribution of thermal conditions within the load with hot-spots (red) and cold-spots (blue) indicated in Figure 7 above.

As discussed by Mitchell and Kettlewell (2008), there is a ventilation rate (by mechanical or passive ventilation), for a certain range of environmental temperatures, that, in theory, reduces the effective temperature within the vehicle to the same level as that outside the vehicle. This rate will depend on many factors, such as the temperature and humidity of the air coming into the vehicle, the heat and moisture and heat generated by the animals, solar radiation and heat loss from the vehicle. To the best of our knowledge, this ventilation rate is unknown for pig transport, as the relevant studies have not been found.

Calculations of total heat production of finisher pigs and sows are around 2.0–2.5 W/kg for finisher pigs (Brown-Brandl et al., 2004) and 1.89 W/kg for breeding sows (Stinn and Xin, 2014), although the total heat production will change depending on temperature, stage of fasting and activity of the animals. Mitchell and Kettlewell (2008) measured the heat production of 100-kg finishing pigs during transport at a stocking density of 0.42 m²/pig. The heat output varied by stage of the journey, being highest after loading at 3.4 W/kg, falling to 1.5 W/kg following 3 h of transport, and rising back to 3.3 W/kg during a 2-h truck stationary period, reducing to 2.0 W/kg as the truck continued on the journey.



The production of water vapour followed a similar pattern (Mitchell and Kettlewell, 2008), with the production level during transport, while the vehicle was moving, being 0.0003 g/s per kg. Given the change in activity and the output of metabolic heat and water vapour, during periods of activity, such as loading and when vehicles are stationary, the values for total metabolic heat output and water vapour could be expected to double (Mitchell and Kettlewell, 2008). Knowledge on the metabolic heat and water vapour output at different stages of transport for a specific weight classification of pigs can be used to estimate the minimum ventilation requirements for temperature ranges, using the equation: [VFR = TMHP/Cp $\times \Delta T$], where: VFR = flow rate (m³/s); TMHP = total metabolic heat production (J/s; with J/s equivalent to W); Cp = specific heat capacity of air (1,226 J/m³ per °C); ΔT = acceptable rise in air temperature (°C) (Mitchell and Kettlewell, 2008). This calculation can be used as an estimation of the ventilation requirements from which to develop and test vehicle ventilation ability under varying conditions. However, this equation does not account for heat loss through the vehicle and does not address partitioning of heat loss between sensible and latent heat (Mitchell and Kettlewell, 2008). The stocking density of a vehicle should consider the ventilation capacity of the vehicle, to ensure that sufficient ventilation can be maintained when it is in use for a specific type and age of pigs (Randall, 1993).

When trucks are stationary, for example for loading and unloading, the risk of heat stress increases. The internal temperature has been measured to increase up to 1°C/min (Weschenfelder et al., 2012; Xiong et al., 2015) in stationary, naturally ventilated vehicles carrying finisher pigs, the extent of the increase being influenced by the environmental temperatures, and type of pig, type of vehicle and space allowance. This evidence suggests that naturally ventilated vehicles therefore must have sufficiently large (or alternatively properly designed) ventilation spaces, to reduce heat build-up when stationary, or that the addition of mechanical ventilation should also be provided and be turned on (see also EFSA, 2004).

However, there exists very little information on how effectively different transport ventilation designs are able to keep pigs within the required temperature ranges when using mechanical vs passive ventilation. Further research is needed in this area. Vehicle designs and ventilation capacities must enable the animals to be transported safely within the required temperature ranges, depending on the local climatic conditions through which the vehicle will travel. Considering the transport of heavier pigs, with higher metabolic outputs, and the potential for increasing microclimatic extremes with climate changes, this is an area that should receive further research.

D) Thermal heat indices

Several indices have been developed to predict stressful environmental conditions that take into consideration multiple weather-related factors and allow execution of abatement strategies. The majority of these have been based on ambient temperature and relative humidity. One, the temperature-humidity index (THI) (as originally described by Thom, 1959), has been taken up by the livestock industry as a weather safety index to monitor and reduce heat-stress-related production losses. The fact that this use of THI thresholds is mainly focused on avoiding production losses, such as in-transit mortality, means that it is not necessarily aligned with animal welfare, as defined by affective states. Recent studies have called for further development of indices used to assess heat stress in livestock due to limitations in for example THI, such as (1) lack of integration of all environmental parameters; and (2) either not reflecting current high producing animals or not specifying production level. Factors like these may limit the usefulness of the indices to accurately predict or assess the thermal status of pigs and other livestock (Herbut et al., 2018; Mayorga et al., 2020). THI has, however, been used to quantify heat stress in studies of pig housing and transport (e.g. Lucas et al., 2000; Weschenfelder et al., 2012, 2013; Fox et al., 2014; Xiong et al., 2015; Pereira et al., 2018; Moak et al., 2022a,b), and studies quantifying THI have also been included in the data underlying the conclusions of this Opinion.

To avoid the limitations presented by the different available THI's or other comparable indexes, psychrometric principles (related to the humidity and temperature of air) have been used to develop other thermal comfort indexes, such as the specific enthalpy of air (de Castro Júnior and Silva, 2021). Enthalpy is the heat energy of the air surrounding an animal, and dictates the degree of heat loss to the microclimate. Physically, the specific enthalpy of air (h) is defined as the total amount of energy existent in a unit of dry air mass (kJ/Kg of dry air) and can be calculated using simple tools, such as thermometer and hygrometer, and mathematical models, as recently reviewed by De Castro and Da Silva (2021). In the future, time derivatives of temperature or enthalpy could be used as non-invasive welfare indicators during animal transport and appear to be more sensitive than values of temperature

or RH. Even though enthalpy has been reported in some studies in relation to the quantification of heat stress in pigs during transport (e.g. Dalla Costa et al., 2019; Machado et al., 2021; Miranda-de la Lama et al., 2021), at the present stage, it is considered that too few data are available to be able to give recommendations based on enthalpy. Thus, this Scientific Opinion focuses on the temperature immediately surrounding the pigs and measures of humidity. The combined ambient temperature and RH can be reported by different indices, such as wet-bulb temperature (Twb), apparent equivalent temperature (AET), previously used by Mitchell (2006) to assess heat stress during transport of broilers, or Enthalpy Comfort Index (ECI) employed in tropical regions as a qualitative indicator of thermal environment of livestock (Rodrigues et al., 2011). However, none of these indices have been validated for pigs during transport.

Irrespective of the index chosen to monitor temperature and humidity, vehicles should be equipped with sensors recording temperature and humidity as close as possible to the position of the animals therein, and at several locations to include hot as well as colder spots (Figure 7). The livestock driver should then monitor the microclimate of the load and adjust the ventilation if the conditions exceed the TCZ levels. Technical issues (e.g. accuracy, maintenance, placement, reliability and calibration) relating to this improved approach will need to be addressed (Samad et al., 2020).

E) Identification of microclimatic conditions to protect pigs from heat stress during transport

In general, as discussed by Bracke et al. (2020), studies of heat stress in pigs have focused on housing conditions, and to a much lesser extent on transport conditions. This means, that knowledge considered in this Scientific Opinion is a combination of information from studies done on-farm and during transport. The vast majority of studies of heat stress in pigs have focused on finishers. In recent years, also sows have received scientific attention in terms of heat stress, but almost only in relation to on-farm productivity and welfare (e.g. Bjerg et al., 2020), whereas only a few studies have focused on heat stress during sow transport (e.g. Thodberg et al., 2019, 2022). Similarly, for weaners of ~ 30 kg, the category of pigs typically transported long distances in Europe, only very limited information exists about heat stress during transport (Averós et al., 2010).

The predominant physiological responses measured as indicators of heat stress in pigs under experimental conditions – often involving respiration chambers (Noblet et al., 1989; Tauson et al., 1998) or temperature chambers (Brandt et al., 2022) – are respiration rate (RR) and rectal temperature (RT) (as reviewed by Mayorga et al., 2019). Some focus has been given to indicators of productivity, such as voluntary feed intake or average daily gain (Huynh et al., 2005), and more recently, also behavioural indicators have been included. Examples of these are thermal preference studies (Robbins et al., 2021). Without any detailed quantification, Tauson et al. (1998) described how the boars in their study of heat stress showed behavioural responses, such as restlessness and biting the fixtures of the respiratory chambers. In a study involving gilts exposed to temperatures of 15–30°C while kept in individual crates, Canaday et al. (2013) observed increased occurrence of postural changes, increased lying, and increased lateral lying at the highest temperature.

In this Opinion, ABMs relevant for the study and the documentation of heat stress in pigs during transport have been listed in Section 3.2.2. An early response of pigs to increasing thermal load is to increase the RR, and thus the respiratory evaporative heat loss. 'First, a rapid shallow breathing called thermal polypnea leads to an increase in the amount of air passage through the upper region of the respiratory tract. When temperature continues to rise, the thermal polypnea shifts to a slower, deeper, panting phase (thermal hyperpnea) characterized by an increase of alveolar ventilation rate' (Renaudeau et al., 2012). As defined by Collier and Gebremedhin (2015), 'panting is, a controlled increase in respiratory tract while preserving alveolar ventilation. The increase in upper respiratory tract ventilation increases evaporative heat loss'. Operationally, panting is often recognised by open mouth breathing, and this behaviour has been reported in pigs before, during and after transport.

The identification of the upper threshold of the TCZ is less certain than identification of the UCT, as it is challenging to identify a point where a response starts, as compared to the identification of a significant increase (or decrease) in a response. The current work has been based on the available scientific literature, and especially knowledge presented in two reviews of scientific literature focusing on heat stress in pigs (Bjerg et al., 2020; de Castro Júnior and Silva, 2021). The conclusions are given for the three main pig categories relevant for transport: weaners < 50 kg, finishers, and sows and boars. Table 15 summarises information from the literature.



Table 15:Summary of literature findings and suggestions for upper threshold of TCZ and for UCT.
For each of the references, the reported dry temperature is listed, as well as the primary
ABM used in the studies

Pig category	Upper threshold of TNZ	Upper critical temperature	ABM	
5 5 7	Temperature (°C)	Temperature (°C)		
Weaners				
Collin et al. (2002)	25 15–22 kg		RT	
Renaudeau (2005)		LW ^(b) : 28–30 Creole ^(b) : 30–32 28–55 kg	RT	
Hörtenhuber et al. (2020) ^(a)		8–30 kg: 30 30–60 kg: 25		
Market weight				
Gourdine et al. (2021) ^(a)	22	25		
Huynh et al. (2005)	22	Gilts, 60 kg: 26	RR, RT	
Hörtenhuber et al. (2020) ^(a)		60 kg: 24		
Sows and boars				
Robbins et al. (2021) ^(c)	Sows: 16		Preference for staying	
Gourdine et al. (2021) ^(a)	Sows: < 18	Sows: 22		
Ribeiro et al. (2018) ^(a)	15–25 Lactating sows			
Bortolozzo et al. (2011)	16–22 Sows			
Quiniou and Noblet (1999)		18 Lactating sows	RT	
Tauson et al. (1998)		Boars: 30–32 100 kg	RT	
Bjerg et al. (2020) ^(a)	< 20 Sows	22 Sows	RR	
Hörtenhuber et al. (2020) ^(a)		After weaning: 24 Lactating 22 Boars: 19		

TNZ: thermoneutral zone; UCT: upper critical temperature.

(a): Reference is a review.

(b): Study involving sows that were either non-pregnant or early-mid pregnant.

(c): When available, the pig breed is specified.

For all three pig categories, the scientific literature shows some variation in the reporting of the upper threshold of the TNZ and of UCT. As discussed by Bjerg et al. (2020), these differences can be explained by the difference in the use of ABMs, differences between breeds, housing or feeding, and also often that studies are comparing two or more environmental conditions, and thus can in principle not predict clear thresholds, but only report differences between, for example, ambient temperatures of 20°C and 30°C. In addition, the age span of the reported studies is wide, which means that results will also be affected by the consequences of the selection for increased productivity. In recent years, the vulnerability of sows towards heat stress has received increasing scientific focus. In their review of studies focusing on this topic, Bjerg et al. (2020) compared results from available studies. Figure 8 shows their findings of RR as a function of air temperature reported in different studies. Based on these data, the authors concluded that the upper threshold of the TNZ for lactating sows may be lower than 20° C, and that UCT is ~ 22° C.





Figure 8: Respiratory rate of sows as a function of air temperature, including data from several studies (Bjerg et al., 2020)

The thresholds suggested by Bjerg et al. (2020) are overlapping with findings from other studies. In their study, based on the establishment of a mechanistic thermophysiological model for sows (Huang et al., 2021a,b), Huang et al. (2022) concluded that when RH > 60%, the effect of a 10% increase in RH on the thermophysiological responses of gestating sows is equivalent to the effects of 1°C increase in ambient air temperature.

For finishing pigs, the available studies suggest that the upper threshold of the TNZ is 18–22°C, and that UCT is in the interval from 24°C to 27°C. Weaners are less sensitive to heat stress than the two other categories because of their lower bodyweight. For weaners, the available studies suggest that the upper threshold of the TNZ is ~ 25°C, whereas UCT corresponds to ~ 30°C. These values might, however, depend on whether the weaners are closer to 10 kg body weight, or above 30 kg, and far from the actual weaning transition period, when feed intake may be low.

Tolerance and adaptation

At present, pig breeding programmes are not including traits associated with thermoregulation, although these are now the subject of investigation by some breeders. Unless such traits are integrated in selection indices, pigs of the future will likely not be more resistant to heat stress than the pigs of today (Gourdine et al., 2021), but may be even more vulnerable (Renaudeau and Dourmad, 2021).

Studies do, however, suggest that there are breed differences in the thermal tolerance of pigs (Tauson et al., 1998; Sejian et al., 2018), and also that acclimatisation to a hotter environment to some extent might be possible, but will take days to weeks (Renaudeau et al., 2010; Collier et al., 2019). It has not been possible to find data to clarify whether pigs of certain breeds, or from certain regions of the EU, are better adapted to hot conditions than others. This constitutes a gap in knowledge.

F) Summary of microclimatic conditions

Pigs are special among mammals and have difficulty in dissipating heat as their sweat glands are almost non-functional (Ingram, 1965). Thus, pigs are more vulnerable to heat stress than most other farm animals. Across the different pig categories, there is growing evidence that genetic selection for productivity and more efficient accretion of lean tissue, has increased their metabolic rate and reduced pigs' ability to cope with heat stress (Mayorga et al., 2019; Mutua et al., 2020; Gourdine et al., 2021).

Larger pigs, especially lactating sows, are more vulnerable to heat stress than smaller pigs. The available information for boars is very limited. Compared to the other categories of pigs, typically subjected to transport, weaners are likely to be the least vulnerable to heat stress.



It has not been possible to find data to clarify whether pigs of certain breeds, or from certain regions of the EU are better adapted to sustain heat stress than others. This constitutes a gap in knowledge.

If a negative impact on animal welfare from the microclimatic conditions during journeys is to be fully prevented, pigs should be transported in their TCZ. This means that, during transport of pigs, the WC, heat stress, may start when pigs are no longer in their TCZ, and the risk and severity of heat stress is likely high when the thermal conditions reach the UCT.

Not only the temperature, but also other environmental conditions influence heat load placed on pigs during transport, such as humidity, thermal radiation, temperature of surrounding surfaces, wind speed, and the opportunity for pigs to moisten their skin. These will all influence the microclimatic conditions experienced by pigs and should, in theory, all be taken into account when microclimatic conditions of pigs during transport are evaluated.

The available information on thermoregulation of the three relevant categories of pigs, is not fully unambiguous, but has allowed for estimates of upper thresholds of the TCZ, and for the UCT to be derived.

For sows, the upper threshold of the TCZ may be lower than 20°C, and that the UCT is ~ 22°C. For finishing pigs, the available studies suggest that the upper threshold of the TNZ is 18–22°C, and that the UCT is in the interval from 24°C to 27°C. Weaners are less sensitive to heat stress than the two other categories because of their lower bodyweight. For weaners, the available studies suggest that the upper threshold of the TCZ is ~ 25°C, whereas the UCT corresponds to ~ 30°C. For variations of dry temperature and relative humidity, the higher the levels of relative humidity, the lower the upper thresholds of TCZ and UCT will be, when measured as a dry temperature only.

Although sensors recording dry temperature have commonly been used in transport of pigs so far, it would be a significant refinement to use improved sensors taking account of humidity effects in addition to temperature.

3.5.3.2. Threshold of space requirements during journeys

A) Introduction and methodology

The stocking or loading density refers to the live weight of pigs within a specified area of floor space (or occasionally the number of pigs of a specified live weight range per unit area). The space allowance can be quantified as the floor area per animal. In this Scientific Opinion, space allowances are given as m² per animal, as well as the estimated k-value for the allometric equation for space allowance (Petherick and Philips, 2009).

The spatial dimensions of compartments holding pigs during transport are important for their welfare – in the horizontal as well as the vertical plane - and a lack of space may lead to several WCs, such as group stress, injuries, restriction of movement, resting problems and heat stress.

Due to the limited research available, the multiple factors that can influence how pigs respond to space during transport, the variability in types of pigs and in journey conditions, it is considered preferable to provide minimum rather than target or recommended space allowances for different types of pigs. The evidence of WCs when inadequate space is available is stronger than that available for determining optimal conditions.

Multiple factors affect the use of space by pigs during transport, including the age and category of pig (weaner, finisher, sows/boars), microclimatic conditions and journey duration. Body weight and dimensions change with the pig category and age (Condotta et al., 2018), and are related to the space required (Arndt et al., 2019, 2020). It is, thus, a complex issue to provide a minimum recommended space allowance during transport that will be applicable within all situations. If the minimum recommended space allowance is set too low for a particular situation, it will likely increase the risk of adverse WCs.

In the assessment of minimum space requirements for pigs during transport, the following approach will be used:

During transport, pigs require a minimum space allowance that will accomodate (a) their physical size in a standing (or sternally lying) posture, and thereby allow them to (b) adjust their posture in response to acceleration and other events, (c) rest in a normal semi-recumbent lying posture, (d) thermoregulate, and (e) eat and drink, if feed and water are provided in the means of transport. Recommendations for a minimum space allowance will be set by the first limiting factor that reduces the ability of the pigs to undertake one of the above biologic requirements, i.e. whichever of the above requirements needs the most space.

The section also discusses compartment height, as another dimension of space, and WCs, hazards and AMBs listed above for the transit stage (Sections 3.2.2 and 3.5.2) are used to draw conclusions.

B) Horizontal space

i) Space to accommodate physical size

The space occupied by pigs in a standing or sternally lying posture depends on the size of the animals, and allometric equations are available that relate live weight to floor space. Allometric equations ($[A = kW^{2/3}]$, where k is a constant and W represents live weight in kilograms) are used to estimate the space that a stationary animal occupies as a consequence of its mass (Petherick and Philips, 2009). The power factor is derived from theoretical relationships between length, volume, weight and surface area (Warriss et al., 1998). Using an exponent of 2/3 makes the assumption that all pigs have a similar shape. Therefore, allometric equations in the k-values used in allometric equations produce a range of recommended space allowances for each live weight range (Warriss et al., 1998; Visser, 2014).

Using allometric equations it has been proposed that the space requirement for animals in a group to stand or lie sternally (legs tucked close to and under the body) can be estimated from k = 0.019 (Petherick, 1983, cited by Ekkel et al., 2003). However, differences in how pig categories socially interact will affect the use of space, as shall be discussed below.

Weaners: It is common practice in areas of Europe for newly weaned piglets to be transported short distances to nursery barns (multisite, integrated production) and for weaners of around 30 kg to be transported to rearing barns for further fattening (Dahl-Pedersen and Herskin, 2021). The transport of newly weaned piglets can pose unique challenges due to their size, having lower body reserves and being less tolerant of cold stress.

There is limited published work on the effect of space allowance on the transport of weaned piglets. Therefore, the limited studies that are available for weaned piglets have been reviewed, and other studies on very young piglets, weaned earlier than the current practice in the EU were also included. The physical space requirements to stand and lie sternally based on a k-value of 0.019 ranges from 0.09 to 0.18 m²/pig for piglets ranging from 10 to 40 kg, respectively (Table 16). For 18-day old piglets (~ 5 kg), transported in summer in the Southern United States for journeys of 2–3 h, Sutherland et al. (2010) suggested that 0.06 m²/piglet (corresponding to a k-value of 0.021) provided space for sternal lying. In the same study, transporting pigs at a lower space (0.05 m²/pig) increased the neutrophil:lymphocyte ratio, suggestive of physiologic stress/and or inflammation, while reduced lying and increased standing was found when they were transported at 0.07 m²/pig, which was interpreted as possibly indicating increased instability (Sutherland et al., 2010).

Newly weaned piglets show a natural propensity to overlie even when a larger space allowance (k-value of 0.039) is provided (Kaur, 2018). No benefits have been found when providing greater space allowances for piglets weaned very early and transported shortly after weaning (6–10 kg) on short trips of up to 1 h at (0.064, 0.084 or 0.145 m²/pig), with piglets huddling, and all available floor space not utilised (Riches and Guise, 1997). Thus, from the point of view of providing the pigs with space enough to lie sternally, their tendency to overlie at this age may mean that they in practice will be able to lie down, even with less space than corresponding to a k-value of 0.019, at least as long as they are kept in their TNZ.

For weaners in nursery housing, the average number of pigs performing overlying behaviour has been observed to reduce in pigs nearing 25 kg in weight (42 and 63 days of age) compared to pigs in the first week after weaning (25–28 days of age), when kept under thermoneutral conditions (Kaur, 2018). Therefore, for weaner pigs (10–40 kg) a space allowance as calculated for a k-value of 0.019 will likely be required for all animals to lie down sternally (Table 16). It should be acknowledged though, that there is insufficient evidence for how weaners will perform during transport at this space allowance.

Table 16: Minimum space requirements for weaners (10-40 kg) to stand or lie sternally (k-value: 0.019)

Pig weight (kg)	m²/pig	kg/m ²
10	0.09	112.5
20	0.14	141.4



Pig weight (kg)	m²/pig	kg/m ²
30	0.19	161.7
40	0.22	177.8

Finishers: Within the EU, pigs are transported by road to abattoirs for slaughter. While the average EU slaughter weight for finisher pigs is 124 kg (ADHB, 2021), the 'heavy pig,' reared in some Mediterranean MS, for the production of certain meat products, is slaughtered at ~ 150–170 kg (Zappaterra et al., 2022).

Image analysis quantifying the area occupied by pigs of different weights and different postures provide data to compare to that estimated by allometric equations. Contrast-based planimetric analysis of images determined that the space occupied by finishers (mean 109 kg) is influenced by the weight and posture of the animals. Standing space ranged from a low of 0.29 m² for a pig standing straight with the head down, to a high of 0.33 m² for a pig standing with the head up and a curve in the body (Arndt et al., 2019), providing a space lower than that calculated by a k-value of 0.019. A pig lying on its sternum occupied between 0.43 and 0.44 m² (Arndt et al., 2019), a value that is aligning to the space required for a 110-kg pig as calculated from a k-value of 0.019 (Table 17).

Heavy finisher pigs (150–170 kg): Given that static space requirements relate to the weight of the pig (Arndt et al., 2019) and that this relationship is not linear, it is important that specific space requirements for heavy finishers are given consideration. Heavy finishers likely have different space requirements due to their increased size, metabolic output, large muscle mass and low cardiac output (van Essen et al., 2018), altogether making these pigs more susceptible to stress and mortality during transport. Heavier pigs have an increased heat output (+2% for every additional 5 kg of liveweight; Brown-Brandl et al., 2004), and a reduced ability to dissipate heat (Renaudeau et al., 2011) making them more susceptible to heat stress. However, even though pre-slaughter data on death loss of Italian heavy pigs (~ 160 kg) has identified that 56% of losses occur during transport, space allowance during transport was not identified as a risk factor (Nannoni et al., 2017). However, given the limited research in this area, studies to investigate the effect of space allowance should be performed. Based on a lack of available evidence on how heavy finishers respond to space allowance in transport, the minimum space required for these to stand up and lie down sternally, without additional room to manoeuvre between positions, can be calculated based on allometric equations with a k-value of 0.019 (Table 17).

Pig weight (kg)	m²/pig	kg/m ²
100	0.42	240.6
110	0.44	248.3
120	0.47	255.5
130	0.50	262.3
140	0.52	268.8
150	0.55	275.0
160	0.57	280.9
170	0.59	286.6

Table 17: Minimum space requirements for finishing pigs to stand or lie sternally (k-value 0.019)

Sows, gilts and boars: Within the EU, gilts (130–180 kg) are transported to farms as replacement breeding animals, and sows at the end of their productive life are transported to slaughter. According to Dahl-Pedersen and Herskin et al. (2021), the number of pigs transported between EU MS for breeding has been quite variable during the last years, ranging from less than 1 million to more than 3 million per year due to a volatile market. For sows, Blair and Lowe (2019) stated that across pig producing countries, a culling rate of up to 50% per year is not unusual, which means that a considerable number of sows are sent to slaughter each year. In the EU, the majority of sows are probably slaughtered in the country of origin, but some are transported to other MS for slaughter.

While there is variation in genetics and between countries, there is evidence that at least in some genetic lines, the size of sows has steadily increased in the last three decades, with sows becoming heavier and longer (Moustsen et al., 2011). However, comparing sow dimensions for length, width and



height between that calculated by the allometric equations proposed by Petherick and Philips (2009) to the measurements taken by Moustsen et al. (2011), in Danish sows ranging from parity 1 to 7, indicates that the allometric equations continue to provide close estimations of the body dimensions of these modern genetics.

Planimetric analysis of the static space floor area occupied by sows in different postures (average sow weight 235 kg, min: 187; max: 310 kg) has identified that when standing, the occupied floor area ranges from 0.42 m² to 0.47 m² (min: 0.34; max: 0.59 m²), and sternal lying requires an average floor space of 0.63 m² (min: 0.52; max: 0.71 m²), (Arndt et al., 2020). This is lower than the minimum space allowance estimated for sows to stand and lie sternally as calculated with a k-value of 0.019. Despite these recent findings, in the face of a lack of evidence on how space allowance during transport influences sow behaviour and welfare, a minimum space requirement for sows to stand and lie sternally should be based upon the minimum space calculated from allometric equations with a k-value of 0.019 (Table 18).

Pig weight (kg)	m²/pig	kg/m ²
180	0.62	292.1
200	0.66	302.4
220	0.70	312.1
240	0.75	321.2
260	0.79	329.8
280	0.83	337.9
300	0.87	345.7
320	0.91	353.1

Table 18: Minimum space requirements for sows to stand and lie sternally (k-value 0.019)

ii) Space required to adjust posture in response to acceleration and other events

During road transport, the space allowance provided to pigs needs to accommodate not only their static space, but additionally space to change and adopt postures to balance and stabilise themselves during vehicle movement, for example during the navigation of corners, acceleration and braking.

Theoretically, both low and high space allowance could increase injury to animals during driving events. Injury in the form of bruises, superficial skin lesions and open wounds will increase if the space allowance is low, resulting in a greater risk for pigs to become trampled, trapped or crushed by others in the load. Lameness can also increase as a result of injuries sustained, and the inability of pigs to adequately stabilise themselves, and achieve a comfortable position throughout the transit stage. Earlier (AHAW, 2002), it has been discussed whether animals, in situations of poor driving or emergency responses, could benefit from a stocking density that provided mutual support, i.e. whether a high space allowance, could also reduce the ability for pigs to adequately stabilise themselves during the journeys. Evidence from other species (Jones et al. (2010) in sheep and Tarrant et al. (1992) in cattle) indicates, however, that if vehicles are driven well on good quality roads, the animals benefit from plenty of space, suggesting that slips and falls in relation to space allowance on the vehicle, are mainly associated with low space allowance, and that animals need increased space compared to stationary conditions to adopt postural changes to brace themselves while standing, and to adjust their footing in order to maintain stability in response to changes in acceleration (including braking, stopping and changes of direction). This has, however, not been investigated in pigs and constitutes a gap in knowledge.

Weaner pigs: There is no literature available observing how piglets adjust posture in response to driving.

Finisher pigs: There is little available evidence of how space allowance influences the ability of pigs to stabilise themselves during driving events of transport. The extent of injury pigs receive during transport is commonly reported, which may indicate increased risk of injury if space allowance is too great or too small (Barton Gade and Christensen, 1998; Urrea et al., 2021). However, it should also be noted that many trials do not describe whether the pre- and post-transport handling has been controlled for, which could contribute to incidence of injury and bruising.

On long durations of transport (\geq 25 h), pigs have been observed to be more active at lower space allowances, repeatedly changing posture because all animals cannot lie down at once (Lambooij



18314732, 2022, 9. Downoaded from https://efsa.onlinelibrary.wiley.com/doi/10.2903/j.sa.2022.7445 by Universidad De Zaragoza, Wiley Online Library on [3001/2025]. See the Terms and Conditions (https://onlinelibrary.wiley.com/terms-and-conditions) on Wiley Online Library for rules of use; OA articles are governed by the applicable Creative Commons License

et al., 1985; Lambooij and Engel, 1991). While the available data may suggest there is a balance to be struck between too low and too great a space allowance. Observations on lying behaviour (Section iii) suggest that when provided with extra space, pigs will use it (Lambooij and Engel, 1991) and will remain recumbent while the vehicle is moving (Gerritzen et al., 2013), particularly on longer journeys. The evidence of greater injury does also suggest that more careful driving is required if pigs are to be transported at larger space allowances, for it could increase the risk of injury. Thus, even though too much space might be counterproductive on shorter journeys, evidence suggests for longer journeys, if vehicles are driven well on good quality roads, pigs benefit from space.

Sows, gilts and boars: There is no research available to provide information on how space allowance during transport influences the ability of sows, gilts and boars to stabilise themselves during transport. This is a knowledge gap that needs to be addressed. The clinical condition of cull sows transported on journeys of up to 8 h has been observed to deteriorate during transport to slaughter at a space allowance of 0.80 m²/250 kg sow (Thodberg et al., 2019), further warranting the need for further study of the behaviour of sows while on the vehicle to determine what occurs on these journeys.

iii) Space required to rest in a normal semi-recumbent lying posture

Whether pigs require the space to lie down and rest on a journey is dependent on age of pigs, category of pig, journey duration, condition of the pigs and within vehicle microclimatic conditions. Loading and transport are stressful for pigs, and ensuring pigs have sufficient space to lie down and rest on the journey may help them to recover and prevent the WC resting problems, and the development of fatigue. In terms of space requirements, it has been proposed that allometric equations should use a k-value of 0.027 to calculate the minimum space required for a group of animals to lie down and rest simultaneously (Petherick and Philips, 2009). The space allowance provided at a k-value of 0.027 does not accommodate additional space required for manoeuvring between a lying and standing posture. However, given that pigs do show evidence of space sharing (not all animals lying down at once, or for young pigs, more overlying behaviour), the space provided at this k-value could be considered a first step to providing enough space for all animals to lie down over the course of a journey.

Weaners: Observations from limited studies, and personal observations of researchers suggest that when transported, newly weaned pigs (21–28 days of age) are inclined to lie down rapidly from the start of the journey, sometimes lying down shortly after loading (Y. Seddon, University of Saskatchewan, personal observation on newly weaned piglets).

In older weaner pigs, the speed at which they lie down may be more variable. Magnani et al. (2014) reported a reduction in activity of 13-14 kg piglets following 4 h of transport. The minimum space requirement for a group of weaner pigs to simultaneously lie down is given in Table 19.

Pig weight (kg)	m²/pig	kg/m ²
10	0.13	79.2
20	0.20	99.5
30	0.26	113.8
40	0.32	125.1

Table 19: Minimum space requirements for weaner pigs (10-40kg) to simultaneously lie semi-recumbent (k-value 0.027)

A space allowance of 0.15, 0.18 or 0.23 m²/pig (corresponding to a k-value of 0.016, 0.019 and 0.024, respectively) was not found to influence the heart rate, salivary cortisol or postures of 30-kg weaner pigs during commercial transport (mean duration: 3.5 h, internal vehicle temperature: 13–15°C). Observations of postures found that almost half of the pigs sat or lay during the journey, with pigs choosing to huddle together, leaving free floor space (Riches et al., 1997; Riches and Guise, 1997). Environmental conditions being below the TNZ of 30-kg piglets may contribute to the overlying; overlying has been observed to reduce in 25-kg pigs in nursery housing under thermoneutral conditions (Kaur, 2018).

In observations from 4 loads of 5-week-old piglets (estimated 13–14 kg) transported one week after weaning, for 14 h, at a space allowance of 0.15 m²/pig, equivalent to k-value = 0.027, on average 65% of the piglets lay down within the first 4 h, changing to 77% of piglets after 10 h



(Magnani et al., 2014). When the majority of the piglets were lying, the second most frequent posture was standing, with low levels of exploratory and social behaviours observed to occur throughout the journey (Magnani et al., 2014). Aggressive behaviour was observed during transport, but at a higher frequency in pens mixed prior to loading (on-farm), than pigs mixed at loading (Magnani et al., 2014).

Observations suggest that in very early weaned piglets transported immediately after weaning (17 days of age), normal social behaviours (fighting commencing between unfamiliar pigs) will take place, which has been observed in North American journeys of long duration (up to 24 h) and space allowance of 0.06 m²/pig (Lewis and Berry, 2006). However, during transport at night, weaners of 20–25 kg transported at a space allowance of 0.18 m²/20 kg (between k-value 0.024 and 0.03 depending on pig weight) pigs have been observed to all lie down (C. Kobek-Kjeldager, Aarhus University, personal observation) as seen in Figure 9, and therefore the use of space sharing may depend on factors, such as age, space allowance and the time of day.

This limited evidence suggests that a space allowance calculated from a k-value of 0.027 could be



Figure 9: Photo taken with fish-eye lens from video recordings of the behaviour of 20- to 25-kg weaners during transport. Taken during driving, after 03:47 h of driving. The pigs were transported for 8 h at commercial Danish density (0.155 m²/pig weighing 22 kg, corresponding to a k-value of 0.019). The deck height was 70 cm and the mean outdoor temperature of the day was 21°C. Photo by C. Kobek-Kjeldager, Aarhus University)

sufficient to provide lying space and (on the basis that typically not all animals will lie down at the same time, or that some huddling will take place) may also accommodate behavioural activities in young pigs transported. It could be expected this may include moving around the compartment and being able to change posture.

Finisher pigs: The minimum space allowance for finishers, both standard and heavy, to simultaneously lie semi-recumbent is provided in Table 20. Lambooij and Engel (1991) reported that transporting finisher pigs (110 kg) at 232 kg/m² (k-value: 0.021) provided just enough room for all animals to lie down sternally. However, pigs lay down sooner in the journey when provided with 0.59 m²/pig (186 kg/m²) - equivalent to a k-value of 0.025.

Table 20:	Minimum space allowance for finisher pigs to simultaneously lie semi-recumbent (k-value
	0.027)

Pig weight (kg)	m²/pig	kg/m ²
100	0.59	169.3
110	0.63	174.7
120	0.67	179.8
130	0.70	184.6
140	0.74	189.2
150	0.78	193.5



18314732, 2022, 9. Downloaded from https://efsa.onlinelibrary.wiley.com/doi/10.2903/j.efsa.2022.7445 by Universidad De Zaragoza, Wiley Online Library on [3001/2025]. See the Terms and Conditions (https://onlinelibrary.wiley.com/terms-and-conditions) on Wiley Online Library for nales of use; OA articles are governed by the applicable Creative Commons License

Pig weight (kg)	m²/pig	kg/m ²
160	0.81	197.7
170	0.84	201.7

Urrea et al. (2021) compared the behaviour and physiology of finishers (~ 118 kg) transported for 8 h at either 0.5 m²/100 kg (k-value: 0.021), 0.43 m²/100 kg (k-value: 0.018) or 0.37 m²/100 kg (k-value: 0.015). A greater proportion of pigs lay down during the journeys when transported at 0.5 m²/100 kg and 0.43 m²/100 kg than 0.37 m²/100 kg suggesting on longer journeys with available space, pigs will lie down. Additionally, a greater proportion of pigs adopted a sitting posture when transported at 0.37 m² than 0.5 m², with pigs transported at 0.43 m² being no different. Upon arrival in lairage, a greater proportion of pigs transported at 0.37 m² lay, while pigs transported at 0.5 m² and 0.43 m² stood more (Urrea et al., 2021). Together these data indicate that having sufficient space to lie down during transport can help to reduce fatigue on longer journeys, with failure for sufficient space resulting in more compensatory postures and a greater fatigue in pigs upon arrival.

There are no published studies of finisher pigs transported at space allowances within the range proposed in Table 20. There is one study exploring the effect of transporting pigs 550 km at a space allowance of 0.59 m²/pig for 110-kg pigs, equivalent to a k-value of 0.0255 (Gerritzen et al., 2013). Based on findings of increased resting behaviour and lower body temperature and heart rate, the authors concluded that pigs are more capable of adapting to long transport conditions when loaded at a density below what was required in the EU at the time of the study (0.47 m²/pig, equivalent to k-value = 0.0205).

Studies of the behaviour of finishers during transport have shown that during shorter (2 h) journeys, pigs spend a numerically greater proportion of observations standing (mean: 46%) than lying (8%), (Torrey et al., 2013a). On longer journeys (8 h), pigs have been observed lying for a numerically greater proportion of observations (lying: 52% vs. standing: 16%) (Torrey et al., 2013b).

Transporting finishers (110 kg) at 0.59 m²/pig (k-value of 0.026) resulted in more fighting activity between mixed pigs, but also more pigs lying down and remaining recumbent throughout the journey compared to pigs transported at 0.42 m²/pig (k-value of 0.018): three hours into the journey 60% of pigs in the low-density treatment were recumbent compared to only 20% in the commercial density treatment (Gerritzen et al., 2013), providing more evidence that given the space, pigs will use it to rest. Given that transport, especially long journeys, are fatiguing for the pigs, it is considered the pigs need to have the space to lie down and rest.

Sows, gilts and boars: There is a lack of published data from which to provide evidence for the requirements of sows, gilts and boars to simultaneously lie semi-recumbent. In this absence, the minimum space requirements as calculated by allometric equation are provided (Table 21).

Pig weight (kg)	m²/pig	kg/m²
180	0.88	205.5
200	0.94	212.8
220	1.00	219.6
240	1.06	226.0
260	1.12	232.0
280	1.18	237.8
300	1.23	243.3
320	1.29	248.5

 Table 21:
 Minimum space requirement for gilts, sows and boars to simultaneously lie semirecumbent (k-value 0.027)

iv) Space required to thermoregulate

The minimal space allowance that can protect the welfare of pigs during transport, will be influenced by the microclimatic conditions, i.e. temperature and humidity inside the vehicle, the effectiveness of the ventilation system (while the vehicle is in motion or stationary) and by the ability of the animals to thermoregulate effectively.

Increasing space allowance can reduce the risk of heat stress by reducing the number of animals, thereby reducing the total metabolic heat and moisture produced by the animals on board a vehicle.



Dewey et al. (2009) reported increased truck internal temperature by 7°C when load density increased from 1 to 3 pigs/m². Additionally, increasing the space allowance helps to increase the evaporative cooling mechanisms via increased exposure of body surface areas to ventilation. However, no research was identified to provide specific quantitative information on appropriate space allowances in relation to thermal conditions. Additional information on microclimatic conditions (temperature, humidity and ventilation) during road journeys of pigs can be found in Section 3.5.3.1. From this section, it is also evident that pigs lack functional sweat glands, and rely on their behaviour to thermoregulate. One such thermoregulatory behaviour is full lateral recumbency, by which the pigs maximise the contact between their body and the flooring, and thus may increase heat dissipation (Bracke et al., 2020).

The minimum space allowance required for pigs to lie in full lateral recumbency, legs outstretched and without space sharing between the legs, can be calculated using a k-value of 0.047 (Petherick, 1983a,b) (Tables 22–24). This provides a space allowance that is similar to the space allowance calculated to enable animals to lie down and move between a standing and lying position and vice-versa, which can be calculated from a k-value of 0.046 (Petherick and Philips, 2009). However, the synchronicity of behaviour should be considered. Gerritzen et al. (2013) reported that the activity of pigs transported differed by trip, although inadequate data were presented to show this. However, it does suggest that space sharing and changes in activity within groups of transported pigs will likely take place, and therefore provision of space for all animals to lie down in a fully recumbent position may not be necessary for animals transported in a group, if there will be changes in activity within the group and space sharing. However, for animals transported in individual pens, such as boars, additional space may be beneficial to enable posture adjustment and stretching of limbs. Therefore, when calculating the space requirement for animals that are transported individually, it may be better to be closer to a k-value of 0.047.

Digital image analysis of the area occupied by finisher pigs calculated the minimum static space requirements ranging from 0.65 m² for 100-kg pig to 0.84 m² for an animal of 160 kg to be fully recumbent, without space sharing (which means that the space between the legs is included per animal), equivalent to calculating space with a k-value of 0.028 (Pastorelli et al., 2006). Planimetric analysis calculated a finisher pig (average weight 109 kg) lying laterally required a space allowance of 0.49 m² (Arndt et al., 2019), much lower than that calculated from allometric equations.

At warmer temperatures, insufficient space for animals to spread out makes it challenging for pigs to dissipate heat effectively by conductive (insufficient surface area to lie in full lateral recumbency) and convection heat loss (insufficient movement of air around the pigs) (Hayley et al., 2010). An increased number of animals within a vehicle also increases the production of heat and water vapour. In warm weather, this can lead to a microclimate around the animals that increases heat stress, resulting in greater requirements for thermoregulatory behaviour to increase heat loss (Mitchell and Kettlewell, 2008), while also making evaporative heat loss from panting less effective. As discussed in Section 3.5.3.1, the uneven temperature distribution and heat zones in vehicles (e.g. Xiong et al., 2015), means that the space required to be able to thermoregulate in principle may differ in different compartments, even on the same vehicle.

Increasing the space allowance by up to 20% has been suggested to allow animals on longer journeys room for thermoregulation (Council Regulation (EC) No 1/2005¹). It has, however, not been possible to find any scientific validation of this suggestion. Based on the data from Arndt et al. (2019), changing from sternal to lateral lying requires a 14% increase in space, which suggests that a recommendation of a 20% increase in space during warm weather conditions may be sufficient for pigs to be fully recumbent and thus perform thermoregulatory behaviour. The extent to which such an increase in space allowance will lead to reduced temperatures inside a vehicle, due to reduced heat production from fewer animals per m², is not known.

 Table 22:
 Minimum space requirement for weaner pigs to lie in full lateral recumbency (k-value 0.047)

Pig weight (kg)	m²/pig	kg/m ²
10	0.22	45.5
20	0.35	57.2
30	0.46	65.4
40	0.56	71.9

Pig weight (kg)	m²/pig	kg/m ²
100	1.03	97.3
110	1.10	100.4
120	1.16	103.3
130	1.23	106.0
140	1.29	108.7
150	1.35	111.2
160	1.41	113.6
170	1.47	115.9

Table 23:Minimum space requirement for finisher pigs to lie in full lateral recumbency (k-value
0.047)

Table 24:	Minimum space requirement for sows, gilts and boars to lie in full lateral recumbency (k-
	value 0.047)

Pig weight (kg)	m²/pig	kg/m ²
180	1.52	118.1
200	1.64	122.2
220	1.74	126.2
240	1.85	129.8
260	1.95	133.3
280	2.05	136.6
300	2.15	139.8
320	2.24	142.8

During cold conditions, pigs require sufficient space to be able to move away from cold areas, such as air movement at ventilation inlets. Otherwise, they could experience the WC cold stress, involving discomfort and potentially distress, caused by, for example frostbite. In colder temperatures, pigs will increase body contact with conspecifics, adjust postures to reduce contact with cold surfaces, and huddle together to reduce heat loss (Hillman et al., 2004). Increased sitting behaviour has been observed in cold weather transport (Goumon et al., 2013), which has been associated with attempts for the pigs to reduce contact with cold surface areas.

v) Space requirements to eat and drink, if feed and water are provided on the vehicle

Due to the tendency of pigs to develop motion sickness (see Section 3.5) and studies suggesting that transport without fasting is associated with increased mortality in finisher pigs (Averós et al., 2008), it is possible that the provision of feed on-board vehicles – provided during driving or when stationary – will increase death loss in pigs. This warrants further study before spatial recommendations can be given. Hence, in this section focus will be on provision of water.

The minimum space allowance considerations when transporting pigs with access to water need to consider the space required for provision of nipple drinkers, along with the ability for pigs to manoeuvre to access these. There is limited research on the provision of water to pigs on vehicles, and no research is available exploring the design and positioning of the access to the water nipples on the vehicle. Sufficient provision of resources in terms of water and space for the number of animals per compartment must also be considered, and this will likely be influenced by the pig category. For pigs transported in groups, there are no available data to indicate how space allowance influences water use on-board. Of the limited studies available, water disappearance (amount of water left in tank after journey) has not been found to differ with space allowance (Gerritzen et al., 2013). No impact of loading density (0.42, 0.5, 0.6 m²/pig) was reported by Chevillon et al. (2003) either.

B) Vertical space

Not only space allowance in the horizontal plane, but also in the vertical plane, may influence welfare of pigs during transport. Trucks used to transport pigs may have from 2 to 5 decks, allowing approximate deck heights from 120 cm to 60 cm, respectively, depending on the category of pigs transported. The majority of finishers are transported in multi-deck trucks with a deck height ranging



from 90 to 112 cm (Weschenfelder et al., 2012; Visser, 2014; Moak et al., 2022a,b). The height of compartments within which pigs are transported influences their ability to adopt a comfortable unimpeded posture. In addition, it is necessary for adequate temperature regulation and removal of noxious gases that the height of the compartment is adequate for effective ventilation to occur (SCAHAW, 2002). Finally, the compartment height may also affect the manoeuvrability and ability of the pigs to locate resources such as preferred orientation and water sources, but at the same time, unrestricted compartment height may facilitate unwanted behaviours such as mounting or fighting. Overall, the available literature on the relationship between deck height and pig welfare is very limited which constitutes a gap in knowledge.

Equations to estimate pig height have been proposed by Petherick (1983a,b): [height (m) = $0.156W^{0.33}$], where W is the weight of the pigs (in kg) and by Foldager et al. (2021): [height (cm) = $15.5 + 2.06W - 0.0500W^2 + 0.000514W^3$] (Table 25). The development of the equation by Foldager et al. (2021) focused on pigs of less than 40 kg of body weight.

Pig type	Pig weight (kg)	Average height (m) ^(a)	Height (cm) ^(b)
Weaner pigs	10	0.33	31.6
	20	0.42	40.8
	30	0.48	46.2
Market pigs	100	0.71	-
	120	0.76	_
	140	0.80	_
Heavy market pigs & gilts	160	0.83	_
	180	0.87	-
Sows & Boars	200	0.90	_
	220	0.92	-
	240	0.95	_
	260	0.98	-
	280	1.00	_
	300	1.02	_
	320	1.05	_

Table 25:Estimates of pig height (highest point of the back, constituting the highest point of the
body when standing in a natural position) by two equations

(a): Petherick (1983a,b).

(b): Foldager et al. (2021): Preliminary data, estimation using the polynomial equation only advised up to pigs of 40 kg.

The CATGP (2017) recommended best practice deck heights of the following: pigs 10–25 kg: 62 cm, pigs 100–120 kg: 88 cm, pigs > 120 kg: 100 cm. However, no scientific evidence to support this statement was found. In addition, there is evidence from North American trailer types that compartment temperature and RH may increase during stationary periods, when the deck height is less than 30 cm (for passively ventilated vehicles) or 15 cm (for mechanically ventilated vehicles) above the highest point of the pig's body (Weschenfelder et al., 2012; Moak et al., 2022a,b).

C) Summary of space requirements during journey

The spatial dimensions of compartments holding pigs during transport are of major importance for their welfare – in the horizontal as well as the vertical plane – and lack of space may lead to several WCs such as group stress, injuries, restriction of movement, resting problems and heat stress – all potentially leading to distress.

Due to a lack of available evidence from observations of pigs in transport over the relevant pig categories (weaners; finishers; and gilts, sows and boars combined), the allometric equation $[A = k \times W^{2/3}]$ (where A is m² per animal, and W is liveweight in kg) can be used to estimate the minimum space allowances during transport of any category of pigs.

Among the five biological functions of space, room to physically fit into the compartment in a standing posture was taken as a starting point of the assessment.

The ability to maintain stability on moving vehicles was the second biological function of space. For pigs, however, almost no studies have been found to provide evidence within this area, and the available data does not allow sufficient conclusions. Research is required in this area.



The third biological function of space was resting in normal semi-recumbent posture. If the space allowance, driving conditions and vehicle flooring provide comfortable conditions, all categories of pigs will attempt to lie down to some extent along a journey. At some point in time, it becomes necessary for pigs to have space to lie down to reduce fatigue. The available evidence suggests that a k-value of at least 0.027 is required to provide the physical space for all pigs in a compartment to lie down some simultaneously. There is evidence that provided with greater space, finisher pigs will lie down sooner on journeys, and lower skin damage occurs.

The fourth biological function of space is room to thermoregulate. According to the allometric calculations, a k-value of 0.047 is needed for pigs to be able to space out and lie in full lateral recumbency (producing no overlap or space sharing), but there is insufficient evidence identified to provide a specific quantitative information on the appropriate space allowance to thermoregulate under transport conditions.

For the fifth biological function of space, room to eat and drink, the tendency for pigs to develop motion sickness when fed just before a journey meant that in-transit feeding was not considered an option, and focus was given to the possibility to provide water during journeys. However, the current knowledge of the efficacy of water provision during transport of pigs is limited, and no research was identified to provide specific quantitative information on appropriate space allowances.

Regarding deck height, the available research is limited. Deck height is important for pigs of all categories to be able to stand freely, unhindered, to adjust their head and neck position, to allow pigs to manoeuvre inside vehicles, and also to allow sufficient flow of air around the pigs. A low deck height may lead to WCs such as injuries and increase susceptibility to heat stress. There are equations to estimate the height of pigs, but a lack of data to determine the appropriate deck height for different pig categories in order to protect their welfare during transport.

3.5.3.3. Thresholds for journey times

As previously mentioned in this Scientific Opinion and reviewed by Nielsen et al. (2011) and Cockram (2007), transport of animals is a complex stressor involving numerous aspects (related to the condition of the animals, their general biological characteristics, as well as the conditions under which the journeys take place including the duration), the majority of which to some extent may influence animal welfare. Despite the need for resulting recommendations and legislation, very few European studies have focused on the journey time (Faucitano and Lambooij, 2019).

Whether welfare issues arise during transport is not just dependent on the journey duration. In addition, there is not a linear relationship between journey duration and stress or animal welfare (Faucitano and Lambooij, 2019). Moreover, the journey duration, per se, is rarely the root cause of poor welfare, while other factors associated with time might lead to this result (Nielsen et al., 2011). It will depend upon multiple factors, including the category of pigs (e.g. age and condition), their fitness for transport, the quality of the journey (including vehicle design, space allowance, ventilation, the standard of driving and quality of the road), the microclimatic conditions and the associated handling and management of the pigs. When considering the implications of journey duration, it is important to consider the influence of each of the potential factors that can affect animal welfare.

Although quantitative limits often are included in legislation, there is no scientific consensus on: (a) the basis to be used to identify maximum journey durations; (b) what maximum journey durations should be specified; (c) what other factors should be considered when specifying a maximum journey duration; or (d) whether the current emphasis on using the intervals required to provide feed, water and rest is always the most appropriate way of specifying a limit on journey duration (Cockram, 2007).

In order to describe how the level of welfare develops over time during journeys, based on the above examination of the highly relevant WCs associated with the transit stage, the available research has been examined to identify what factors associated with transport have the potential to either increase or decrease the risk of WCs as a journey continues.

Cockram (2007) proposed that a rationale for a scientific justification of journey durations could be made based on one or more of the following criteria:

- a) there are aspects of welfare that are negatively affected after a specific journey duration, and thus stopping a journey before this occurs would help to minimise these;
- b) transported animals are exposed to continuous or periodic WCs, and restricting journey duration would minimise the duration of this exposure;



c) there are many risk factors associated with a specific form of transport that have the potential to negatively affect aspects of animal welfare; therefore, the risk that these will occur will increase the longer the journey takes to complete.

In this Scientific Opinion, the work carried out has been based on the approach suggested by Cockram (2007), involving the categorisation of highly relevant WCs for the transit stage into these three categories.

There are, however, several factors limiting this work: there have only been a limited number of studies that have investigated the effects of journey duration on the welfare of pigs, and the experiments available have often been often done using conditions that conform to what is considered best practice. For example, most studies were conducted in journey conditions that were close to or within the TNZ of pigs. As the quality of these journey conditions are likely to be high, and only fit and healthy animals were used, the results of many of these studies may not have identified major WCs associated with commercial journeys (as discussed by Cockram, 2007, 2019).

Multifactorial studies of commercial situations that apply epidemiological approaches to identify risk factors affecting specific outcomes, such as mortality or clinical deterioration, can sometimes identify a potential relationship between a WC as indicated by an ABM and journey duration (e.g. Thodberg et al., 2019), but often focus on rather extreme welfare end-points (such as dead-on-arrival (DOA), e.g. Malena et al., 2007) and not on the protection of animals from WCs. In this Scientific Opinion, signs of activation of coping mechanisms are taken as an indication of the presence of a hazard potentially leading to the corresponding WC.

The scenarios considered in this section refer to the transport of animals within the EU and take into account the recommendation made on microclimatic conditions (Section 3.5.3.1) and space requirements (Section 3.5.3.2). This means that the two WCs heat stress and restriction of movement are not considered highly relevant, and have not been part of the assessment.

The remaining highly relevant WCs during the transit stage are: motion stress, sensory overstimulation, group stress, injuries, prolonged hunger, prolonged thirst and resting problems. In addition, pain and/or discomfort associated with pre-existing or newly caused health conditions have been included. Below, the relationship between journey duration and these WCs are examined, ordered to align with the three different categories suggested by Cockram (2007): continuous or semicontinuous WCs, progressively developing WCs, and the more sporadic health conditions. Across studies underlying this assessment, it is important to consider in their interpretation whether the animals had limited access to water, and more restricted space allowance than permitted in EU legislation.

A) Motion stress and sensory overstimulation

Several studies involving transport of weaners or finisher pigs have found increased plasma concentrations of cortisol (Bradshaw et al., 1996a,b; Brown et al., 1999), which is an indicator of stress. For other pig categories, similar studies are not available or are very few.

The studies almost always only examined plasma concentration of cortisol before and after journeys, thereby involving many different WCs, and not describing the temporal relationship of the activity in the HPA-axis.

There are few studies available, where blood samples were taken during a journey, typically from catheterised animals. Bradshaw et al. (1996b) exposed 35-kg weaners to 8 h of being individually penned in a commercial pig truck, either stationary or while the truck was driven (Figure 10). Plasma cortisol increased rapidly after loading into the truck in both conditions, remaining elevated during the journey in the moving truck in comparison with the stationary treatment.







Bradshaw et al. (1996a) transported pigs over 2 days and found higher cortisol responses on the first compared to the second day, emphasising that also aspects of novelty influence the stress responses of pigs to being transported.

Pigs transported by road have been shown to exhibit motion sickness-related behaviours, such as foaming, chomping, retching and vomiting, along with increase in plasma lysine vasopressin (Bradshaw et al., 1996a,b, 1999; Randall and Bradshaw, 1998; Santurtun and Phillips, 2015). Consequently, it has been suggested that pigs are more vulnerable to motion stress than many other species transported for commercial purposes. No studies relating journey duration to the occurrence of motion sickness in pigs have been found.

During transport, pigs are exposed to vertical, lateral, and horizontal vibrations (See Section 3.5.2 [motion stress and sensory overstimulation]). Unpaved roads or roads with strong wind currents transmit a more significant amount of vibrations. Pigs' reaction to transport has been shown to be affected by road type: rough journeys result in higher cortisol elevations. Bradshaw et al. (1996a) transported pigs of 40 kg for 100 min on rough vs smooth roads and took blood samples via jugular catheters every 20 min (Figure 11). However, since the rough part of the journey was also the initial phase, the difference between the response in plasma concentration of cortisol between the two types of road may be confounded by time or novelty.



Figure 11: Plasma cortisol (nmol/L) in 40-kg pigs transported on rough vs smooth roads for 100 min, respectively. The pigs were blood sampled every 20 min by use of catheters (Bradshaw et al., 1996a)



Driving quality (e.g. speed) and conditions (e.g. poor roads, wind and precipitations) have been also positively associated to increased animal losses, carcass bruising, leg fractures, blood CK concentrations (indicator of fatigue) at slaughter and frequency of pale, soft and exudative meat (Dalla Costa et al., 2019; Passafaro et al., 2019; Zou et al., 2020; Driessen et al., 2020a,b; Romero et al., 2022).

Vibration associated with transport has been demonstrated to be aversive to pigs. Stephens et al. (1985) taught pigs to press a switch panel to turn off a transport simulator which produced noise and vibration. When the vibration component was switched off, pigs continued to switch the noise component off, though with a lower frequency, which indicates a greater aversiveness to vibrations than to sound. All pigs responded behaviourally to terminate the simulator. During the series of studies, pigs were always exposed to 1 h of simulations, thereby not providing evidence on the any temporal association between vibrations and aversion.

B) Group stress and injuries

Group stress is a highly relevant WC for pigs during the transit stage. Injuries in the form of skin lesions (scratches and wounds) are often the physical outcome of group stress in pigs, and therefore these two WCs are combined here.

When pigs are mixed, it is part of their natural behaviour to establish a new social hierarchy. This is done by social behaviour of which aggressive behaviours and fights are a central component, and may go on for days (Turner et al., 2006b; Fabrega et al., 2013). As described in Section 3.3, during pig transport, pigs are typically mixed at least once during the hours preceding the journey, and also often mixed at loading as vehicle compartment size and the size of home pens or 'pick-up' pens or other facilities are often not matched. This practice constitutes a hazard for animal welfare, and means that pigs will be exposed to the WC group stress during most journeys, except for boars, that are typically transported in single compartments.

Only a few studies have quantified pig behaviour during journeys, and even fewer have compared pigs that are mixed vs non-mixed in order to quantify group stress during journeys. In addition, almost all of the available studies involve finishers even though weaners (Magnani et al., 2014) and sows (Herskin et al., 2017) will also fight when mixed.

Gerritzen et al. (2013) transported finishers of ~ 110 kg on four 550 km journeys at each of two stocking densities (230 or 179 kg/m²), all fasted for 8 h pre-transport. Fighting occurred during all journeys. Most fights were observed at loading, in the first 2 h after departure and immediately following the driver's break after ~ 4 h of transport (Figure 12).



Figure 12: Number of fighting events recorded during 4 journeys transporting finishers at each of two stocking densities (230 or 179 kg/m²), all fasted for 8 h pre-transport. A 45-min driver break was held after ~ 4 h (Gerritzen et al., 2013)

Bradshaw et al. (1996b) transported 90-kg pigs (N = 12) to slaughter in a commercial pig truck, with a journey duration of 1.5 h. At loading, the pigs were either kept with pen-mates or mixed with unfamiliar pigs. During journeys with unfamiliar pigs, the pigs showed increased activity and increased fighting. The pigs were sampled for saliva cortisol before, in the middle of the journey and upon



arrival. For pigs transported with pen-mates, the saliva concentration of cortisol increased throughout the journey. Pigs transported with unfamiliar individuals showed higher saliva cortisol concentrations as compared to the familiar treatment in the middle of the journey, at the end, and after the journey (Figure 13). Thus, being transported with unfamiliar conspecifics is more stressful to pigs than transport with familiar individuals. However, no studies to describe the development of group stress or injuries in relation to journey time have been found.



Figure 13: Plasma concentration of cortisol in 90-kg pigs transported 1.5 h to slaughter either with pen-mates (dark bars) or when mixed with unfamiliar pigs (hatched bars) (Bradshaw et al., 1996b)

C) Prolonged hunger

Despite the existence of motivational tools to use in farm animals, including operant techniques (e.g. Jensen et al., 2015) to quantify affective states, such as hunger, these approaches have not been applied to transport conditions. In studies of the welfare implications of transport, carried out without access to feed, emphasis has been placed on physiological measurements that indicate a mobilisation of body energy reserves by monitoring peripheral blood concentrations of metabolites, as well as tissue concentrations of energy reserves, e.g. liver glycogen concentrations. Some studies have included post-transport measurements of feed intake, which are easier to interpret in terms of motivation. However, across transport-related studies involving either physiological indicators of hunger or measures of feed intake, the level of hunger has been assessed before and after journeys, and no studies to describe the development of hunger in relation to journey time have been found.

Due to the tendency for pigs to develop motion sickness (Randall and Bradshaw, 1998; Santurtun and Phillips, 2015), as described in Section 3.5), as well as the requirement for pre-slaughter fasting (EFSA AHAW Panel, 2020), most pigs are fasted before the initiation of journeys.

In his review of pig welfare during transport, Warriss et al. (1998) estimated a rate of loss in body weight of ~ 0.2% of liveweight per hour. Part of this body weight loss was due to the urination and excretion during transport. Indications of loss of body reserves started 9 h after the last meal with the reduction of carcass weight and 18 h after last meal for the liver weight loss. Associated with the liver weight loss is a reduction in liver glycogen content. This follows a logarithmic pattern (Warriss and Bevis, 1987), so that very little remains after 24 h, over 2/3 having been lost in the initial 12 h. According to Warriss et al. (1998), it is likely that pigs are feeling very hungry at this time.

Driessen et al. (2020) stated that to estimate the total fasting time, time without feed on-farm, loading time, transport time, unloading time, and time in lairage (for animals sent to slaughter) must be taken into account. To determine the start of fasting, one should start from the estimated moment of slaughter, or arrival for further fattening, and then work backwards to determine the duration to determine the start of the fasting period. In the final publication from the EU Transport Guides Project (Consortium on the Animal Transport Guidelines Project, 2017), it was recommended to differentiate



pre-transport fasting time depending on the expected duration of the future journey. The authors recommended to fast finishers and cull sows/boars for a minimum of 10–12 h before the animals are loaded onto the vehicle, if the expected journey duration is less than 8 h, and to fast them on-farm for a minimum of 5 h if the journey duration is expected to exceed 8 h. For pigs transported for further fattening, the authors recommended minimum 5 h of on-farm fasting in preparation of all journeys. No scientific evidence underlying these recommendations is provided, and no maximum fasting time recommendations were given.

To the best of our knowledge, no recent studies have identified pros and cons of pre-journey fasting, examined effects of fasting duration or established guidelines, for example taking into account different types of feed (wet feed vs dry feed fed from feed stations) in pigs transported for further fattening. Such knowledge would be valuable for the assessment of WCs including prolonged hunger and aggression as a consequence of hunger. Another potential WC of prolonged fasting, especially relevant for pigs not destined for slaughter, is the development of gastric ulcers. As reviewed by Driessen et al. (2020), earlier studies have suggested a positive relationship between overnight fasting and the occurrence of gastric ulcers at slaughter, but there are also studies failing to find an effect of fasting time on the prevalence of gastric ulcers (Eisemann et al., 2002; Dalla Costa et al., 2008). Whether the duration of pre-journey fasting influences the risk of development of gastric ulcers in pigs transported for further studies not known but cannot be excluded.

D) Prolonged thirst

Thirst is a sensation that motivates animals to seek and drink water to maintain homeostasis (McKinley and Johnson, 2004). If thirst is severe and prolonged, it can be associated with dehydration and weakness. Physiologically, thirst is initiated by an increase in the osmolality of body fluids and by a decrease in body fluid volume (De Araujo et al., 2003). Receptors detect the increased osmolality and decreased extracellular volume and stimulate activation of physiological homeostatic mechanisms to conserve body water and promote thirst to motivate the animals to drink (McKinley and Johnson, 2004).

The WC prolonged thirst describes a situation where an animal experiences a craving or urgent need for water, accompanied by an uneasy sensation (a negative affective state), and eventually leading to dehydration as metabolic requirements are not met. If access to water is limited, or prevented, then a significant and prolonged increase in the motivation to drink may result, the thwarting of which may be associated with a negative subjective experience of thirst (Fraser and Duncan, 1998; Jensen and Vestergaard, 2021).

In a study of lactating sows, kept loose-housed, Jensen et al. (2016) showed that behavioural signs of thirst increase with increasing duration of water deprivation (from 0 to 12 h of over-night deprivation. Based on observations of a post-deprivation reduced latency to drink and an increased compensatory water intake, the authors suggested that these behaviours were associated with a negative subjective experience of thirst. This suggestion was supported by observations of behaviour indicative of frustration and restlessness, such as sniffing and forcefully manipulating the water trough before it was opened, and reduced lying and inactivity during the 30 min before trough opening. These behaviours were mainly seen in sows deprived for water for 6 or 12 h. However, only a few studies have examined the intake of water during pig transport.

When water is available on the vehicle, there is evidence that pigs use drinkers during the journey, however, it is not clear how much is consumed vs. wasted. Frotin et al. (2002) determined the quantity of water disappearance by finishing pigs during a 24-h journey according to three types of watering (trough/nipples/bowl). The average water use (as assessed by water flow meter every 4 h of transport) varied between 2.4 and 4.5 L/pig, with more than 80% of water use during the transit stage (compared with stops) and greater use during the first half of the journey. Chevillon et al. (2003) reported that larger volumes of water were used when finishers were transported as compared to pigs staying on-farm, but did not quantify the water intake of the pigs, and reported significant spilling. In two studies involving finishers (Lambooij, 1983; Lambooij et al., 1985) showed that pigs drank only very small volumes of water during transport. In journeys of 26-31 h, pigs consumed an average of 0.65 L/pig (Lambooij, 1983). Finisher pigs transported for 44 h (the journey duration included several short stops, and one stop of a longer duration on the second day of the journey) with water provision used very little water (5.4 L/pig), and were observed to waste water by leaning on the nipple drinkers (Lambooij et al., 1985). Comparing their findings to a predicted normal water consumption of 7-20 L/day (Warriss, 1998), the authors concluded that provision of water drinkers onboard vehicles is not effective.



More recently, Gerritzen et al. (2013) reported that finishers of 110 kg used 0.9 L water/pig during a 550 km journey lasting almost 8 h. Whether the water was consumed or wasted, was not reported, and no clear change in water use from when the vehicle was stationary to when it was in transit, was observed. For weaners, transported for 14 h with water provision, no on-board drinking behaviour was observed (Magnani et al., 2014). When water was continuously available to weaners transported for 23 h, and water wastage was scored from pictures (as in Figure 14), the level of water wastage seemed to be quite variable between trips and between truck compartments (Cecilie Kobek-Kjeldager, Aarhus University, personal communication).



Figure 14: Picture taken after unloading of weaners of 20–25 kg transported at 0.18 m²/20 kg for 23 h as part of an experimental study in Denmark. The nipple drinker is marked on the picture, and water was continuously available. On some trips, and in some compartments, water wastage led to pools of water, despite the ample amount of bedding used. Photo: Cecilie Kobek-Kjeldager, Aarhus University

The provision of water on-board vehicles is, thus, an area that requires further study, in particular to determine if drinker use and water consumption occurs, to determine the appropriate pig to drinker ratio, and location of drinkers on board vehicles transporting pigs. Due to the lack of documentation for the efficacy of water provision on-board vehicles transporting pigs, the present assessment of the relationship between the WC prolonged thirst and journey duration, is not split into vehicles with or without drinkers.

The capacity of pigs and other non-ruminant livestock to respond to water deprivation is low (as reviewed by Vogel et al., 2019). Warriss et al. (1983) reported increased water consumption in lairage for pigs transported for 6 h compared with animals transported only 1 h. Brown et al. (1999) evaluated the effect of 8, 16 and 24 h of transport (without stops and water availability) in pigs of 80–100 kg. Total protein and albumin levels increased progressively, but differences were only significant after 24 h. The authors concluded that dehydration developed gradually. When observing the behaviour of the pigs during a 6-h lairage period after unloading, all three journey durations led to temporal patterns of drinking behaviour, involving what seemed to be giving higher priority to drinking instead of resting or exploring, the longer the journey the pigs had experienced. In addition, water intake in lairage was higher for pigs transported 24 h than for the two other journey durations.

Becerril-Herrera et al. (2010) transported gilts and barrows to slaughter in Mexico by use of commercial trucks. The authors did not report pig weight, ambient temperatures, level of pre-transport fasting, road conditions or space allowance. The pigs did not have access to feed or water during journeys of either 8 or 16 h. Based on quantification of haematocrit from blood samples taken before and after the journeys, the authors concluded that signs of dehydration were present for both journey durations.

Goumon et al. (2013b) studied the effect of 6, 12 and 18 h of transport and of vehicle compartment on the physiology and behaviour of pigs in summer and winter in western Canada, and highlighted that, at the lairage, drinking behaviour did not differ between transport duration treatments in summer, while in winter, pigs from the 18 h treatment drank more than pigs that were transported 6 or 12 h, thus showing greater evidence of thirst.



18314732, 2022, 9, Downoaded from https://efsa.onlinelibrary.wiley.com/doi/10.2903/j.sa.2022.7445 by Universidad De Zaragoza, Wiley Online Library on [3001/2025]. See the Terms and Conditions (https://onlinelibrary.wiley.com/terms-and-conditions) on Wiley Online Library for rules of use; OA articles are governed by the applicable Creative Commons License

E) Resting problems

Irrespective of the space provided for the animals, the movement of a vehicle obliges pigs to make continuous postural adjustment to acceleration changes, in order to maintain their balance and avoid falling. In addition, when pigs are mixed at or before loading, in some animals, the motivation to rest may be thwarted by other animals fighting in the compartment or by disturbance from attacking individuals. Thus, even if the space allowance is large enough for pigs to lie down during journeys, they may still experience the WC resting problems, potentially leading to fatigue or to pigs being non-ambulatory – a condition not uncommon in pigs upon arrival at slaughterhouses, at least in some geographical regions (as discussed by Garcia et al. (2019)). Fatigued pigs are pigs without obvious injury or disease showing signs of severe acute stress that result from exhaustion.

As mentioned in Section 3.5.3.2, studies of behaviour of finishers during transport have shown that during short (2 h) journeys, pigs spend a numerically greater proportion of observations standing (mean: 46%), than lying (8%), (Torrey et al., 2013a). On longer journeys (8 h), pigs have been observed lying for a numerically greater proportion of observations (lying: 52% vs. standing: 16%) (Torrey et al., 2013b).

Brown et al. (1999) evaluated the effect of 8, 16 and 24 h transport (without stops and water availability) in pigs of 80–100 kg, transported at space allowance of $0.50-0.54 \text{ m}^2/100 \text{ kg}$. When observing the behaviour of the pigs during a 6 h lairage period after unloading, the authors stated that the pigs seemed tired, however without direct quantification thereof. Gerritzen et al. (2013) transported finishers of ~ 110 kg on four 550 km journeys at each of two stocking densities (230 or 179 kg/m²), all fasted for 8 h pre-transport. The authors observed that, during the driver break (45 min of stationary vehicle, 4:15 h after the start of the journey), up to 80% of the pigs were recumbent, irrespective of loading density. The authors considered such an urge to lie down as an indication of a motivation to rest, which was presumably restricted when the vehicle was moving.

Blood and tissue measurements indicative of exhaustion of body energy reserves (e.g. hypoglycaemia and reduced liver and muscle glycogen concentration) and the accumulation of metabolites, e.g. blood and muscle lactate concentration (Frese et al., 2016) in response to prolonged and anaerobic exercise, might be associated with fatigue (Terlouw and Bourguet, 2022). Yu et al. (2009) reported increased blood plasma CK and LDH levels after transport of finishers for 1, 2 or 4 h, respectively. The longest journeys were associated with increased serum CK activity.

The effects of journey duration on the welfare of pigs were investigated by Aradom et al. (2012) in journeys lasting 4, 8, or 12 h, during two seasons. In their study, involving pigs of 100 kg, transported at 0.55 m²/pig, a continuous increase in serum CK was associated with the prolonged journeys, suggesting increasing muscular fatigue with longer journey time (Figure 15).





In a study from Western Canada, Sommavilla et al. (2017) transported finishers at $\sim 0.37 \text{ m}^2/100 \text{ kg}$ in pot-belly trailers, and studied the effects of the season and journey duration (6, 12 or 18 h), and found that blood CK was greater in pigs after 18 h of transport, showing a linear increase with the increasing transport time.

Among the available pig studies, the majority have focused on non-ambulatory pigs as a sign of fatigue. The vast majority of the available studies are from North or South America, and there are



examples of surveys suggesting that relatively short journey times (30 min to 4 h) was one of the factors influencing the risk of non-ambulatory pigs (Sutherland et al., 2009), but also studies finding no effect of journey time (comparing < 1 vs. 3 h) on the prevalence of non-ambulatory pigs (Pilcher et al., 2011). In an observational study involving more than 300 loads of finishers shipped from 60 farms to three different slaughter plants, where the prevalence of non-ambulatory pigs was reported to be 0.37%, Dalla Costa et al. (2019) showed that the probability of becoming non-ambulatory was greater for every additional hour that the pigs spent during transport.

F) Summary on journey duration

Regardless of how optimal the conditions of the journey provided are, pigs can potentially be exposed to a number of hazards during transport that might, either on their own or in combination, result in impaired animal welfare. The amount of time the animals are exposed to these hazards is dependent on the journey duration. Any aversive effects of resting problems, or reduced availability of, or restricted access to water and feed, are likely to increase with journey duration and could interact with other factors, such as temperature, that might also change during a journey.

Above, the available research has been appraised to identify relationships between journey duration and highly relevant WCs. The information is summarised below. Based on estimates of risk, the prevalence and severity of the WCs, a table (Table 26) has been created to show the estimated journey duration after which the WCs are expected to be present. The assessment of journey duration takes as a starting point that recommendations on microclimatic conditions and space allowance are followed.

Summary – **motion stress and sensory overstimulation:** As soon as a vehicle starts moving, and during all time when it is moving, all pigs are to some extent exposed to motion stress, and often also, at least periodically (repeated intermittent), to sensory overstimulation. As a consequence of the vehicle motion, animals experience stress potentially leading to fatigue and other negative affective states such as fear and distress, due to the forces exerted as a result of acceleration, braking, stopping, cornering, gear changing, vibrations and uneven road surface. Motion stress is regarded as a highly relevant WC in the transit stage. The prevalence is high, as motion stress is likely to affect all animals in a moving vehicle. The duration of the WCs depends on journey duration and onset of vehicle motion. For pigs, the tendency to develop motion sickness means that this WC is a major consideration. Motion stress will vary over time depending on the journey conditions, but the severity of its effects will most likely increase over time and may eventually lead to fatigue. Based on the constant presence of motion stress, it is not possible to estimate a temporal cut-off for onset of this WC after initiation of the transit stage.

Summary – **group stress and injuries:** Group stress and resultant injuries (skin lesions) are regarded as a highly relevant WC in the transit stage for pigs exposed to the hazard of mixing at any pre-journey stage. The prevalence is high, as mixing is industry practice. Group stress likely affects all animals in a moving vehicle, however with a variation in severity, as not all pigs will be involved in fighting to the same extent. The duration of group stress depends on journey duration, but the temporal development is not known. Severity is expected to increase with increasing duration, as the lack of resting (in fighting animals, and animals being disturbed by other animals fighting) may become problematic for the animals. Group stress may lead to fatigue. No studies to describe the development of group stress or injuries in relation to journey duration have been found.

Summary – **prolonged hunger:** The WC prolonged hunger is regarded as highly relevant in the transit stage. The prevalence is expected to be very high, as pigs – especially animals sent for slaughter – likely have been fasted before the initiation of the journey. The severity of hunger will develop further during the journey. The duration of hunger depends on fasting practices, journey duration and availability of feed in transit, and severity is expected to increase with increasing duration, as the need for feed becomes more problematic for the animals. Prolonged hunger may lead to frustration, exhaustion and a weakened condition. The available data does not allow a detailed determination of the interval between journey start and initiation of prolonged hunger. Based on the available knowledge on physiological indicators of hunger (e.g. liver glycogen) as well as behavioural studies of post-journey eating behaviour, hunger is expected to be present after 12 h of feed deprivation, which (under current practice), for a large proportion of pigs transported to slaughter, corresponds to the initiation of the journey.

Summary – **prolonged thirst:** The WC prolonged thirst is regarded as highly relevant in the transit stage. The prevalence may be high, if water is not provided to the animals or they, for some reason (such as lack of familiarity or fear of other animals) are not able to obtain enough water. So far, very limited documentation for proper intake of water, even in journeys fitted with drinkers, is available. Depending on factors such as time off water before journey start and/or microclimatic



conditions before and during the journey, pigs may not be thirsty during the initial phase of the journey, but thirst will develop over time if water is not freely accessible. The duration of prolonged thirst depends on accessibility of water and journey duration, and severity is expected to increase with increasing duration, as the need for water becomes problematic for the animals. Prolonged thirst may lead to dehydration, frustration, discomfort and suffering. The available data does not allow a detailed determination of the interval between journey start and initiation of prolonged thirst. Based on the available knowledge, it is possible that the increase in total plasma protein concentration (indicative of a reduction in plasma volume) as well as behavioural changes seen in some studies within 8 h without water might be associated with thirst, whereas for example lactating sows most probably are thirsty after a shorter interval.

Summary – **resting problems:** Resting problems are regarded as a highly relevant WC in the transit stage. The prevalence is at least moderate, as resting problems may affect a large proportion of animals in a moving vehicle, even when space for lying is available. Duration depends on journey duration, and severity is expected to increase with increasing duration, as the lack of resting becomes problematic for the animals. Resting problems may lead to fatigue. Irrespective of the space provided, resting problems may arise due to exposure to vehicle motion, as well as associated with the consequences of group stress – the participation in fighting or the disturbance from fighting conspecifics. Based on the constant presence of motion stress, it is not possible to estimate a temporal cut-off for onset of this WC after initiation of the transit stage.

Other summarising considerations: In addition to the WCs summarised above, the risk of pigs experiencing pain and/or discomfort, as well as the severity of it, will also increase with journey time. This may happen if a pig has a pre-existing, but non-identified, painful condition. Even though this should not happen, it is not always possible to identify pathological conditions in pigs while they are on-farm, that could subsequently affect their ability to respond to transport (as surveyed by Herskin et al. (2020) and Thodberg et al. (2020) among sow farmers and livestock drivers, respectively).

In addition, pigs which did not show a health condition before the journey may get injured during the journey due to, for example aggressive behaviour, and the pain and discomfort from such conditions will continue, and likely worsen, until the animal can be unloaded. In this weakened state, pigs may be less able to cope with the challenges associated with transport, and their condition may deteriorate with time and journey duration (as shown for cull sows in Denmark after journeys up to 8 h (Thodberg et al., 2019)).

The pain and/or discomfort from both types of the above-mentioned health conditions are not expected to be prevalent, but for the affected animals, the consequences may be severe, and will often develop over time. The duration of these negative affective states will depend on journey duration, as they cannot be terminated until the journey is stopped (or sometimes not until post-transport healing). During a journey, such health conditions may lead to suffering. It is, however, not possible to establish a temporal cut-off for when pain and/or discomfort may start.

Among the available studies of welfare of pigs during transport, many have focused on pigs dying in transit. Pigs dying in transit most likely experience negative WCs during their last hours alive. In Italian and German epidemiological studies, journey duration (more than 2 h journeys in summer, and more than 8 h journeys in winter) contributed to increased mortality rate (Werner et al., 2007; Vitali et al., 2014).

Outside Europe, information on the effects of journey duration on pig mortality is available from the USA and South America. Similar to the effects of journey duration on non-ambulatory pigs, some studies found shorter journeys to be one factor increasing the incidence of mortality (Sutherland et al., 2009). However, in a survey carried out on about 40,000 pigs, Kephart et al. (2010) found no effect of journey time on rates of mortality (0.06%) after transport to a packing plant comparing two journey durations (< 2.5 h vs. \geq 2.5 h; ranging from 1 to 14.5 h). Similarly, Pilcher et al. (2011), examining effects of journey duration (< 1 h vs. 3 h) and floor space on the prevalence of transport losses found no effect of either journey duration or floor space on the mortality. There are, however, also studies showing positive relationships between the occurrence of dead-on-arrival and journey duration (e.g. Dalla Costa et al., 2019; Romero et al., 2022).

Table 26 summarises the estimated interval from journey initiation and until presence of the highly relevant WCs.
Type of welfare	Welfare consequence	Development over time	Expected development over time
Continuous or semi-continuous	Motion stress	Continuing throughout the transit stage	Severity will increase over time leading to fatigue
	Sensory overstimulation	Sensory overstimulation repeated intermittent	Can lead to fear and distress
	Group stress and resultant injuries (skin lesions)	Continuing throughout the transit stage, but development over time not known	Group stress may lead to fatigue, injuries will lead to pain and/or discomfort.
	Resting problems	Continuous throughout the transit stage	Severity will increase over time leading to fatigue
Progressively developing	Prolonged thirst	Available information shows that behavioural and physiological signs of thirst can be present after 8 h of transport without effectively accessing water.	Severity will increase with time, leading to dehydration
	Prolonged hunger	Available information shows that behavioural and physiological signs of hunger can be present after 12 h of feed deprivation. Depending on the pre- transport fasting this may correspond with initiation of the journey.	Severity will increase with time, leading to weakening and exhaustion
Sporadic	Pain and/or discomfort from health conditions	May start at any time if pre-existing health conditions are present or occur during transport	If present, severity will increase with time, leading to suffering

Table 26:	Welfare consequence	es estimated to	start and dev	elon over	iournev	v time
	wenale consequence	lo countateu to			journe	

3.6. Journey breaks

At some stage during a journey, pigs will need feed, water and rest in order to avoid WCs. This can be done in two ways: (1) by providing the animals with feed, water and rest while the truck is stationary; and (2) by unloading the animals and providing them with feed, water and rest there (at a CP). Below, the different possibilities are discussed, and hazards, WCs, preventive and corrective/ mitigating measures are assessed for the use of CPs.

3.6.1. Provision of water and/or feed on the vehicle while stationary

Regarding provision of feed, the tendency for pigs to develop motion sickness (Randall and Bradshaw, 1998; Santurtun and Phillips, 2015), as described in Section 3.5.2, means that provision of feed to pigs on stationary vehicles during breaks in driving, likely will be difficult, as the journey probably cannot be continued until hours after the feeding of the animals. Research is needed to clarify costs and benefits of such practice.

3.6.2. Unloading of the pigs into a pen where water and/or feed is provided for a period of time before reloaded to continue the journey

Theoretically, unloading pigs into a pen allows the animals to have access to resting, watering and feeding areas in order to mitigate the WCs of transport. However, stopping a journey to unload the animals to provide a period of rest, feed, and water involves a number of hazards relevant for animal welfare such as the risk of stress, injury and infectious diseases. Some of these can be mitigated by the duration of the stay, some by offering high quality conditions to the animals, whereas others, such as the novelty of the surroundings, the extra unloading and reloading, and the mixing of unfamiliar animals as well as the inherent biosecurity risk, cannot be avoided. Thus, whether there are always beneficial effects for the welfare of pigs for a long journey to be interrupted to offer feed and water while unloaded is equivocal.

However, as most research involving long journeys for pigs have been carried out in North America, where journeys are typically not interrupted, the available knowledge in this field relating interruptions in long journeys with the welfare of pigs, is very limited. This limits the possibility to reach conclusions



and recommendations. Below, the current knowledge and use of CPs to unload pigs during long journeys are discussed.

3.6.3. Control posts

CPs are specialised livestock facilities, usually private, where animals can be offered a journey break after reaching the maximum journey time. Currently, a stay in a CP has to be 24 h before the journey can be continued. The CPs are used exclusively for receiving, feeding, watering, resting, housing, caring for and dispatching transient animals. The operators of CPs are obliged to ensure that the animals receive the necessary care, feed and water, and before animals leave a CP, an official veterinarian must verify that they are fit to continue their journey (Schmid and Kilchsperger, 2010). Also, CPs typically offer facilities for vehicles, drivers and competent authorities. In this Scientific Opinion, the section on CPs includes all kinds of actions and management of the animals, which take place during the interval from when the animals have been unloaded from the vehicle, and until reloading is started to continue the onward journey. Loading and unloading are covered in Section 3.4. For management, planning and logistics issues, readers are advised to consult the recommendations of the EU Transport Guidelines (Consortium of the Animal Transport Guides Project, 2017).

Two EU projects were funded by DG Sante of the European Commission to renovate and promote high quality CPs in the EU and to develop an EU-wide animal transport certification system. The first project 'Renovation and promoting high quality control posts in the European Union' was concluded in September 2013, while the second initiative 'Development of an EU wide animal transport certification system and renovation of control posts in the European Union' was concluded in 2015. Both projects showed that many CPs had problems to be financially profitable and needed rebuilding or renovating to reach high quality standards. To the best of our knowledge, newer data on CP standards are not available. In this Scientific Opinion, the recommendations from the projects are referred to as Porcelluzzi (2013), but it is important to bear in mind that this is not a reference to a scientific study.

The photo below is taken from an anonymous CP in France, but it is not known whether or not the conditions observed there are representative for the standard of pig CPs across (or outside) EU (Figure 16).



Figure 16: Photo taken at anonymous control post authorised for pigs in France. The pigs are 20- to 25-kg weaners transported on a journey from Denmark. Photo: Mette S. Herskin, Aarhus University

Porcelluzzi (2013) stated that a resting period in a CP is the most appropriate solution to the challenges of long journeys in terms of animal welfare, because the animals are supposed to be getting adequate rest, feed and water, according to their needs, and to access comfortable bedding areas, as well as feeding and watering resources. Based on this, the authors of the report concluded that the use of CPs is an efficient means to improve animal welfare. However, as mentioned by Padalino et al. (2018), this conclusion does not seem to be supported by scientific data, as currently very limited studies on cattle welfare at CPs in the EU are available.

Irrespective of the duration of the stay, a stay in a CP may involve several hazards for animal welfare. Whether the WCs of the use of CPs are increased by successive stays (during very long

journeys where pigs are unloaded for rest more than once) has not been examined but cannot be excluded. In this chapter, the main focus is on WCs, their hazards, recommendations to prevent hazards, as well as to mitigate WCs connected to the stay at a CP. In Section 3.8 (Specific scenarios: Special Health Status), potential WCs from omitting CPs and letting pigs be fed, watered and rested on the vehicle as an alternative, are discussed.

The text of this chapter focuses on weaners sent for further fattening, and to a lesser extent young boars or gilts transported for breeding, as these are the main pig categories currently experiencing journeys with rest stops. Other pig categories are mentioned briefly when relevant.

3.6.3.1. Current practice

According to the list retrieved from https://www.pvd.gov.lv/lv/media/539/download in January 2022, there were 62 CPs for pigs on the EU approved list. CPs are distributed unevenly across MSs, and few countries (e.g. Germany, Italy, Poland, Ireland and France) host the majority of them. No list of approved CPs outside the EU exists, which has been described as a concern for competent authorities, as they may have difficulty to verify whether the place outside the EU, where animals are planned to stop to be unloaded, has suitable conditions for this.

3.6.3.2. Highly relevant welfare consequences

The highly relevant WCs are: group stress, handling stress, heat stress, injuries, prolonged hunger, prolonged thirst, resting problems and restriction of movement. The WCs and associated hazards are explained below. The presence and severity of these WCs depend on the management (e.g. cleaning and disinfection procedures, availability of reservation system), the housing conditions (type of barn, ventilation, bedding, etc.), the equipment (e.g. ramps to (un)load, drinkers, feeders) and training of the staff at the CP. In addition to the WCs listed above, another welfare hazards related to the use of CPs are animals developing health conditions or becoming unfit for further transport. Below, this hazard is listed together with other welfare hazards associated with the different WCs. In addition, preventive and corrective/mitigating measures are listed.

i) Group stress and injuries

Group stress is a highly relevant WC for pigs in a CP. Injuries in the form of skin lesions (scratches and wounds) are often the physical outcome of group stress in pigs, and therefore these two WCs are combined here. The main hazards for group stress and injuries are:

Mixing unfamiliar animals: Due to the biology of pigs, mixing will unavoidably lead to aggression and fighting, which may impair animal welfare and lead to another WC, injuries, in terms of skin lesions. In addition, taking part in fighting prevents pigs from resting, and the presence of fighting individuals in a pen can limits the possibility of pen-mates to acquire the rest that they need.

• PRE: In the European handbook (Porcelluzzi, 2013), it is recommended that CP pens are divided so that animals can be kept in the same groups as in the vehicle, and thus avoid mixing. For pigs, however, the group size in vehicles is likely often lower than the group size in CP pens, and mixing is probably difficult to avoid with the current CP layout. Therefore, a larger space allowance is recommended in CP pens than during on-farm conditions.

Limited resources: In general, the consequences of fighting in terms of welfare are increased when resources are limited. After many hours in a vehicle, pigs arriving in a CP will be hungry (although neither weight loss nor feeding motivation have been quantified in CPs so far), and likely also thirsty. The presence of fighting pen-mates may limit the access of themselves as well as other pigs to the resources. In addition, limited resources, e.g. fewer eating places than pigs, and too few drinkers, may increase the likelihood of aggression.

• PRE: Feed and water should be available in adequate quantities and homogenously distributed to avoid fighting for resources. In addition, the presence of bedding and/or other materials to attract the exploratory motivation of pigs may inhibit or delay aggression. However, these effects have not been documented in CPs.

Corrective/mitigating measures of group stress

If signs of group stress, such as aggression, are observed in the CP, animals from different groups should be separated as much as possible, to allow a proper rest. In addition, enrichment (e.g. straw) may be provided to distract the pigs.



ii) Handling stress

Handling stress is a highly relevant WC in CPs. The hazards contributing to handling stress (i.e. inappropriate handling), preventive and mitigation measures are identical to the ones previously described in the preparation and loading/unloading phase (Sections 3.3 and 3.4).

iii) Heat stress

As reviewed in Section 3.5.3.1, pigs are sensitive to heat stress. The European handbook (Porcelluzzi, 2013) recommends that pigs at CPs are always kept in their TCZ. See Section 3.5.3.1 for recommendations on TCZs for the pig categories typically transported via CPs. The main hazard for heat stress at a CP is:

High/low temperature in the CP: If CPs are not equipped with proper ventilation and temperature control, there is a risk (however unquantified) that pigs will experience heat stress while kept in the CP pen. During winter months, there is also a risk that the animals may experience cold stress. For animals arriving from a vehicle, where it can also be difficult to keep the microclimatic conditions in the recommended comfort zone, the risk for WCs is likely higher than during periods where the outdoor temperature allows pigs to be transported within their TCZ.

• PRE: To prevent this hazard, pigs should be transported in their TCZ, as explained above in Section 3.5.3.1, where the upper threshold of the comfort zone and the upper threshold of the TNZ (UCT) are identified as the basis of a quantitative recommendation on microclimatic conditions during transport of different categories of pigs. CPs should be adequately designed and ventilated to maintain pigs in their TCZ. During cold weather conditions, CPs should provide indoor heating, and/or sufficient quantity of bedding and shelter so pigs can bury themselves into the bedding and maintain their body temperature at the required comfort levels.

Corrective/mitigating measure of heat stress

When animals show signs of heat stress, the effective temperature in the CP should be lowered. This can be achieved by increasing ventilation and/or by increasing space allowance. Animals can be cooled down using water sprinklers, showers, or equivalent.

iv) Prolonged hunger

See Sections 3.5.2iv and 3.5.3.3C for examination of the WC prolonged hunger during the transit stage. The main hazards for this WC in a CP are:

Time off feed: Pigs will typically have been off feed for at least 24 h when they arrive at a CP. Hence feeding motivation is (although yet un-quantified) most likely very high.

• PRE: The European handbook (Porcelluzzi, 2013) recommends that the minimum quantity of feed provided should correspond to the amount of food required for the body maintenance, and that feed must be of homogenous quality to avoid competition. In cases where *ad libitum* feeding is used, the report recommended one feeding place per 10 animals. If animals are not fed for *ad libitum* intake, one eating place per animal is recommended. To the best of our knowledge, these recommendations are not based on scientific data obtained in CPs. In addition, as mentioned above, if all feed-motivated pigs cannot eat simultaneously, it may lead to fighting and frustration. As the pigs will start a new transit stage after re-loading it is, however, important to make sure that they are not overfed at the time of re-loading, in order to prevent motion sickness. Research is needed to examine costs and benefits in terms of CP feeding management of pigs. See Section 3.3.2 for a discussion of time off-feed before the initiation of journeys.

Competition for access to feed: Feeding during journey breaks may cause competition between animals, and the stronger individuals may exclude the weaker ones.

• PRE: It is important that feeding space is enough for all animals to have access to feed simultaneously.

Corrective/mitigating measures of prolonged hunger

In the case that prolonged hunger is suspected in the CP, additional feed should be homogenously distributed among the animals. If re-loading is soon to take place, it will need to be delayed, allowing the pigs to eat first, and then digest before being reloaded.



v) Prolonged thirst

Prolonged thirst is regarded as a highly relevant WC in the CP stage even though pigs may arrive from journeys with access to water, because, so far, very limited documentation for proper intake of water, even in journeys in vehicles fitted with drinkers, is available. See Sections 3.5.2v and 3.5.3.3D for examination of the WC prolonged thirst during the transit stage. The hazards for this WC in a CP are:

Time off water: Irrespective of whether pigs have been drinking during the journey or not, the intake of feed at the CP will likely stimulate drinking motivation (however, not yet quantified).

• PRE: In the European handbook (Porcelluzzi, 2013), it is recommended for CP pens to be equipped with drinking devices and drinker height designed for the animal age group, and at least one drinker per 10 pigs is recommended. However, whether this recommendation is enough to avoid WCs is not known. The ratio of pigs to drinkers in CP is an area that should be studied, considering that animals may be thirsty when arriving, and all animals need to have the opportunity to drink and rehydrate during their stay at the CP.

Competition for access to water: Drinking during journey breaks may cause competition between animals, and the stronger individuals may exclude the weaker ones.

• PRE: It is important that drinking space is enough for all animals to have access to water simultaneously.

Corrective/mitigating measures of prolonged thirst

In the case that prolonged thirst is suspected in the CP, additional water should be homogenously distributed among the animals.

vi) Resting problems and restriction of movement

The physical conditions in a CP should allow for the pigs to be able to rest. Due to the novelty of the conditions, as well as mixing-induced group stress, pigs may experience thwarted resting motivation, and thus also be restricted in their movements. The main hazards for these WCs are:

Insufficient horizontal space (space allowance): The functionality of a CP depends on the ability of the animals staying there to rest. During a stay in a CP, it is crucial that pigs have time and possibility to eat, drink and rest. As mentioned above, pigs will most often be mixed at arrival, which means that some individuals likely will fight – sometimes during prolonged periods. The aggressive behaviours take up space.

• PRE: It is recommended that all animals in a CP pen should be able to rest at the same time. To the best of our knowledge, no studies have focused on resting behaviour of pigs in CPs. Both the European handbook (Porcelluzzi, 2013) and the Consortium on the Animal Transport Guidelines Project (2017), mention the risk of inadequate space allowance in CPs. In these guides, weaners of 20-30 kg are recommended an area of 0.26 m²/pig, whereas gilts are recommended 0.63 m²/pig, and sows/boars 1.22 m²/pig. The reasons for these recommendations are not stated, and no evidence is provided to suggest that animal welfare is protected, as these numbers are lower than what is required on-farm. Whether the recommendations allow all pigs to rest when motivated to do so (because of individuals participating in fighting, exploring or seeking food and water, thereby disturbing their rest-seeking pen-mates) is not known. Research is needed to examine this, as well as to further develop CP-conditions to improve animal welfare, for example involving provision of functional areas for the pigs. A reasonable starting point would be to allow pigs at least the space required during housing conditions on farm (where they have time to wait for others to finish eating/drinking.

Corrective/mitigating measures of resting problems

In the case that resting problems or restriction of movements are suspected in the CP, a detailed inspection of the facilities should be performed, and potentially extra space and bedding should be provided.

Animals developing health conditions or becoming unfit: The flow of animals at a CP is a major issue with respect to the spread of diseases in and outside the EU (Transport Guidelines Project Consortium, 2017), with potential collateral consequences for animal welfare. Disease is always a

18314732, 2022, 9. Downloaded from https://efsa.onlinelibrary.wiley.com/doi/10.2903/j.efsa.2022.7445 by Universidad De Zaragoza, Wiley Online Library on [3001/2025]. See the Terms and Conditions (https://onlinelibrary.wiley.com/terms-and-conditions) on Wiley Online Library for nales of use; OA articles are governed by the applicable Creative Commons License



challenge for animal welfare (as reviewed by Broom (2006)) and can lead to animals needing treatment or even turning unfit for transport during a journey. When journeys are prolonged, changes in the clinical condition of animals are a significant risk to their welfare. However, no scientifically published data to document changes in the clinical condition of pigs during and/or after journeys involving rest stops have been found. Hence, it is not possible to assess the size of this risk. This means that the use of mid-journey rest periods where pigs are unloaded to rest, feed and water have the potential to be counterproductive if inadequate attention is given to biosecurity. There are numerous infectious diseases to which pigs are susceptible that require biosecurity practices to prevent transmission. When pigs are unloaded during a journey, there is an increased risk of contact with infected animals or facilities, e.g. inadequately cleaned and disinfected areas. Pathogens can be transmitted to susceptible pigs by direct contact with infected animals, by mechanical transfer on people, non-susceptible animals, birds, vehicles and fomites, and by aerosols (Neumann and Hall, 2019). Among relevant examples are faecal pathogens (Salmonella enterica biovars and Escherichia coli) transmitted through shedding (Isaacson et al., 1999). Many other pathogens including porcine epidemic diarrhoea, porcine reproductive and respiratory syndrome, foot and mouth disease or African swine fever can be potentially spread while unloading animals, if adequate biosecurity measures are not put in place (Lowe et al., 2014; Filippitzi et al., 2018 and Beltran-Alcrudo et al., 2019). To the best of our knowledge, whether the physical conditions at CPs allow proper veterinary inspection (checking individual animals can be difficult when pigs are kept in large groups) or not, is not known.

To date, only one study has been identified reporting data on mortality or morbidity (including animals declared unfit for transport by official veterinarians at the CP). Padalino et al. (2018) reported the official recordings of mortality and morbidity in an Italian CP during 2010–2015, a period where 111,536 animals of different species passed through. Of these, 4.8% (corresponding to 5,333) were pigs, the majority of which were either light (110–120) or heavy (150–160 kg) finishers and the rest were > 200 kg animals. According to the data reported by Padalino et al. (2018), no pigs were recorded as 'unfit for transport', no pigs were reported dead, and no morbidity was recorded either. Whether these findings are a true picture, reflect lack of proper inspection, or can be explained by the relatively low sample size is not known.

- PRE: Staff should be trained on the inspection of animals at the CP to detect signs of weakness or illness on fatigued or injured animals that may compromise their fitness for further transport (see Section 3.3.3). Animals showing signs of weakness or disease should be inspected and treated accordingly, as described below.
- PRE: The risk of spread of diseases can be limited by managing an all-in all-out principle at CPs as well as by thorough cleaning and disinfection between batches.

Corrective/mitigating measures for animals developing health conditions

Animals showing signs of weakness or disease should be treated appropriately, preferably in a hospital pen or similar. These animals may require specific handling and treatment, so contingency plans should be in place for injured and sick animals. Animals that are unfit for further transport should not follow the consignment at re-loading, but be slaughtered, treated or euthanised according to the prognosis of their condition.

3.6.4. Journey break – summarising considerations

At some stage during a journey, pigs will need feed, water and rest in order to avoid being exposed to hazards leading to WCs such as prolonged hunger, prolonged thirst and resting problems.

Regarding provision of feed, the tendency for pigs to develop motion sickness, means that provision of feed to pigs on stationary vehicles during breaks in driving, likely will be difficult, as the journey probably cannot be continued until hours after the feeding of the animals. Research is needed to clarify costs and benefits of such practice.

Theoretically, unloading pigs into a pen allows the animals to have access to resting, watering and feeding areas in order to mitigate the WCs of transport. However, stopping a journey to unload the animals to provide a period of rest, feed, and water involves a number of hazards relevant for animal welfare such as the risk of stress, injury and infectious diseases. Some of these can be mitigated by the duration of the stay, some by offering high quality conditions to the animals, whereas others, such as the novelty of the surroundings, the extra unloading and reloading, and the mixing of unfamiliar animals as well as the inherent biosecurity risk, cannot be avoided. Thus, whether there are always beneficial effects for the welfare of pigs for a long journey to be interrupted to offer feed and water

while unloaded is equivocal, as currently, no data documenting the benefits of a stay in a CP (including physical layout and/or the animal management) in terms of pig welfare are available, and no data are available to assess the appropriate duration of a stay in a CP for pigs.

3.7. Specific scenarios: Transport of cull sows to slaughter

A) Current practice

Even though the main product of the pig industry is meat from finisher pigs (\sim 100–130 kg depending on MS), also sows (and boars) are slaughtered after their productive lives. The term 'cull sow' describes sows, for whom stockpersons/herd owners have decided to end the productive phase. The majority of cull sows are sent to slaughter, but some may be killed on-farm. This section focuses on sows sent to slaughter, a group of animals that constitute a distinct animal category within pig production. The timing of a decision to cull a sow depends on several factors, and cull sows are, thus, a rather diverse group constituting sows as young as parity 1 and above parity 10. Among the available studies of cull sows, a median parity of \sim 5 seems typical (De Jong et al., 2014; Zhao et al., 2015; Fogsgaard et al., 2018).

Across European countries, it is not unusual to cull approximately half of the sows in a herd per year. This means that millions of European sows are culled and sent to slaughter each year. The transport of sows to slaughter takes place entirely by road, either within or between MS. Among the MS, a few (Sweden and Denmark) have national regulation setting a limit of 8 h of journey duration for sows.

Farmers typically send a relatively low proportion of their sow population to slaughter at the same time, which means that vehicles typically will visit several herds to make a full load on their way to a slaughterhouse, or collect animals from an assembly centre. In addition, only relatively few slaughterhouses accept sows. Taken together, this means that sows often will be transported either quite far, or for quite some time, to reach the slaughterhouse. The need for several stops on the way, to load sows from other herds, can be a challenge for the biosecurity of the visited herds. One way to limit the risk of spread of diseases from vehicles transporting sows visiting several herds, can be the use of so called 'pick-up vehicles' (or 'transfer vehicles', see Figure 1 of this Scientific Opinion). Instead of allowing vehicles already holding other sows on board to back-up close to their buildings, some farmers load the sows to be culled on a particular day into some vehicle and transports this vehicle to the nearest public road. When the truck arrives to pick up the sows, it does not come near the farm buildings, but the sows are loaded from the 'pick-up vehicle' and onto the truck. Depending on the duration of the stay in the pick-up vehicle, the interval between leaving the home pen and arriving at the slaughterhouse may be considerably longer than the journey itself, and these vehicles can be said to constitute one extra link in the pre-slaughter logistic chain (Miranda-De La Lama et al., 2014; Herskin et al., 2017).

Another category of cull pigs is boars. Boars are used in pig production facilities as 'teaser' boars for facilitating detection and stimulation of sows in heat, and in large numbers in genetic boar studs for the production of semen for artificial insemination (Hook et al., 2010) where the turnover of boars is high and annual replacement rates range from 50% to 145% (Robinson and Buhr, 2005). Boars therefore form a smaller, yet consistent sub-population of adult pigs being transported. Boars being transported from breeding nucleus herds to commercial farms for use as teaser boars are likely to be young and healthy, whereas cull boars at the end of their productive life may show signs of impaired health. Health impairment accounted for 13–60% of annual culling from seven boar studs in Europe and North America (Robinson and Buhr, 2005), and feet and leg problems were reported to be the 3rd most common reason for culling boars identified in a survey of North American boar studs (Knox et al., 2008). Additionally, due to the typically isolated location of genetic nucleus herds, and the general trend towards consolidation of facilities that will slaughter and process boars, boars may undergo longer transport durations than other pig categories. Despite the potential welfare concerns during boar transport, no studies have been found to inform conclusions or recommendations.

B) Concerns specific to cull sows being transported by road

From an animal welfare point of view, transport of cull sows involves a number of concerns, the characteristics of which have been brought up by scientific publications (Nielsen et al., 2011; Grandin, 2016a,b), for example based on increased mortality upon arrival at the slaughterhouse compared with finishers (Lykke et al., 2007; Malena et al., 2007; Peterson et al., 2017). Recently, scientific attention has been directed towards this category of pigs, but knowledge about cull sows is still massively



lacking behind the other pig categories such as finishers or lactating sows on farm. Among the primary concerns for the welfare of cull sows during transport are (i) fitness for transport (Grandin, 2016a,b); (ii) the high sensitivity towards heat stress that characterises modern pig breeds (Brown-Brandl et al., 2014; Cabezon et al., 2017); and (iii) the tendency to fight when unfamiliar sows are mixed. Below, these three concerns are described further.

i) Fitness for transport

Herskin et al. (2017) analised the behaviour of cull sows while waiting for transport, reviewing as well the main reasons for culling them: 'As discussed by De Jong et al. (2014), reasons for culling of sows are influenced by factors such as sow genotype, housing conditions and management policies. Studies focusing on reports from farmers, such as Zhao et al. (2015), found that almost 80% of sows sent to slaughter were not culled as part of a strategy, and that the vast majority of sows were culled due to reproductive problems or reduced health. Similar conclusions have been drawn from studies involving clinical examination of sows upon arrival at slaughterhouses in Europe (De Jong et al., 2014) and the US (De Hollander et al., 2015). By studying the interval from insemination to culling, De Hollander et al. (2015) showed that a large proportion of sows were culled after weaning, and similar findings were reported by Engblom et al. (2007)'.

The concerns for the fitness for transport of cull sows are thus based on the following: (1) health impairment and/or reproductive problems (including lack of pregnancy) are among the main reasons to cull sows; (2) sows slaughtered at the end of their reproductive life are typically not young at the time of slaughter (although some young sows failing to get pregnant will be included); and (3) studies showing increased mortality upon arrival at the slaughterhouse in sows (Malena et al., 2007; Lykke et al., 2007a,b; Peterson et al., 2017).

Only a few studies have characterised cull sows clinically before departure from herds. In one of these, Fogsgaard et al. (2018) presented data from 47 loads of sows, originating from 12 Danish herds, and sent to slaughter over a 1-year period. From this study, involving only sows that the farmers had already selected for slaughter, more than 50% of the sows had at least one wound (defined as at least 1 cm) on the body, almost 1/3 had superficial skin lesions, 11% were observed with decubital shoulder ulcers, 25% had at least one udder lesion, 12% were observed with vulva discharge and 10% showed signs of abnormal gait. In addition, 39% of the sows were lactating at the time of the clinical examination (the morning before departure).

Even though no data exist to document the proportion of lactating animals among cull sows, recent studies suggest that it is not unusual to send sows to slaughter on the day of (or the day after) weaning. In comparison with dairy cows where milk leakage and reduced lying time have been reported after abrupt cessation of milking (e.g. Leitner et al., 2007; Zobel et al., 2013), limited scientific attention has been focused on the discomfort of sows due to abrupt cessation of lactation at weaning. Modern lactating sows are, due to the large litter sizes and high genetic potential for milk production, sensitive towards heat stress, especially close to weaning, where milk production is peaking (Williams et al., 2013). Hence, sending sows to slaughter may involve thermal challenges, especially for lactating sows, and hyperthermia has been suggested to be among the main reasons for cull sow mortality upon arrival to slaughterhouses (Peterson et al., 2017).

As reviewed by Fogsgaard et al. (2018) 'decubital shoulder ulcers have been associated with behavioral changes (Larsen et al., 2015) and are suggested to be painful (Herskin et al., 2011; Dahl-Pedersen et al., 2013). The consequences of pretransportation conditions/activities and actual transportation for sows with decubital shoulder ulcers are not studied. Mixing, fighting, and standing in the moving truck are likely to result in worsening of the ulcers, as the ulcers might be torn open or to be perceived more painful due to mechanical pressure/'bumping into inventory'. When cull sows were examined before and after transport, shoulder ulcer characteristics were reported to change. Thodberg et al. (2019) observed increased post-transport occurrence of reddening in and around the ulcers. This means that the presence of ulcers and perhaps even scars may have adverse consequences for sows during transport.

The concern for the fitness for transport of cull sows was deepened by the data from Thodberg et al. (2019), involving clinical characterisation of the same sows as involved in the study by Fogsgaard et al. (2018), upon arrival to the slaughterhouse after a road journey in approved vehicles for up to 8 h (46–469 min; mean: 232 min) under conditions conforming to Council Regulation (EC) No $1/2005^{1}$) regarding space, ventilation, etc. When comparing the clinical condition of the sows before and after the journey, a deterioration was found for a considerable proportion of the clinical measures, such as

an increased number of superficial skin lesions, increased number of wounds, increased occurrence of swollen and red areas, increased number of torn off hoofs and increased number of vulva lesions.

If the findings of the two Danish studies are representative for European conditions, they raise concern regarding the vulnerability of this particular category of pigs when exposed to stressors common during transport such as mixing or lack of adequate possibility to thermoregulate. Thus, depending on how animal fitness for transport is defined, (e.g. if fitness means that animals should arrive at their destination in a similar condition to that assessed before loading with minimal deterioration during their journey), it can thus be discussed whether cull sows are fit for the presently used transport conditions at all. There has, however, not been extensive research on how sows with different health issues respond to transport.

Recommendations from the World Organisation for Animal Health (WOAH) (2011) underline the possible vulnerability of cull sows and state that, for example, old, lactating and aggressive animals require special conditions on the day of slaughter in order to ensure their welfare.

ii) Group stress

Under natural or semi-natural conditions, adult female pigs live in relatively small groups consisting of genetically related individuals. Within a group, the animals have a well-established hierarchy, and overt fighting between group members is not often seen. In international pig production, pregnant sows are increasingly kept in groups, but often managed in a way where mixing of unfamiliar individuals happens several times during each production cycle. This management practice has received considerable scientific attention, and it is well-documented that sows – when mixed with unfamiliar individuals – will fight to re-establish a hierarchy, and that the fighting can be severe and of considerable duration (Mount and Seabrook, 1993; Arey and Edwards, 1998). Consequently, the mixing of unfamiliar sows has long been recognised as a welfare problem, mainly due to inter-sow aggression (Greenwood et al., 2014).

Herskin et al. (2020) and Thodberg et al. (2020) conducted surveys among sow herd owners and livestock drivers, respectively. One central outcome of these was the clear answer 'yes', when asked about whether sows are mixed before and/or during transport to slaughter. Consequently, it is highly likely that sows are also fighting when mixed in this phase of their life. No studies have documented sow behaviour onboard transport vehicles, but Thodberg et al. (2019) found increased occurrence of superficial skin lesions in sows after arrival to the slaughterhouse, possibly the outcome of intrajourney fighting. Herskin et al. (2020) studied the behaviour of cull sows during 6 h after mixing in onfarm pick-up pens before transport and observed a median occurrence of aggressive interactions per pen holding 3-4 sows of 33, ranging from 11 to 55. In addition, based on a small data set on the behaviour of sows when kept in pick-up vehicles before being picked up by larger trucks, Herskin et al. (2017) recorded incidents of aggression even when the stay was shorter than 10 min. Hence, even though documentation as such is missing, it is highly likely that sows will fight during transport, when mixed with unfamiliar individuals, and this behaviour constitutes a hazard for the continuous fitness for transport, the risk of heat stress, of group stress as well as the overall welfare of the sows.

iii) Heat stress

As mentioned in Section 3.5.3.1F, the upper threshold of the TCZ of lactating sows may be lower than 20°C, and the UCT is suggested to be ~ 22 °C for these animals. Lactating sows are especially sensitive to heat stress in the days before weaning of the piglets, when milk production is peaking (Williams et al., 2013). Examination of heat production of sows has shown a steady increase from farrowing until weaning (Brown-Brandl et al., 2014). However, as reviewed by Bracke et al. (2020), almost all available data on heat stress in sows comes from on-farm studies, and deals with lactating sows kept in typical housing systems such as farrowing crates.

Temperature data collected from inside the compartments of approved trucks (unknown number of trucks), transporting 39 loads of cull sows to slaughter in Denmark in spring, summer, autumn and winter in journeys lasting up to 8 h, identified that, when sows were kept at a maximum of the recommended Danish stocking density of 0.80 m²/250 kg sow, and with passive ventilation and the possibility for the driver to activate mechanical ventilation, truck temperatures (measured by sensors placed near the ceiling of the rear compartment on the lower deck) did not stay within the TCZ of sows, and even rose above the UCT in summer during non-stationary periods, and in summer and autumn during waiting at unloading (Thodberg et al., 2022). Importantly, during the journeys, the drivers of the vehicles were not notified about these data, and thus, may not have adjusted the use of mechanical ventilation accordingly. This suggests that improved ventilation methods and/or warning

18314732, 2022, 9, Downoaded from https://efsa.onlinelibrary.wiley.com/doi/10.2903/j.sa.2022.7445 by Universidad De Zaragoza, Wiley Online Library on [3001/2025]. See the Terms and Conditions (https://onlinelibrary.wiley.com/terms-and-conditions) on Wiley Online Library for rules of use; OA articles are governed by the applicable Creative Commons License



systems based on temperatures measured close to the bodies of the pigs, are required, maybe in combination with changes in stocking density (at least in the warmest compartments in the front top of the truck), as these journeys took place during a regular summer with not many extremes of temperature.

As mentioned, in order to obtain a full load, sow trucks may stop at several farms to collect sows and most trucks experience some waiting time at the slaughterhouse for unloading or stationary periods due to traffic or driver breaks (according to Thodberg et al. (2022) reported to range from 5 to 159 min), situations in which temperatures rise quickly to levels where sows would experience heat stress, with the internal truck temperature increasing by 7°C during a 10 min stationary period (Thodberg et al., 2022).

The frequency of sow aggressive interactions has been found to correlate with temperature inside the pick-up vehicle (Herskin et al., 2017), which may be a product of sows becoming more restless and also possibly trying to move away from one another in warmer temperatures, but being hindered by limited space.

3.8. Specific scenarios: 'Special health status pigs'

A) Current practice

In some cases, pigs are transported from a herd (or region/country) with a higher health status than the general animal population, through an area with a lower health status, to a new herd (or region/country) with a higher health status. In such cases, the herd (or region/country) receiving the pigs does not want them exposed to the general pig population during the course of any given journey. Therefore, journeys taking place without unloading the pigs during the journey may be advantageous. The focus here is on the transport of pigs on journeys by road without unloading them before their final destination.

Due to biosecurity concerns, pigs (typically weaners or breeding animals) are not unloaded to provide required rest, feed and water, when the maximum journey time is reached.

B) Concerns specific to pigs when they would not be unloaded from the vehicle

The WCs selected as highly relevant for pigs during the transit stage (group stress, heat stress, injuries, motion stress and sensory overstimulation, prolonged hunger, prolonged thirst, resting problems and restriction of movement) have already been dealt with in Section 3.5.2. Recommendations have been made in Section 3.5.2 that attempt to prevent the hazards and mitigate the WCs. However, pigs not unloaded before their final destination presents a larger risk with respect to their welfare. In such cases, facilities must be available on the vehicle to provide the necessary resting, feeding and drinking, and a suitable microclimatic environment. No scientific studies have been found that can demonstrate effective feeding and watering of pigs during transport. Knowledge about the quality of resting on a vehicle (moving or stationary) is also lacking. Air and bedding quality are other important issues that have not been studied in such a context.

The maximum duration of journeys without unloading the pigs has not been studied. The impact of rest periods has not been studied either, nor has the impact of repeated driving and resting. This constitutes gaps in knowledge.

3.9. Specific scenario: The transport of pigs by air

A) Current practice

The transport of livestock by air happens at low level when compared to road transport but it can still be part of industry practice, involving especially breeding animals. According to TRACES, in 2019, 2020 and 2021, 8,582, 29,661 and 65,664 pigs were transported by air, between MS or exported to third countries.

The transport of pigs by air starts with a journey by road from a farm or assembly centre to an airport where the pigs are unloaded from the road transport vehicles and loaded into transport crates. These crates are then loaded into an aircraft. Upon arrival at the airport of destination, the crates are unloaded from the plane and brought to an area where the pigs are removed from the crates and loaded into road transport vehicles to continue their journey.

www.efsa.europa.eu/efsajournal

B) Concerns specific to pigs transported by air

The WCs, hazards, preventive and corrective measures explained before in the transport by road (Sections 3.3, 3.4 and 3.5) also apply here. In addition, the transport by air presents certain concerns that are addressed below. No studies have been found focusing on the welfare of pigs during air transport. Hence, this assessment is based on expert opinion and general knowledge about air transport and animal transport.

Among the welfare concerns identified for air transport of pigs are high density confinement in crates, lengthy waiting times, extended periods of water and feed deprivation, variation in microclimatic conditions, and mixing of unfamiliar individuals. Other factors that may cause animal discomfort are motion stress and loud noises.

3.10. Specific scenario: The transport of pigs by train

A) Current practice

Rail is the means of transport used the least for pigs. According to TRACES, in 2020 and 2021, \sim 500 pigs were transported by rail between MS. The transport of pigs by rail starts with a journey by road from a farm or assembly centre to a train station where the pigs are unloaded from the road transport vehicles and loaded unto the train. Upon arrival at the train station of destination, the pigs are unloaded from the train and loaded into road transport vehicles to continue their journey.

B) Concerns specific to pigs transported by rail

No scientific literature pertaining to the transport of pigs by rail were found. However, based on expert opinion and general knowledge about rail transport and animal transport, the following welfare concerns were identified.

The WCs and hazards identified in Sections 3.3, 3.4 and 3.5 are all applicable to the transport of pigs by rail. The unloading of pigs from a vehicle at a railway station with the subsequent loading onto a railway carriage is a procedure requiring care. Similarly, the unloading of a railway carriage and the loading of a vehicle requires equal care. The animal handling facilities present in a railway station are of particular importance. This includes races/chutes and pens that will allow the pigs to be safely moved from the ramp of a vehicle to the loading ramp of the railway carriage. Among other welfare concerns identified for rail transport of pigs are high density confinement in railway carriages, lengthy waiting times, extended periods of water and feed deprivation, variation in microclimatic conditions, and mixing of unfamiliar conspecifics.

3.11. Uncertainty analysis

The uncertainty in the assessment performed for this Scientific Opinion was investigated in a qualitative manner following the procedure detailed in the EFSA guidance on uncertainty analysis in scientific assessments (EFSA Scientific Committee, 2018). The outcome of this Scientific Opinion is the identification and description of the highly relevant WCs, the related ABMs - measured in a qualitative or quantitative way - and hazards causing these WCs. Based on this identification and listing of WCs and ABMs, conclusions and recommendations are formulated allowing for different mitigation and preventing measures for the identified WCs (resource and management-based measures). As the identification and listing of the highly relevant WCs and ABMs was mainly based on expert opinion (integrating the severity, duration and prevalence of each WC) and not on a full comprehensive risk assessment, the uncertainty analysis was limited to the identification and description of the sources of uncertainty in the assessment carried out. A table describing the sources of uncertainty associated with the methodology used in the assessment is presented below (Table 27).

Table 27:Sources of uncertainty (in a non-prioritised order) associated with the assessment
methodology and inputs (extensive literature search, expert's opinions) for the
identification and assessment of the most relevant WCs and ABMs

Source of Uncertainty	Nature or cause of the uncertainty	Impact of the uncertainty on the assessment
Literature search – Language	The search was performed exclusively in English. More studies could have been identified by including references in languages other than English.	WCs might have been selected that in reality belonged to another category than highly relevant, and WCs that in reality were highly relevant might have missed to be selected.
Literature search – Publication type	The studies considered included primary research studies identified through the extensive literature search and grey literature (fact sheets, guidelines, conference papers, EU reports, book chapters, etc.) known to the EFSA Experts, but an extensive search of the grey literature was not conducted. Therefore, there may be reports and other guidance documents on animal welfare of which the EFSA Experts were not aware off.	Underestimation of the published relevant studies.
Literature search – Search strings	Although the search criteria were thoroughly discussed, some synonyms may have not been used in the search strings, and thus less hits might have been retrieved. In addition, literature from non- transport conditions, that may still have been relevant for the assessment, may also not have been found.	The understanding of the relation between hazards and ABMs may not be complete due to having missed data.
Literature search – data sources	The search was limited to Web of Science. Although the search was complemented by internet searches and manual searches of the publicly available literature, no data were retrieved from other sources (e.g. industry, NGO or authority data). More information could have been retrieved by applying a different methodology (e.g. public call for data).	The understanding of the relation between hazards and ABMs may not be complete due to having missed data.
Literature search – inclusion and exclusion criteria	The screening phase might have led to the exclusion of certain studies that could have included relevant information.	Underestimation of the published relevant papers.
Expert group – number of experts, type of experts	This SO was carried out by a working group of 12 EFSA Experts, of whom 3–5 were species-specific experts. The approaches underlying the SO is based on expertise from the whole working group, whereas the vast majority of the text within the SO has been written by the species-specific experts. Experts had to show they have no conflict of interest due to, e.g. involvement with the pig industry or NGOs. This may have resulted in reduced level of technical and applied expertise.	As the highly relevant welfare consequences were selected by expert opinion, the experts might have selected WCs that in reality belonged to another category than the highly relevant ones, and might have missed to select WCs that were in reality highly relevant.
Transport conditions of the studies retrieved in the extensive literature search	The transport conditions of the studies retrieved might have differed from the ones currently used in the EU, thus requiring an extrapolation exercise from the experts.	Under- or overestimation of the level of magnitude of the welfare consequences and related ABMs.



Source of Uncertainty	Nature or cause of the uncertainty	Impact of the uncertainty on the assessment
Husbandry practices and pig breeds and categories of the studies retrieved in the extensive literature search	The studies retrieved may have involved husbandry practices and pig breeds and categories differing from EU-standards. Thus, experts had to extrapolate findings to the EU relevant conditions in some cases.	Under- or overestimation of the level of magnitude of the welfare consequences and related ABMs.
Transport conditions of the studies retrieved in the extensive literature search	Transport conditions (e.g. driving style, ventilation capacity of the vehicle, external temperature) were not always specified in all the studies retrieved.	Under or overestimation of the effects of the transport conditions on the WCs selected.
Time allocation	The time and resources allocated to this SO were limited and additional time for reflection would have facilitated a more in- depth discussion of some of the aspects.	Inclusion, under- or overestimation of the level of magnitude of the WCs and related ABMs.
Lack of ABMs that are documented to be useful during transport in terms of feasibility, sensitivity or specificity	Based on the available knowledge, it was not possible to use single ABMs to assess the effect of exposure variables and transport conditions on welfare consequences.	Under- or overestimation of the level of magnitude of the WCs.
Transport being a complex stressor, for which animal welfare has been studied much less than animal housing	The complexity of animal transport with the many interacting hazards and thus WCs, means that many WCs are relevant, and thus that some can be missed in the selection of the highly relevant.	WCs that are in reality highly relevant are missed and thus underestimated.
Lack of available studies done under the recommended conditions	The number of studies available involving the conditions recommended in this SO is very limited. In addition, there are very few recent European studies. Thus, in some cases, and especially for the assessment of the journey time, experts had to extrapolate findings from studies done under different conditions.	Under- or overestimation of the level of magnitude of the welfare consequences.

4. Conclusions

The following section lists the conclusions of this Scientific Opinion

4.1. General conclusions on transport of pigs

- There are published protocols to assess animal welfare on farm and at slaughter, but no validated protocols are available to assess the welfare of pigs during transport.
- ABMs based on expert opinion for all highly relevant WCs along the transport stages are available (see Section 3.2.2). None of these, however, have been documented to be useful during transport in terms of feasibility, sensitivity or specificity.
- The use of ABMs in animal transport is hampered by the reduced access to animals particularly during the transit stage, but their use may be more feasible during other transport stages (e.g. loading/unloading).
- Technological development, such as intelligence-based camera systems or motion sensors, may increase the possibility to record and/or monitor ABMs during the entire transport process. However, such systems are not yet available in practice.
- Many hazards during the transport of pigs have been identified (see examples below for each transport stage). The consequences of the exposure to them often goes across journey stages (see Sections 3.3, 3.4, 3.5 and 3.6).



- Some hazards affecting the condition in which the animal begins the journey (e.g. level of hunger or thirst or health status) can only be mitigated or prevented before transport, although their associated WCs may appear later.
- Several sources of uncertainty were identified during the assessment, including (a) transport being a complex stressor, the consequences of which in terms of animal welfare have been studied much less than, e.g. for animal housing, especially under European conditions; (b) lack of ABMs that are documented to be useful during transport in terms of feasibility, sensitivity or specificity; (c) lack of available studies done under the recommended conditions; (d) lack of time; and (e) low number of experts involved. However, the impact of uncertainty was not quantified. A list of the major sources of uncertainty can be found in Table 27.

4.2. Conclusions on the preparation of pigs before transport

- The preparation phase is important for the protection of the welfare of pigs throughout the journey, lack of good preparation may lead to WCs.
- At present, no published protocols to assess animal welfare during preparation for transport are available.
- If transport of animals involves complex journeys including markets, assembly centres or other temporary stops, there will in principle be preparation at several levels before the initiation of the journey and before each re-loading of the animals.
- So far, the scientific focus addressing the preparation phase has been limited, especially concerning the transport of cull sows and boars as well as pigs transported for further fattening.
- Group stress, handling stress, heat stress, injuries and resting problems are the highly relevant WCs during the preparation of pigs for transport.
- Pigs, especially finishers, are often fasted before transport. This is done for several reasons not directly related to animal welfare (e.g. food safety and hygiene) that are not covered in the Scientific Opinion. From an animal welfare point of view, fasting leads to WCs, such as prolonged hunger and may indirectly lead to group stress, injuries and handling stress. Lack of fasting may also have consequences for the animals in terms of increased risk of motion sickness and hyperthermia. The underlying evidence to support fasting from an animal welfare point of view is not strong, and only involves finishers transported to slaughter. This constitutes a gap in knowledge.
- The longer the pre-transport fast, the sooner the WC prolonged hunger will appear during the following transport stages.

A) Fitness for transport

- The assessment of fitness for transport (Section 3.3.3) before departure is of utmost importance in the protection of pig welfare. However, currently no scientific definition of the concept of fitness for transport exists.
- If pigs are not properly assessed, and unfit animals are loaded, it is a hazard for their welfare, predisposing them to different WCs during later transport stages, and potentially leading to negative affective states such as discomfort, pain and suffering.
- Characteristics rendering animals unfit for transport are mainly related to health impairment, but not always, as for example certain age groups or certain physiological stages lead to animals being unfit for transport. Even though some guidance for assessing fitness for transport is available, however due to difference in implementation this leads to unreliability. Variation between assessments is documented. A list of conditions that make animals unfit to be transported has been provided in this Opinion. Some of these still require scientific validation and further study to identify the severity of the potential WCs that can arise when they are present.
- At present, thresholds for ABMs as indicators of animals being unfit for transport have seldom been established or validated. More knowledge about these is needed.
- Additional knowledge about pre-existing conditions that potentially lead to negative affective states during transport (e.g. lameness), is also needed. If these conditions are not mentioned in this Opinion, they will need to be added and ABMs need to be established to assess them.



- Successful assessment of fitness for transport requires well-educated staff (including professional groups such as veterinarians, farmers, herdsmen and livestock drivers), full clarity on responsibility, and a clear definition of the concept of fitness for transport.
- Advanced pregnancy is associated with increased risks of WCs during transport. There is European legislation as well as consensus across different available guidelines that pigs should not be transported in the last 10% of their pregnancy. However, scientific evidence to support this threshold is lacking, and the risk of reduced animal welfare may be present earlier.
- In cases where animals are considered fit for transport, but their fitness may be reduced, potential measures mitigating WCs are (a) shortened journey duration, (b) increased amount of bedding, (c) loading the affected animal last and unloading the animal first, and (d) providing sufficient space to lie down. However, the effectiveness of such mitigation measures to avoid the additional suffering has not been proven. This constitutes a gap in knowledge.

4.3. Conclusions for loading/unloading of pigs during road transport

- The highly relevant WCs during loading/unloading of pigs are: handling stress, heat stress, and sensory overstimulation.
- Across the highly relevant WCs, the major hazards are inappropriate handling, unsuitable facilities and high temperatures. A delay in loading and/or unloading is a hazard for a deterioration of these WCs.
- The main preventive measures are establishment and maintenance of proper facilities, avoiding loading during hot hours and education and training of handlers.

4.4. Conclusions for the transit stage during road transport of pigs

- The highly relevant WCs for pigs during the transit stage are group stress, heat stress, injuries, motion stress and sensory overstimulation, prolonged hunger, prolonged thirst, resting problems and restriction of movement.
- Among the major hazards for animal welfare during the transit stage are high effective temperatures, mixing of unfamiliar animals, insufficient space allowance, time off feed and water, reduced intake of water, vehicle movements and lack of possibility to rest.
- Preventive and corrective/mitigating measures have been suggested (see Section 3.5.2). Several of the highly relevant WCs of the transit stage (e.g. motion stress, resting problems, restriction of movement) cannot be fully prevented.
- So far, no scientific evidence quantifying water intake as adequately is available, even in journeys with vehicles fitted with drinkers.
- The severity of the WCs during the transit stage of transport will depend on the exact conditions pertaining to an individual journey (e.g. microclimatic conditions, space allowance and road conditions). The hazards relating to these conditions potentially interact. The exposure to these hazards will continue at least as long as the journey continues.

A) Microclimatic conditions

- Pigs are special among mammals in that they have difficulty in dissipating heat. Thus, pigs are more vulnerable to heat stress, especially at ambient temperatures with high humidity, than most other farm animals.
- There is growing evidence that genetic selection for productivity has reduced the pigs' ability to cope with heat stress. It has not been possible to find data to clarify whether pigs of certain breeds, or from certain regions of the EU are better adapted to sustain heat stress than others. This warrants further study.
- Larger pigs, especially lactating sows, are more vulnerable to heat stress than smaller pigs. Compared to the other categories of pigs typically subjected to transport, weaners are likely to be the least vulnerable to heat stress. The available information for boars is very limited.
- If temperature in the transport vehicle remains below the upper limit of the TCZ, pigs will most likely not experience stress or negative affective states associated with heat stress during transport.
- The WC heat stress may start when pigs are no longer in their thermal comfort zone, and the risk and severity of heat stress is high when the thermal conditions reach the UCT.



- Not only the temperature in the vehicle, but also other environmental conditions influence heat load placed on pigs during transport, such as humidity, thermal radiation, temperature of surrounding surfaces and wind speed. These will all influence the microclimatic conditions experienced by pigs and should, in theory, all be taken into account when microclimatic conditions of pigs during transport are evaluated. In addition to dry temperature, humidity is considered the most important of these to take into consideration.
- The available information on thermoregulation of the three relevant categories of pigs has allowed for estimates of thresholds for TCZ and for UCT (Table 28) although uncertainty exists regarding these values.

Table 28:	Estimates	of	upper	thermal	thresholds	for	the	thermal	comfort	zone	and	the	upper
	critical tem	npe	rature ((°C) for p	igs								

	Upper threshold of thermal comfort zone	Upper critical temperature
Weaners of ~ 30 kg	25°C	30°C
Finishing pigs	22°C	25°C
Sows	20°C	22°C

Although sensors only recording temperature have been used in transport of livestock so far, it
would be a significant refinement to use sensors taking account of other environmental
conditions influencing heat load placed on pigs during transport (preferably a combination of
temperature and humidity). For variations of dry temperature and relative humidity, the higher
the levels of relative humidity, the lower the upper thresholds of TCZ and UCT will be, when
measured as a dry temperature only.

B) Space requirements

- The allometric equation $[A = k \times W^{2/3}]$ (where A is Area in m² per animal, and W is liveweight in kg) can be used to calculate the physical space requirements of any category of pigs.
- For this Scientific Opinion, the minimum space allowance is set by the first limiting factor reducing the ability of pigs to undertake relevant biological functions during transport. One such function is the ability to adjust posture in response to acceleration and other events related to driving. However, no studies have been found to provide evidence regarding this biological function. This is a gap in knowledge.
- Regarding space for thermoregulation and provision of water during transport, there is insufficient evidence to provide a quantification of the minimal space required to fulfil these biological functions under transport conditions.
- Based on the available evidence for the remaining biological functions, the available evidence suggests that a k-value of at least 0.027 is required. Providing pigs this space during transport will allow all animals in a compartment to lie down in a semi-recumbent posture. Whether provision of this space during transport can allow pigs space to adjust posture in response to acceleration and other events related to driving, space for thermoregulation and for effective provision of water during transport is not known.
- Table 29 gives estimates of the minimum space allowance suggested for pigs during transport.
- Table 29:
 Minimum space allowance (based on allometric equation) suggested for pigs of different body weights

Approximate weight	Area (m²/animal)
30 kg	0.26
110 kg	0.62
160 kg	0.79
240 kg	1.04

• The vertical space in a means of transport is important for animal welfare. Low vertical space is associated with reduced ventilation, lack of ability to move around and lack of space for natural movements, and should be prevented in order to avoid WCs such as heat stress and restriction of movement. No studies have established a proper deck height for pigs during



transport. There are equations to estimate the height of pigs, but a lack of data to determine the appropriate deck height for different pig categories in order to protect their welfare during transport. Research is needed to establish evidence-based thresholds.

C) Journey times

The conclusions regarding journey time are based on a scenario where animals are transported under the microclimatic conditions and with the minimal space allowance recommended in Sections 3.5.3.1 and 3.5.3.2, respectively. The implications of these are that the risk and severity of the WCs heat stress and restriction of movement are much reduced and are thus given less weighting in the conclusions and recommendations on journey duration.

The remaining highly relevant WCs can be classified into those that are **continuous or semicontinuous** (i.e. begin with the onset of the journey and occur continuously or intermittently throughout its duration), those that are **progressive** (i.e. may not be present at the beginning of the journey but develop progressively as it continues), and those that are **sporadic** (i.e. problems in individual animals which may be exacerbation of a pre-existing condition or may occur spontaneously at any point in the journey and whose WCs will continue thereafter). These are summarised in Table 30.

Type of welfare consequence	Welfare consequence	Development over time	Expected development over time
Continuous or semi-continuous	Motion stress	Continuing throughout the transit stage	Severity will increase over time leading to fatigue
	Sensory overstimulation	Sensory overstimulation repeated intermittent	Can lead to fear and distress
	Group stress and resultant injuries (skin lesions)	Continuing throughout the transit stage, but development over time not known	Group stress may lead to fatigue, injuries will lead to pain and/or discomfort.
	Resting problems	Continuous throughout the transit stage	Severity will increase over time leading to fatigue
Progressively developing	Prolonged thirst	Available information shows that behavioural and physiological signs of thirst can be present after 8 h of transport without effectively accessing water.	Severity will increase with time, leading to dehydration and potentially suffering
	Prolonged hunger	Available information shows that behavioural and physiological signs of hunger can be present after 12 h of feed deprivation. Depending on the pre- transport fasting this may correspond with the early hours of the journey.	Severity will increase with time, leading to weakening and exhaustion
Sporadic	Pain and/or discomfort from health conditions	May start at any time if pre-existing health conditions are present or occur during transport	If present, severity will increase with time, leading to suffering

Table 30: Summary of conclusions on the development of welfare consequences over journey time

Below, the prevalence, the severity and the duration of each of the highly relevant WCs involved in this assessment are summarised:

- Motion stress and sensory overstimulation start as soon as a vehicle starts moving, and continues while the vehicle is moving, affecting all animals in the moving vehicle. Animals experience stress potentially leading to fatigue and negative affective states such as fear and distress;
- The prevalence of group stress and resultant injuries is high with a variation in severity as not all pigs will be involved in fighting to the same extent. The duration of group stress depends on journey duration and may lead to fatigue.
- The prevalence of prolonged hunger is expected to be very high, as pigs especially animals sent for slaughter – likely have been fasted before the initiation of the journey. The severity of hunger will develop further during the journey. Physiological indicators of hunger (e.g. liver

glycogen) as well as behavioural studies of post-journey eating behaviour, hunger is expected to be present after 12 h of feed deprivation, which (under current practice), for a large proportion of pigs transported to slaughter, corresponds to the initiation of the journey.

- The prevalence of prolonged thirst may be high if water is not provided to the animals or they, for some reason, such as lack of familiarity or fear of other animals, are not able to drink enough water. So far, very limited documentation for proper intake of water, even in journeys fitted with drinkers, is available. Behavioural and physiological signs of thirst can be present after 8 h of transport. The severity is expected to increase with increasing duration, as the need for water becomes problematic for the animals. Prolonged thirst may lead to frustration, discomfort and dehydration.
- The prevalence of resting problems is at least moderate, as resting problems may affect a large proportion of animals in a moving vehicle, even when space for lying is available. Duration depends on journey duration, and severity is expected to increase with increasing duration and may lead to fatigue.
- The pain and/or discomfort from health conditions or injuries the consequences of which might be severe and will worsen over time during transport and may lead to suffering;

To conclude, during transport pigs will be exposed to a number of hazards, either on their own or in combination, leading to WCs. The amount of time the animals are exposed to these hazards is dependent on the journey duration.

4.5. Conclusions on journey breaks and control posts

- By definition, journey breaks remove the animals from the hazards that they are exposed to during transit, and allow them to recover from the associated WCs.
- The science underlying the duration of a journey break necessary to protect the welfare of pigs is ambiguous and further research is needed.
- It has not been clearly shown that it is possible for pigs to drink and rest according to their needs in a journey break on a stationary truck. This constitutes a gap in knowledge.

If pigs are to recover from the WCs experienced during transit they need to be unloaded from the vehicle for a sufficiently long period of time.

The highly relevant WCs for pigs during the CP stage are group stress, handling stress, heat stress, injuries, prolonged hunger, prolonged thirst, restriction of movement and resting problems.

Among the major hazards for animal welfare during the CP stage are mixing of unfamiliar animals, high effective temperatures, competition for resources such as feed and water, and lack of space.

- When CPs are used, animals must be unloaded and reloaded. These procedures involve hazards potentially leading to WCs such as handling stress, heat stress and sensory overstimulation (Section 3.4).
- In addition, CPs involve biosecurity risks as animals can be exposed to infectious diseases through direct or indirect contact with other animals and opportunistic pathogens.
- Across the categories of pigs typically transported on journeys involving journey breaks, the scientific focus on CPs has been very limited. This means that whether CPs in their current state fulfil their intended function is not known.
- For a stay in a CP or similar to be beneficial for the welfare of pigs during transport, a journey break needs to be long enough for each animal to eat and then drink and rest. Due to the tendency of pigs to develop motion sickness during transport, the animals should not be reloaded immediately after eating. The time needed for the WCs from the transit stage to be mitigated by a stay at a CP constitutes a gap in knowledge.

4.6. Welfare of pigs when transported by air and rail

- For pigs, air transport involves only a small fraction of transported animals, and mainly breeding animals. Rail is the mode of transport used the least for pigs.
- The WCs, hazards, preventive and corrective/mitigating measures explained in the transport by road (Sections 3.3–3.6) also apply here. However, the EFSA experts identified additional concerns for pig welfare during air or rail transport, namely: high density confinement in crates, lengthy waiting times, extended periods of water and feed deprivation, variation in



microclimatic conditions, and potential exposure to noxious gases. Other factors that may lead to WCs are motion stress and loud noises.

The available evidence to evaluate the welfare of pigs when transported by air or rail is very scarce. This constitutes a gap in knowledge.

4.7. Conclusions on specific scenarios

A) Transport of cull sows to slaughter

- Cull sows are often collected from different farms before transport to the slaughterhouse. This may increase the journey time and transport-related risks to animal welfare.
- The WCs, hazards, preventive and corrective measures explained for transport of pigs by road (Sections 3.3–3.6) also apply here, but when exposed to the hazards, the severity of the resultant WCs will most likely be higher for cull sows.
- In addition to the above, three major welfare concerns for the transport of cull sows to slaughter are (1) the fitness for transport, (2) the high sensitivity towards heat stress, and (3) the risk of group stress due to mixing with unfamiliar sows at one or more of the transport stages.
- No studies are available to allow a quantitative assessment of journey time for cull sows.
- Due to the general health impairment of a large proportion of cull sows, they may experience a higher risk of worsening of pre-existing health conditions (e.g. lameness), as well as a higher risk of new health conditions occurring during transport, than average pigs.
- Therefore, cull sows benefit from a restriction on journey duration compared to other categories of pigs. The current knowledge does not allow estimates of such a journey duration restriction, which constitutes a gap in knowledge.

B) Transport of special health status pigs

- The WCs, hazards, preventive and corrective/mitigating measures explained in the transport by road (Sections 3.3, 3.4 and 3.5) also apply here, with some extra additions due to the lack of possibility to unload the animals.
- The available evidence to evaluate the welfare of pigs during journeys where animals cannot be unloaded is very scarce. This constitutes a gap in knowledge.

5. Recommendations

The following section lists the recommendations of this Scientific Opinion.

5.1. General recommendations on the transport of pigs by road

- Protocols for the assessment of welfare of pigs during transport should be developed and validated, preferably involving aspects of pre-and post-transport housing and production systems, e.g. if animals are transported to or from intensive or extensive husbandry conditions.
- In order to have useful ABMs in animal transport, research should be carried out to develop them. This should include the identification and validation of technological solutions to assess welfare, aiming to assess outcomes more than input.

5.2. Recommendations for preparation of pigs before transport

- Protocols to assess animal welfare during the preparation phase, including scenarios characterised by repeated re-loading of animals should be developed and validated.
- The lack of scientific focus on the preparation phase means that the possibility to give evidence-based recommendations is limited. Based on general knowledge about pigs and the current practice, the following preventive measures are recommended in order to protect the welfare of pigs during this phase:
 - Avoid mixing of unfamiliar conspecifics.
 - Pigs should have unlimited access to water during the whole preparation phase. The type and number of drinkers should be adjusted to the category and number of pigs (taking onfarm recommendations as a starting point).



- In the hours before loading onto a means of transport, pigs should be given the opportunity to rest, in terms of space, microclimatic conditions and possibility to avoid aggressive interactions with conspecifics.
- In the hours before loading onto a means of transport, pigs should be kept in conditions allowing them to avoid solar radiation and providing microclimatic conditions corresponding to the TCZ of the specific pig category.
- Periods of pre-transport fasting should be adjusted to the category of pigs and be appropriate for the planned journey duration, and in addition take into consideration whether pigs are transported for slaughter or for further fattening/breeding.
- Until evidence-based thresholds are established, it is recommended to expose finishers being transported to a pre-transit fasting of less than 10 h. The time taken to load should be included within this 10-h period. For other pig categories pre-transit fasting should likely be shorter.
- Research should be carried out to identify pros and cons of pre-journey fasting in terms of animal welfare. This should include examination of the effects of fasting duration and the establishment of guidelines, which could, e.g. take into account different types of feed in different relevant categories of pigs transported for slaughter as well as for fattening or breeding.

A) Fitness for transport

- In order to avoid WCs such as pain and discomfort, animals should be fit for transport. Guidelines based on ABMs for conditions leading to animals being unfit, including thresholds, should be established and validated. Among suggested candidate ABMs are lameness score, pyrexia, dyspnoea, ataxia, disorientation/abnormal behaviour, abnormal navel, wounds, aspect, demeanor, swollen joints, abscess, hernias, rectal and vaginal prolapse, late pregnancy, bone fractures, body condition score, eyesight deficiency.
- Risk associated with transport of animals with a number of conditions, potentially involving negative affective states during transport, such as pregnancy, should be examined.
- In order to avoid doubt and misclassification of animals in relation to fitness for transport, the concept should be properly defined, professional groups (including farmers, handlers, drivers, haulers, inspectors and veterinarians) should be well-educated, and questions on responsibility between the groups should be clarified.
- The effectiveness of measures to mitigate risk of transport of animals with reduced fitness should be examined.

5.3. Recommendations for loading/unloading of pigs during road transport

In order to minimise handling stress and other WCs during loading and unloading, handlers should be properly educated and trained.

Delays in loading and unloading should be avoided because of the increased exposure to hazards such as high effective temperatures, and the potential resultant WCs such as heat stress.

5.4. Recommendations for the transit of pigs during road transport

A) Microclimatic conditions

- Among the environmental factors affecting the heat load placed on pigs during transport, it is recommended to take temperature and at least humidity into account when pig welfare during transport is evaluated. The relationship between these can be expressed by different indexes. Further research should be carried out to assess costs and benefits of different choices of index.
- In order to reduce the risk of WCs due to exposure to high effective temperatures, the temperature inside vehicles transporting pigs should not exceed the UCT estimated to be 30°C for weaners, 25°C for finishers and 22°C for sows. However, it is recommended to keep temperatures below the upper threshold of the TCZ, estimated to be 25°C for weaners, 22°C for finishers and 20°C for sows, in order to protect the welfare of the animals during transport.
- Means of transport should be equipped with sensors recording microclimatic conditions as close as possible to the position of the animals in the vehicle, and at several locations to



include hot as well as cold spots, and representative points in between, thereby allowing monitoring of the microclimate (preferably a combination of temperature and humidity) of the load, and the adjusting of the ventilation if the conditions exceed the comfort zone levels. Technical issues (e.g. accuracy, maintenance, placement, reliability and calibration) relating to this improved approach should be clarified.

• Future research is recommended in the development of systems to maintain the microclimatic conditions in stationary as well as moving vehicles across different compartments and deck heights, when using forced and natural ventilation. Research should include possibilities to maintain low temperatures by, e.g. air conditioning.

B) Space allowance

- In order to protect the welfare of pigs during transport, it is recommended to use the allometric equation $A = kW^{2/3}$ to calculate the minimum space allowance.
- Until evidence-based thresholds are established for the space requirements of pigs to be able to adjust to acceleration events, it is recommended that the minimum space allowance is based on a k-value of at least 0.027, thereby allowing pigs to be able to lie down during journeys. This recommendation applies to all categories of pigs transported in groups, and takes as a prerequisite that pigs are transported under microclimatic conditions corresponding to what is recommended in Section 3.5.3.1 of this Scientific Opinion.
- Research is needed to establish evidence-based thresholds for vertical space, in order to prevent hazards such as reduced ventilation, lack of ability to move around and lack of space for natural movements, and thus avoid WCs such as heat stress and restriction of movement.
- Future research is recommended in the following areas:
 - Determine the effect of different space allowances, as based on the biological functions of pigs during transport on the behaviour and clinical condition of different pig categories during transport. In particular, research should explore the ability of pigs to lie, to stabilise themselves, to thermoregulate and perform social behaviours (including fighting) during transport of long and short duration.
 - Determine the required space and design of the vehicle, including the location of drinkers, the drinker type and the pig to drinker ratio for pigs to drink during transport. Focus should be on water intake by the animals and on potential consequences of water wastage in terms of animal welfare.
 - Determine vehicle design components, including the deck height and ventilation capacities to protect the welfare of different categories of pigs

C) Journey time

- The number and the severity of hazards that animals are exposed to during transport influence the resultant WCs, but on the basis of evidence on continuous WCs involving stress and negative affective states, for the benefit of animal welfare, the journey duration should be kept to a minimum.
- To limit the impact of transport on animal welfare, in an effort to reduce the exposure to hazards and related WCs, continuous, progressively developing and sporadic, it is recommended to consider that:
 - Animals experience motion stress and sensory overstimulation throughout the entirety of the journey potentially leading to fatigue, fear and distress;
 - Group stress starts as soon as pigs are mixed and continues during the journey and may lead to fatigue;
 - The pain and/or discomfort due to health conditions and injuries might be severe and will worsen over time during transport;
 - Resting problems severity is expected to increase with increasing duration and may lead to fatigue;
 - Behavioural and physiological changes that are likely to be associated with thirst have been identified after 8 h of transport, even when drinkers are fitted to the transport vehicle;



- Behavioural and physiological changes indicative of hunger can be present after 12 h of feed deprivation. Depending on the pre-transport fasting this may correspond with the early hours of a journey.
- Future research is recommended in the following areas:
 - Investigation of relationships between journey time, journey conditions and ABMs considered to reflect affective states of pigs for all animal categories, including knowledge about the progressively developing WCs and their changes over time. Such research could inform the appropriate limits on journey duration if the recommended conditions of temperature and space are not fulfilled.
 - Development of transport practices by which the risk of motion sickness can be reduced, and pre-journey fasting of pigs therefore can be avoided.

5.5. Recommendations for journey breaks and control posts

Based on general knowledge about pigs and the current practice, the following is recommended to protect the welfare of pigs during this transport stage. New scientific knowledge may lead to adjustments of these.

- Avoid mixing unfamiliar pigs in a CP.
- The microclimatic conditions in the CP should allow pigs to be kept in their TCZ throughout their stay.
- In order to cover the basic needs of pigs, conditions regarding space and access to enrichment laid down in the legislation regarding pig husbandry conditions are recommended as the starting point for CPs in and outside EU. In addition, future regulation and recommendations on these matters, regarding pig housing on farm, should also apply for CPs.
- When feed is provided, all pigs should be able to eat at the same time.
- Access to drinkers should be easy at all times.
- Animals showing signs of weakness or disease should be inspected and treated accordingly. Animals unfit for further transport should not follow the consignment at re-loading, but be slaughtered, treated or euthanised according to the prognosis of their condition. Contingency plans should be in place for injured and sick animals.
- Pigs should not be overfed before re-departure.
- Pigs should not enter buildings that have not been cleaned properly, and are not dry.
- Until evidence is available, it is recommended that the number of times that pigs stay in a CP should be as low as possible.
- Until evidence-based thresholds are established for the duration of rest periods, alignment with the current 24 h period is recommended.
- Future research in the following areas is recommended:
 - Research is needed to examine whether CPs fulfil their function, and how they should be designed and managed in order to protect pig welfare.
 - Research is needed to determine the effects on pig welfare of the duration of a stay at a CP as well as determine WCs of repeated stays at CPs.

5.6. Recommendations for the transport of pigs by air and rail

• Until evidence-based thresholds are established for these means of transport, alignment with the recommendations for road transport is recommended.

5.7. Recommendations on specific scenarios

A) Transport of cull sows to slaughter

Due to the similarity in the WCs and hazards, alignment with the recommendations for road transport for preparation, loading/unloading and transit microclimatic conditions and space allowance (see Sections 3.3, 3.4, 3.5.3.1, 3.5.3.2 and 3.6) is recommended.

In order to avoid WCs, such as pain and discomfort, animals should be fit for transport. Guidelines based on ABMs for conditions leading to animals being unfit, including thresholds, should be established and validated.



In order to avoid doubt and misclassification of animals in relation to fitness for transport, the concept should be properly defined, professional groups (including farmers, stockpersons, drivers, haulers, inspectors and veterinarians) should be well-educated, and questions on responsibility between the groups should be clarified.

If these animals are fit for transport, the journey to a slaughterhouse should be kept to a minimum, be direct and not involve any unloading and reloading at any interim premises. Sows should be slaughtered as soon as possible upon arrival.

If these animals are not fit for transport and are without the prospect of recovery in a reasonable period of time, they should be killed on farm as soon as is possible.

B) Special health status

The recommendations for road transport (see Sections 3.3, 3.4 and 3.5) are applicable in this context therefore it is recommended to apply them, except for journey breaks, as these do not apply to this scenario. Research is needed to develop vehicles and procedures, including stationary resting periods, to ensure that the welfare of pigs during this type of transport is protected.

References

- Acevedo-Giraldo JD, Sánchez JA and Romero MH, 2020. Effects of feed withdrawal times prior to slaughter on some animal welfare indicators and meat quality traits in commercial pigs. Meat Science, 167, 107993.
- Alsop JE, Hurnik D and Bildfell RJ, 1994. Porcine ketosis: a case report and literature summary. Journal of Swine Health and Production., 2, 5–8.
- Andersen EMO, Spangsberg R, Pedersen KS, Barington K and Jensen HE, 2014, Umbilical hernia and differential diagnoses in slaughter pigs. In Proceedings of the 23rd International Pig Veterinary Society (IPVS) Congress. pp. 8–11.
- Aradom S, Gebresenbet G, Bulitta FS, Bobobee EY and Adam M, 2012. Effect of transport times on welfare of pigs. Journal of Agricultural Science and Technology, 2, 544.
- Arey DS and Edwards SA, 1998. Factors influencing aggression between sows after mixing and the consequences for welfare and production. Livestock Production Sciences, 56, 61–70.
- Arndt H, Volkmann N, Spindler B, Hartung J and Kemper N, 2019. Do pigs have adequate space in animal transportation vehicles?-Planimetric measurement of the floor area covered by finishing pigs in various body positions. Frontiers in Veterinary Science, 5. https://doi.org/10.3389/fvets.2018.00330
- Arndt H, Spindler B, Hoemier S, Hartung J and Kemper N, 2020. Planimetric determination of the static space of cull sows as the first step towards a recommendation of loading densities for cull sows during road transportation in the European Union. Agriculture, 11, 20. https://doi.org/10.3390/agriculture11010020
- Ashkenazy S and Ganz FD, 2019. The differentiation between pain and discomfort: a concept analysis of discomfort. Pain Management Nursing, 20, 556–562.
- Averós X, Knowles TG, Brown SN, Warris PD and Gosálvez LF, 2008. Factors affecting the mortality of pigs being transported to slaughter. Veterinary Record, 163, 386–390.
- Averós X, Knowles TG, Brown SN, Warriss PD and Gosálvez LF, 2010. Factors affecting the mortality of weaned piglets during commercial transport between farms. Veterinary Record, 167, 815–819.
- Barton Gade P and Christensen L, 1998. Effect of different stocking densities during transport on welfare and meat quality in Danish slaughter pigs. Meat Science, 48, 237–247.
- Bates R, 2016. Knowledgebase: transporting pregnant sows #294794. Ask extension forum. Available online: transporting pregnant sows #294794 Ask Extension Date accessed: 23 May 2022.
- Baumgard LH and Rhoads RP, 2013. Effects of heat stress on postabsorptive metabolism and energetics. Annnual Review Animal Biosciences, 1, 311–337.
- Becerril-Herrera M, Alonso-Spilsbury M, Ortega MT, Guerrero-Legarreta I, Ramirez-Necoechea R, Roldan-Santiago P and Mota-Rojas D, 2010. Changes in blood constituents of swine transported for 8 or 16 h to an Abattoir. Meat Science, 86, 945–948.
- Beltran-Alcrudo D, Falco JR, Raizman E and Dietze K, 2019. Transboundary spread of pig diseases: the role of international trade and travel. BMC Veterinary Research, 15, 1–14.
- Bertol TM, Ellis M, Ritter MJ and McKeith FK, 2005. Effect of feed withdrawal and handling intensity on longissimus muscle glycolytic potential and blood measurements in slaughter weight pigs. Journal of Animal Science, 83, 1536–1542.
- Bertol TM, Braña DV, Ellis M, Ritter MJ, Peterson BA, Mendoza OF and McKeith FK, 2011. Effect of feed withdrawal and dietary energy source on muscle glycolytic potential and blood acid-base responses to handling in slaughter-weight pigs. Journal of Animal Science, 89, 1561–1573.
- Bjerg B, Brandt P, Pedersen P and Zhang G, 2020. Sows' responses to increased heat load–A review. Journal of Thermal Biology, 94, 102758.



- Blair B and Lowe J, 2019. Describing the cull sow market network in the US: a pilot project. Preventive Veterinary Medicine, 162, 107–109.
- Bohmanova J, Misztal I and Cole JB, 2007. Temperature-humidity indices as indicators of milk production losses due to heat stress. Journal of Dairy Science, 90, 1947–1956.

Boissy A, 1995. Fear and fearfulness in animals. The Quarterly Review of Biology, 70, 165–191.

- von Borell E, Langbein J, Després G, Hansen S, Leterrier C, Marchant-Forde J, Marchant-Forde R, Minero M, Mohr E, Prunier A, Valance D and Veissier I, 2007. Heart rate variability as a measure of autonomic regulation of cardiac activity for assessing stress and welfare in farm animals - A review. Physiology and Behaviour, 92, 293– 316.
- Bortolozzo FP, Gaggini TS and Wentz I, 2011. Infertilidade sazonal no suíno: caracterização e consequências durante a fase gestacional. Simposio international de suinocultura. Produção, Reprodução e Sanidade Suína, 6, 117–131.
- Braastad BO, 1998. Effects of prenatal stress on behaviour of offspring of laboratory and farmed mammals. Applied Animal Behaviour Science, 61, 159–180.
- Bracke MBM, Herskin MS Marahren M, Gerritzen MA and Spoolder HAM, 2020. Review of climate control and space allowance during transport of pigs. European Reference Centre for Animal Welfare: Pigs (EURCAW-Pigs), version 1.0. Available online: https://library.wur.nl/WebQuery/wurpubs/fulltext/515292 [Accessed: 15 January 2022].
- Bradshaw RH, Parrott RF, Forsling ML, Goode JA, Lloyd DM, Rodway RG and Broom DM, 1996a. Stress and travel sickness in pigs: effects of road transport on plasma concentrations of cortisol, beta-endorphin and lysine vasopressin. Animal Science, 63, 507–516.
- Bradshaw RH, Parrott RF, Goode JA, Lloyd DM, Rodway RG and Broom DM, 1996b. Behavioural and hormonal responses of pigs during transport: effect of mixing and duration of journey. Animal Science, 62, 547–554.
- Brandt P and Aaslyng MD, 2015. Welfare measurements of finishing pigs on the day of slaughter: a review. Meat Science, 103, 13–23.
- Brandt P, Bjerg B, Pedersen P, Sørensen KB, Rong L, Huang T and Zhang G, 2022. The effect of air temperature, velocity and humidity on respiration rate and rectal temperature as an expression of heat stress in gestating sows. Journal of Thermal Biology, 104, 103142.
- Broom DM, 2006. Behaviour and welfare in relation to pathology. Applied Animal Behaviour Science, 97, 73–83.
- Brown SN, Knowles TG, Edwards JE and Warriss PD, 1999. Behavioural and physiological responses of pigs being transported up to 24 hours followed by six hours recovery in lairage. Veterinary Record, 145, 421–426.
- Brown SN, Knowles TG, Wilkins LJ, Chadd SA and Warris PD, 2005. The response of pigs to being loaded or unloaded onto commercial animal transporters using three systems. The Veterinary Journal, 170, 91–100.
- Brown-Brandl TM, Eigenberg RA, Nienaber JA and Kachman SD, 2001. Thermoregulatory profile of a newer genetic line of pigs. Livestock Production Science, 71, 253–260.
- Brown-Brandl TM, Nienaber JA, Xin H and Gates RS, 2004. A literature review of swine heat production. Transactions of the ASAE, 47, 259–270.
- Brown-Brandl TM, Hayes MD, Xin H, Nienaber JA, Li H, Eigenberg RA, Stinn JP and Shepherd T, 2014. Heat and moisture production of modern swine. ASHRAE Transactions, 120, 469.
- Bulitta FS, 2015. Effects of handling on animals welfare during transport and marketing (Vol. 2015, No. 2015: 117).
- Cabezon FA, Schinckel AP, Marchant-Forde JN, Johnson JS and Stwalley RM, 2017. Effect of floor cooling on late lactation sows under acute heat stress. Livestock Science, 206, 113–120.
- Canaday DC, Salak-Johnson JL, Visconti AM, Wang X, Bhalerao K and Knox RV, 2013. Effect of variability in lighting and temperature environments for mature gilts housed in gestation crates on measures of reproduction and animal well-being. Journal of Animal Science, 91, 1225–1236.
- de Castro Júnior SL and Silva IJOD, 2021. The specific enthalpy of air as an indicator of heat stress in livestock animals. International Journal of Biometeorology, 65, 149–161.
- Chevillon P, Frotin P, Rousseau P, Kerisit R, Buffet S, Vautier A and Chauvel J, 2003. Incidence de la densité de chargement et du système d'alimentation sur le bien-être des porcs lors d'un transport de porcs de 36 heures incluant 9 heures d'arrêt. Journées de la recherche porcine en France, 35, 179–186.
- Clemens ET, Schultz BD, Brumm MC, Jesse GW and Mayaes HF, 1986. Influence of market stress and protein level on feeder pig hematologic and blood chemical values. American Journal of Veterinary Research, 47, 359–362.
- Cockram MS, 2007. Criteria and potential reasons for maximum journey times for farm animals destined for slaughter. Applied Animal Behaviour Science, 106, (4 SPEC. ISS.), 234–243.
- Cockram MS, 2019. Fitness of animals for transport to slaughter. The Canadian Veterinary Journal, 60, 423.
- Cockram MS and Hughes BO, 2018. Health and disease. Animal Welfare, 3, 141–159.
- Collier RJ and Gebremedhin KG, 2015. Thermal biology of domestic animals. Annual Review in Animal Bioscience, 3, 513–532.
- Collier RJ, Baumgard LH, Zimbelman RB and Xiao Y, 2019. Heat stress: physiology of acclimation and adaptation. Animal Frontiers, 9, 12–19.
- Collin A, Vaz MJ and Le Dividich J, 2002. Effects of high temperature on body temperature and hormonal adjustments in piglets. Reproduction Nutrition Development, 42, 45–53.



- Condotta IC, Brown-Brandl TM, Stinn JP, Rohrer GA, Davis JD and Silva-Miranda KO, 2018. Dimensions of the modern pig. Transactions of the ASABE, 61, 1729–1739.
- Consortium of the Animal Transport Guides Project (CATGP, 2017) (2017-rev1). Revision May 2018 'Guide to good practices for the transport of pigs.' Available online: http://animaltransportguides.eu/wp-content/uploads/2016/ 05/D3-Pigs-Revised-Final.pdf [Accessed: 5 March 2022].
- Correa JA, Torrey S, Devillers N, Laforest JP, Gonyou HW and Faucitano L, 2010. Effects of different moving devices at loading on stress response and meat quality in pigs. Journal of Animal Science, 88, 4086–4093.
- Dahl-Pedersen K and Herskin MS, 2021. Transportation of cattle and pigs between EU member states 2014–2018– can data from TRACES be used to Create Overview and Inform about Potential Welfare Consequences? Journal of Applied Animal Welfare Science, 1–14.
- Dahl-Pedersen K, Bonde MK, Herskin MS, Jensen KH, Kaiser M and Jensen HE, 2013. Pathogenesis and pathology of shoulder ulcerations in sows with special reference to peripheral nerves and behavioural responses to palpation. The Veterinary Journal, 198, 666–671.
- Dahl-Pedersen K, Foldager L, Herskin MS, Houe H and Thomsen PT, 2018. Lameness scoring and assessment of fitness for transport in dairy cows: agreement among and between farmers, veterinarians and livestock drivers. Research in Veterinary Science, 119, 162–166.
- Dalla Costa OA, Costa MJRPD, Ludke JV, Coldebella A, Kich JD, Peloso JV and Dalla Roza D, 2008. Relation of fasting time during pre-slaughter management to weight loss, weight of stomach contents and incidence of gastric ulcer in pigs. Ciência Rural, 38, 199–205.
- Dalla Costa FA, Devillers N, Paranhos da Costa MJ and Faucitano L, 2016. Effects of applying preslaughter feed withdrawal at the abattoir on behaviour, blood parameters and meat quality in pigs. Meat Science, 119, 89–94. https://doi.org/10.1016/j.meatsci.2016.03.033
- Dalla Costa FA, Lopes LS and Dalla Costa OA, 2017. Effects of the truck suspension system on animal welfare, carcass and meat quality traits in pigs. Animals, 7, 5.
- Dalla Costa OA, Dalla Costa FA, Feddern V, dos Santos LL, Coldebella A, Gregory NG and de Lima GJ, 2019. Risk factors associated with pig pre-slaughtering losses. Meat Science, 155, 61–68.
- D'Allaire S, Drolet R and Brodeur D, 1996. Sow mortality associated with high ambient temperatures. The Canadian Veterinary Journal, 37, 237.
- Dalmau A, Temple D, Rodríguez P, Llonch P and Velarde A, 2009. Application of the Welfare Quality[®] protocol at pig slaughterhouses. Animal welfare, 18, 497–505.
- Dalmau A, Geverink NA, Van Nuffel A, Van Steenbergen L, Van Reenen K, Hautekiet V, Vermeulen K, Velarde A and Tuyttens FAM, 2010. Repeatability of lameness, fear and slipping scores to assess animal welfare upon arrival in pig slaughterhouses. Animal, 4, 804–809.
- Department for Transport, United Kingdom, 2014. Simplified guidelines. Simplified guidance on European Union (EU) rules on drivers' hours and working time. Available online: https://assets.publishing.service.gov.uk/ overnment/uploads/system/uploads/attachment_data/file/856360/simplified-guidance-eu-drivers-hours-working time-ules.pd
- De Araujo IE, Kringelbach ML, Rolls ET and McGlone F, 2003. Human cortical responses to water in the mouth, and the effects of thirst. Journal of neurophysiology, 90, 1865–1876.
- De Hollander CA, Knol EF, Heuven HCM and van Grevenhof EM, 2015. Interval from last insemination to culling: II. Culling reasons from practise and the correlation with longevity. Livestock Science, 181, 25–30.
- De Jong E, Appeltant R, Cools A, Beek J, Boyen F, Chiers K and Maes D, 2014. Slaughterhouse examination of culled sows in commercial pig herds. Livestock Science, 167, 362–369.
- Dewey CE, Haley C, Widowski T, Poljak Z and Friendship RM, 2009. Factors associated with in-transit losses of fattening pigs. Animal Welfare, 18, 355–361.
- Driessen B, Peeters E and Geers R, 2008. Influence of olfactory substances on the heart rate and lying behaviour of pigs during transport simulation. Animal Welfare, 17, 155–160.
- Driessen B, Van Beirendonck S and Buyse J, 2020a. Effects of transport and lairage on the skin damage of pig carcasses. Animals, 10, 1–15.
- Driessen B, Freson L and Buyse J, 2020b. Fasting finisher pigs before slaughter influences pork safety, pork quality and animal welfare. Animals, 10, 2206.
- DVFA (The Danish Veterinary and Food Administration), 2019. Transport guide. Available online: foede varestyrelsen.dk/SiteCollectionDocuments/Dyresundhed/Transportegnethed/Transportguide%20-%20Vurdering %20af%20transportegnethed%20og%20%20skader%20opstået%20under%20transport.pdf
- EFSA (European Food Safety Authority), 2004. Opinion of the Scientific Panel on Animal Health and Welfare on a request from the Commission related to the Standards for the microclimate inside animal road transport vehicles. EFSA Journal 2004;2(10):122, 25 pp. https://doi.org/10.2903/j.efsa.2004.122
- EFSA AHAW Panel (EFSA Panel on Animal Health and Welfare), 2011. Scientific Opinion concerning the welfare of animals during transport. EFSA Journal 2011; 9(1):1966, 125 pp. https://doi.org/10.2903/j.efsa.2011.1966
- EFSA AHAW Panel (EFSA Panel on Animal Health and Welfare), 2012. Guidance on risk assessment for animal welfare. EFSA Journal 2012;10(1):2513, 30 pp. https://doi.org/10.2903/j.efsa.2012.2513



- EFSA AHAW Panel (EFSA Panel on Animal Health and Animal Welfare), More S, Bicout D, Botner A, Butterworth A, Calistri P, Depner K, Edwards S, Garin-Bastuji B, Good M, Gortazar Schmidt C, Michel V, Miranda MA, Saxmose Nielsen S, Velarde A, Thulke H-H, Sihvonen L, Spoolder H, Stegeman JA, Raj M, Willeberg P, Candiani D and Winckler C, 2017. Scientific Opinion on the animal welfare aspects in respect of the slaughter or killing of pregnant livestock animals (cattle, pigs, sheep, goats, horses). EFSA Journal 2017;15(5):4782, 96 pp. https://doi.org/10.2903/j.efsa.2017.4782
- EFSA AHAW Panel (EFSA Panel on Animal Health and Welfare), Nielsen SS, Alvarez J, Bicout DJ, Calistri P, Depner K, Drewe JA, Garin-Bastuji B, Gonzales Rojas JL, Gortázar Schmidt C, Michel V, Miranda Chueca MÁ, Roberts HC, Sihvonen LH, Spoolder H, Stahl K, Viltrop A, Winckler C, Candiani D, Fabris C, Van der Stede Y and Velarde A, 2020. Scientific Opinion on the welfare of pigs at slaughter. EFSA Journal 2020;18(6):6148, 113 pp. https://doi.org/10.2903/j.efsa.2020.6148
- EFSA AHAW Panel (EFSA Panel on Animal Health and Welfare), Nielsen SS, Alvarez J, Bicout DJ, Calistri P, Canali E, Drewe JA, Garin-Bastuji B, Gonzales Rojas JL, Gortázar Schmidt C, Herskin M, Miranda Chueca MA, Michel V, Padalino B, Pasquali P, Roberts HC, Spoolder H, Stahl K, Velarde A, Viltrop A, Edwards S, Sean A, Candiani S, Fabris C, Lima E, Mosbach-Schulz O, Rojo Gimeno C, Van der Stede Y, Vitali M and Winckler C, 2022a. Scientific Opinion on the methodological guidance for the development of animal welfare mandates in the context of the Farm To Fork Strategy. EFSA Journal 2022;20(7):7403, 45 pp. https://doi.org/10.2903/j.efsa.2022.7403
- EFSA AHAW Panel (EFSA Panel on Animal Health and Welfare), Nielsen SS, Alvarez J, Bicout DJ, Calistri P, Canali E, Drewe JA, Garin-Bastuji B, Gonzales Rojas JL, Gortázar Schmidt C, Herskin M, Miranda Chueca MA, Michel V, Padalino B, Pasquali P, Roberts HC, Spoolder H, Stahl K, Velarde A, Viltrop A, Edwards S, Sean A, Candiani S, Fabris C, Lima E, Mosbach-Schulz O, Rojo Gimeno C, Van der Stede Y, Vitali M and Winckler C, 2022b. Welfare of pigs on farm. https://doi.org/10.2903/j.efsa.2022.742
- EFSA AHAW Panel (EFSA Panel on Animal Health and Welfare), Nielsen SS, Alvarez J, Bicout DJ, Calistri P, Canali E, Drewe JA, Garin-Bastuji B, Gonzales Rojas JL, Gortázar Schmidt C, Herskin M, Miranda Chueca MA, Michel V, Padalino B, Pasquali P, Roberts HC, Spoolder H, Stahl K, Velarde A, Viltrop A, Winckler C, Mitchell L, Vinco LJ, Voslarova E, Candiani D, Mosbach-Schulz O, Van der Stede Y and Velarde A, 2022c. Welfare of domestic birds and rabbits transported in containers. EFSA Journal 2022; https://doi.org/10.2903/j.efsa.2022.7441
- EFSA Scientific Committee, Benford D, Halldorsson T, Jeger MJ, Knutsen HK, More S, Naegeli H, Noteborn H, Ockleford C, Ricci A, Rychen G, Schlatter JR, Silano V, Solecki R, Turck D, Younes M, Craig P, Hart A, Von Goetz N, Koutsoumanis K, Mortensen A, Ossendorp B, Germini A, Martino L, Merten C, Mosbach-Schulz O, Smith A and Hardy A, 2018. Scientific Opinion on the principles and methods behind EFSA's Guidance on Uncertainty Analysis in Scientific Assessment. EFSA Journal 2018;16(1):5122, 235 pp. https://doi.org/10.2903/j.efsa.2018.5122
- Eisemann JH, Morrow WM, See MT, Davies PR and Zering K, 2002. Effect of feed withdrawal prior to slaughter on prevalence of gastric ulcers in pigs. Journal of the American Veterinary Medical Association, 220, 503–506.
- Ekkel ED, Spoolder HAM, Hulseggec I and Hopster H, 2003. Lying characteristics as determinants for space requirements in pigs. Applied Animal Behaviour Science., 80, 19–30.
- Engblom L, Lundeheim N, Dalin AM and Andersson K, 2007. Sow removal in Swedish commercial herds. Livestock Science, 106, 76–86.
- van Essen GJ, te Lintel HM, Sorop O, Heinonen I, van der Velden J, Merkus D and Duncker DJ, 2018. Cardiovascular function of modern pigs does not comply with allometric scaling laws. Scientific Reports, 8, 792.
- EURCAW, 2021. EU Reference Center for Animal Welfare -Pigs factsheets: fitness for transport of pigs. Available online: https://library.wur.nl/WebQuery/edepot/545867 [Accessed: 19 April 2022].
- Fàbrega E, Puigvert X, Soler J, Tibau J and Dalmau A, 2013. Effect of on farm mixing and slaughter strategy on behaviour, welfare and productivity in Duroc finished entire male pigs. Applied Animal Behaviour Science, 143, 31–39.
- Faucitano L, 2001. Causes of skin damage to pig carcasses. Canadian Journal of Animal Science, 81, 39–45.
- Faucitano L, 2018. Preslaughter handling practices and their effects on animal welfare and pork quality. Journal of Animal Science, 96, 728–738. https://doi.org/10.1093/jas/skx064
- Faucitano L and Goumon S, 2018. Transport to slaughter and associated handling. In: Špinka M (ed.). Advances in Pig Welfare. Woodhead Publishing, London. pp. 261–293.
- Faucitano L and Lambooij E, 2019. Transport of pigs. In: Grandin T (ed.). Livestock Handling and Transport, 5th Edition. CAB International.
- Faucitano L and Pedernera C, 2016. Reception and unloading of animals. In: Velarde A and Raj M (eds.), Animal Welfare at Slaughter. 5 m Publishing. Sheffield, UK, pp. 33–50.
- Faucitano L and Raj M, 2022. Chapter 5: Pigs In Preslaughter handling and slaughter of meat animals (pp. 198-202). Wageningen Academic Publishers. https://doi.org/10.3920/978-90-8686-924-4
- Faucitano L, Saucier L, Correa JA, Méthot S, Giguère A, Foury A, Mormède P and Bergeron R, 2006. Effects of feed texture, meal frequency and pre-slaughter fasting on carcass and meat quality, and urinary cortisol in pigs. Meat Science, 74, 697–703.
- Filippitzi ME, Brinch Kruse A, Postma M, Sarrazin S, Maes D, Alban L, Nielsen LR and Dewulf J, 2018. Review of transmission routes of 24 infectious diseases preventable by biosecurity measures and comparison of the implementation of these measures in pig herds in six European countries. Transboundary and Emerging Diseases, 65, 381–398.



- Fogsgaard KK, Herskin MS and Thodberg K, 2018. Transportation of cull sows a descriptive study of the clinical condition of cull sows before transportation to slaughter. Translational Animal Science. https://doi.org/10.1093/tas/txy057
- Foldager L, Kaiser M and Herskin MS, 2021. Note on prediction of height from weight for pigs weighing 5 to 40 kg. 10 pages. Advisory memoran-dum from DCA Danish Centre for Food and Agriculture, Aarhus Univer-sity, submitted: 17. December 2021.
- Fox J, Widowski T, Torrey S, Nannoni E, Bergeron R, Gonyou HW, Brown JA, Crowe T, Mainau E and Faucitano L, 2014. Water sprinkling market pigs in a stationary trailer. 1. Effects on pig behaviour, gastrointestinal tract temperature and trailer micro-climate. Livestock Science, 160, 113–123.
- Franchi GA, Jensen MB, Foldager L, Larsen M and Herskin MS, 2022. Effects of dietary and milking frequency changes and administration of cabergoline on clinical udder characteristics in dairy cows during dry-off. Research in Veterinary Science, 143, 88–98.

Fraser D and Duncan IJ, 1998. 'Pleasures', 'pains' and animal welfare: toward a natural history of affect.

- Frese DA, Reinhardt CD, Bartle SJ, Rethorst DN, Hutcheson JP, Nichols WT and Thomson DU, 2016. Cattle handling technique can induce fatigued cattle syndrome in cattle not fed a beta adrenergic agonist. Journal of Animal Science, 94, 581–591.
- Frotin P, Chevillon P, Rousseau P and Prunier A, 2002. Systèmes d'abreuvement et comportement des porcs lors d'un transport de 24 h. Techni Porc., 25, 15–24.
- Garcia A, Johnson AK, Ritter MJ, Calvo-Lorenzo MS and McGlone JJ, 2019. Transport of market pigs: improvements in welfare and economics. In: Grandin T (ed.). Livestock Handling and Transport. CABI, Wallingford, OX. pp. 328–346.
- Gebresenbet G, Aradom S, Bulitta FS and Hjerpe E, 2011. Vibration levels and frequencies on vehicle and animals during transport. Biosystems Engineering, 110, 10–19.
- Gerritzen MA, Hindle VA, Steinkamp K, Reimert HGM, van der Werf JTN and Marahrens M, 2013. The effect of reduced loading density on pig welfare during long distance transport. Animal Welfare, 7, 1849–1857.
- Gesing LM, Johnson AK, Selsby JT, Feuerbach C, Hill H, Faga M and Ritter MJ, 2010. Effects of presorting on stress responses at loading and unloading and the impact on transport losses from market-weight pigs. The Professional Animal Scientist, 26, 603–610.
- Goumon S and Faucitano L, 2017. Influence of loading handling and facilities on the subsequent response to preslaughter stress in pigs. Livestock Science [Internet]. 200, 6–13. https://doi.org/10.1016/j.livsci.2017.03.021
- Goumon S, Faucitano L, Bergeron R, Crowe T, Connor ML and Gonyou HW, 2013. Effect of ramp configuration on easiness of handling, heart rate, and behavior of near-market weight pigs at unloading. Journal of Animal Science, 91, 3889–3898.
- Goumon S, Brown JA, Faucitano L, Bergeron R, Widowski TM, Crowe T and Gonyou HW, 2013b. Effects of transport duration on maintenance behavior, heart rate and gastrointestinal tract temperature of market-weight pigs in 2 seasons. Journal of animal science, 91, 4925–4935.
- Gourdine JL, Rauw WM, Gilbert H and Poullet N, 2021. The genetics of thermoregulation in pigs: a review. Frontiers in Veterinary Science, 8.
- Grandin T, 2001. Perspectives on transportation issues: the importance of having physically fit cattle and pigs. Journal of Animal Science, 79(suppl_E), E201–E207.
- Grandin T, 2012. Recommended Animal Handling Guidelines and Audit Guide: a Systematic Approach to Animal Welfare. American Meat Institute Foundation, Washington, DC. 108 pp.
- Grandin T, 2016a. Practical methods to improve animal handling and restraint. In: Velarde A and Raj M (eds.), Animal Welfare at Slaughter. 5 m Publishing, Sheffield, UK. pp. 71–90.
- Grandin T, 2016b. Transport fitness of cull sows and boars: a comparison of guidelines on fitness for transport. Animals, 6, 77. https://doi.org/10.3390/ani6120077

Grandin T (ed.), 2019. Livestock handling and transport. CABI.

- Greenwood EC, Plush KJ, van Wettere HEJ and Hughes PE, 2014. Hierarchy formation in newly mixed, group housed sows and management strategies aimed at reducing its impact. Applied Animal Behaviour Science.
- Guàrdia MD, Gispert M and Diestre A, 1996. Mortality rates during transport and lairage in pigs for slaughter. Meat Focus International (United Kingdom).
- Hayley C, Dewey CE, Widowski T and Friendship R, 2010. Relationship between estimated finishing-pig space allowance and in-transit loss in a retrospective survey of 3 packing plants in Ontario in 2003. Canadian Journal of Veterinary Research, 74, 178–184.
- Herbut E, Sosnówka-Czajka E and Skomorucha I, 2018. Air ionizAtion in livestock buildings a review. Annals of Animal Science, 18, 899–905.
- Herskin MS and Di Giminiani P, 2018. Pain in pigs: characterisation, mechanisms and indicators. Advances in Pig Welfare, 325–355.
- Herskin MS, Bonde MK, Jorgensen E and Jensen KH, 2011. Decubital shoulder ulcers in sows: a review of classification, pain and welfare consequences. Animal, 5, 757–766.
- Herskin MS, Fogsgaard KK, Erichsen D, Bonnichsen M, Gaillard C and Thodberg K, 2017. Housing of cull sows in the hours before transport to the abattoir an initial description of sow behaviour while waiting in a transfer vehicle. Animals, 7.



- Herskin MS, Christensen SW and Rousing T, 2020. Handling and moving cull sows upon arrival at the slaughterhouse effects of small versus large groups of sows. Applied Animal Behaviour Science, 232, 105113. https://doi.org/10.1016/j.applanim.2020.105113
- Herskin MS, Holm C and Thodberg K, 2020a. Clinical and behavioural consequences of on-farm mixing of cull sows after weaning. Applied Animal Behaviour Science, 228, 105028.
- Herskin MS, Aaslyng MD, Anneberg I, Thomsen PT, Gould LM and Thodberg K, 2020b. Significant variation in the management of cull sows before transport for slaughter: results from a survey of Danish pig farmers. Veterinary Record, 186, 185–185.
- Herskin MS, Gerritzen MA, Marahrens M, Bracke MBM and Spoolder HAM, 2021. Review of fitness for transport of pigs. EURCAW-Pigs review. Avilable online: https://edepot.wur.nl/546057
- Hillman E, Mayer C and Schrader L, 2004. Lying behaviour and adrenocortical response as indicators of the thermal tolerance of pigs of different weights. Animal Welfare, 13, 329–335.
- Hody S, Rogister B, Leprince P, Wang F and Croisier J-L, 2011. Muscle fatigue experienced during maximal eccentric exercise is predictive of the plasma creatine kinase (CK) response. Scandinavian Journal of Medicine and Science in Sports., 23, 501–507.
- Hook TJ, Stookey JM and Wagner H, 2010. Rethinking cull boar transport. Canadian Veterinary Journal., 51, 315–322.
- Hörtenhuber SJ, Schauberger G, Mikovits C, Schönhart M, Baumgartner J, Niebuhr K and Zollitsch W, 2020. The effect of climate change-induced temperature increase on performance and environmental impact of intensive pig production systems. Sustainability, 12, 9442.
- Hovmand-Hansen T, Nielsen SS, Jensen TB, Vestergaard K, Nielsen MBF and Jensen HE, 2021a. Survival of pigs with different characteristics of umbilical outpouching in a prospective cohort study of Danish pigs. Preventive Veterinary Medicine, 191, 105343.
- Hovmand-Hansen T, Jensen TB, Vestergaard K, Nielsen MBF, Leifsson PS and Jensen HE, 2021b. Early risk factors, development, disappearance and contents of umbilical outpouching in Danish pigs. Livestock Science, 251, 104654.
- Huang T, Rong L, Zhang G, Brandt P, Bjerg B, Pedersen P and Granath SW, 2021a. A two-node mechanistic thermophysiological model for pigs reared in hot climates–Part 1: physiological responses and model development. Biosystems Engineering, 212, 302–317.
- Huang T, Rong L, Zhang G, Brandt P, Bjerg B, Pedersen P and Granath SW, 2021b. A two-node mechanistic thermophysiological model for pigs reared in hot climates—Part 2: model performance assessments. Biosystems Engineering, 212, 318–335.
- Huang T, Zhang G, Brandt P, Bjerg B, Pedersen P and Rong L, 2022. An effective temperature derived from a mechanistic thermophysiological model for sows reared in hot climates. Biosystems Engineering, 220, 19–38.
- Humane slaughter association, 2022. Transport of Livestock. Online guide. Available online: https://www.hsa.org. k/transport-of-livestock-introduction/introduction-8
- Huynh TTT, Aarnink AJA, Verstegen MWA, Gerrits WJJ, Heetkamp MJW, Kemp B and Canh TT, 2005. Effects of increasing temperatures on physiological changes in pigs at different relative humidities. Journal of Animal Science, 83, 1385–1396.
- Ingram D, 1965. Evaporative cooling in the pig. Nature, 207, 415-416.
- Isaacson RE, Firkins LD, Weigel RM, Zuckermann FA and DiPietro JA, 1999. Effect of transportation and feed withdrawal on shedding of Salmonella typhimurium among experimentally infected pigs. American Journal of Veterinary Research, 60, 1155–1158.
- Jensen MB and Vestergaard M, 2021. Invited review: freedom from thirst—Do dairy cows and calves have sufficient access to drinking water? Journal of Dairy Science, 104, 11368–11385.
- Jensen MB, Schild SL, Theil PK, Andersen HL and Pedersen LJ, 2016. The effect of varying duration of water restriction on drinking behaviour, welfare and production of lactating sows. Animal, 10, 961–969.
- Jones TA, Waitt C and Dawkins MS, 2010. Sheep lose balance, slip and fall less when loosely packed in transit where they stand close to but not touching their neighbours. Applied Animal Behaviour Science, 123, 16–23.
- Kaur R, 2018. Evaluation of the optimal space allowance for nursery pigs. Unpublished MSc thesis, University of Saskatchewan. Available online: https://harvest.usask.ca/handle/10388/8467 [Accessed: 23 April 2022].
- Kephart KB and Mills EW, 2005. Effect of withholding feed from swine before slaughter on carcass and viscera weights and meat quality. Journal of Animal Science, 83, 715–721.
- Kephart KB, Harper MT and Raines CR, 2010. Observations of market pigs following transport to a packing plant. Journal of Animal Science, 88, 2199–2203.
- Kilbride AL, Gillman CE and Green LE, 2009. A cross-sectional study of the prevalence of lameness in finishing pigs, gilts and pregnant sows and associations with limb lesions and floor types on commercial farms in England. Animal Welfare, 18, 215–224.
- Kingma BR, Frijns AJ, Schellen L and van Marken Lichtenbelt WD, 2014. Beyond the classic thermoneutral zone: including thermal comfort. Temperature, 1, 142–149.
- Knight JW, Bazer FW, Thatcher WW and Franke DE, 1977. Conceptus development in intact and unilaterally hysterectomized-ovariectomized gilts: Interrelations among hormonal status, placental development, fetal fluids and fetal growth. Journal of Animal Science, 44, 620–636.



Knowles G, 1999. A review of the road transport of cattle. Veterinary Record, 144, 197–201.

- Knox R, Levis D, Safranski T and Singleton W, 2008. An update on North American boar stud practices. Theriogenology, 70, 1202–1208.
- Lambooij E, 2014. Transport of pigs. Livestock handling and transport, (Ed. 4), 280–297.
- Lambooij E and Engel B, 1991. Transport of slaughter pigs by truck over a long distance: some aspects of loading density and ventilation. Livestock Production Science, 28, 163–174.
- Lambooij E, Garssen GL, Walstra P, Mateman G and Merkus GSM, 1985. Transport of pigs by car for two days; some aspects of watering and loading density. Livestock Production Science, 13, 289–299.
- Larsen T, Kaiser M and Herskin MS, 2015. Does the presence of shoulder ulcers affect the behaviour of sows? Research Veterinary Science, 98, 19–24. https://doi.org/10.1016/j.rvsc.2014.11.001
- Larsen MLV, Wang M and Norton T, 2021. Information Technologies for Welfare Monitoring in Pigs and Their Relation to Welfare Quality[®]. Sustainability, 13, 692.
- Leitner G, Jacoby S, Maltz E and Silanikove N, 2007. Casein hydrolyzate intramammary treatment improves the comfort behavior of cows induced into dry-off. Livestock Science, 110, 292–297.
- Lewis NJ and Berry RJ, 2006. Effects of season on the behaviour of early-weaned piglets during and immediately following transport. Applied Animal Behaviour Science, 100, 182–192.
- Llonch P, King EM, Clarke KA, Downes JM and Green LE, 2015. A systematic review of animal based indicators of sheep welfare on farm, at market and during transport, and qualitative appraisal of their validity and feasibility for use in UK abattoirs. The Veterinary Journal, 206, 289–297.
- Lowe J, Gauger P, Harmon K, Zhang J, Connor J, Yeske P, Loula T, Levis I, Dufresne L and Main R, 2014. Role of transportation in spread of porcine epidemic diarrhea virus infection, United States. Emerging Infectious Diseases, 20, 872.
- Lucas EM, Randall JM and Meneses JF, 2000. Potential for evaporative cooling during heat stress periods in pig production in Portugal (Alentejo). Journal of agricultural engineering research, 76, 363–371.
- Lucy MC and Safranski TJ, 2017. Heat stress in pregnant sows: thermal responses and subsequent performance of sows and their offspring. Mol Reprod Dev, 84, 946–956. https://doi.org/10.1002/mrd.22844
- Lykke L, Blaabjerg L and Hartung J, 2007a. Investigation of pig transports for more than 8 hours in cold and warm weather conditions and of the requirements for ventilation during the transport. Danish Meat Research Institute report, 24th July 2007. Available online: https://www.teknologisk.dk/_/media/64606_Investigation% 20of%20pig%20transports.pdf [Accessed: 16 January 2022].
- Lykke L, Blaabjerg L and Hartung J, 2007b. Investigation of pig transports for more than 8 hours in cold and warm weather conditions and of the requirements for ventilation during the transport. Report from Danish Meat Research Institute, Taastrup, Denmark. 81 pp.
- Machado NA, Barbosa-Filho JA, Ramalho GL, Pandorfi H and Da Silva IJ, 2021. Trailer heat zones and their relation to heat stress in pig transport. Engenharia Agrícola, 41, 427–437.
- Magnani D, Cafazzo S, Calá P, Razzuoli E, Amadori M, Bernardini D, Gerardi G and Nanni Costa L, 2014. Effect of long transport and environmental conditions on behaviour and blood parameters of postweaned piglets with different reactivity to backtest. Livestock Science, 162, 201–208.
- Malena M, Voslarova E, Kozak A, Belobradek P, Bedanova I, Steinhauser L and Vecerek V, 2007. Comparison of mortality rates in different categories of pigs and cattle during transport for slaughter. Acta Veterinary Brno., 76, S109–S116.
- Marchant-Forde RM and Marchant-Forde JN, 2004. Pregnancy-related changes in behavior and cardiac activity in primiparous pigs. Physiology and Behavior, 82, 815–825.
- Mason GJ and Burn C, 2011. Behavioural restriction. In: Mench JA, Olsson A and Hughes BO (eds.), Animal Welfare Apple, CAB International, Wallingford, pp. 98-119.
- Mayorga EJ, Renaudeau D, Ramirez BC, Ross JW and Baumgard LH, 2019. Heat stress adaptations in pigs. Animal Frontiers, 9, 54–61.
- Mayorga EJ, Ross JW, Keating AF, Rhoads RP and Baumgard LH, 2020. Biology of heat stress; the nexus between intestinal hyperpermeability and swine reproduction. Theriogenology, 154, 73–83.
- McConn BR, Gaskill BN, Schinckel AP, Green-Miller AR, Lay DC Jr and Johnson JS, 2021. Thermoregulatory and physiological responses of nonpregnant, mid-gestation, and late-gestation sows exposed to incrementally increasing dry bulb temperature. Journal of Animal Science, 99, 1–8.
- McGlone J, Johnson A, Sapkota A and Kephart R, 2014. Establishing bedding requirements during transport and monitoring skin temperature during cold and mild seasons after transport for finishing pigs. Animals, 4, 241–253.
- McKinley MJ and Johnson AK, 2004. The physiological regulation of thirst and fluid intake. Physiology, 19, 1-6.

McMillan FD (ed.), 2020. Mental health and well-being in animals. CABI.

- McPherson RL, Ji F, Wu G, Blanton JR and Kim SW, 2004. Growth and compositional changes of fetal tissues in pigs. Journal of Animal Science., 82, 2534–2540.
- Miguel-Pacheco G and Seddon YM, 2021. Developmental influences on pig behaviour. In: Edwards SA (ed.). Understanding the behaviour and improving the welfare of pigs. Burleigh Dodds Science Publishing Limited, Cambridge, UK.



- Miranda-De La Lama GC, Villarroel M and María GA, 2014. Livestock transport from the perspective of the preslaughter logistic chain: a review. Meat Science, 98, 9–20.
- Miranda-de la Lama GC, Bermejo-Poza R, Formoso-Rafferty N, Mitchell M, Barreiro P and Villarroel M, 2021. Longdistance Transport of finisher pigs in the Iberian Peninsula: effects of season on thermal and enthalpy conditions, welfare indicators and meat pH. Animals, 11, 2410.
- Mitchell MA, 2006. Using physiological models to define environmental control strategies. In: Gous RM, Morris TR and Fisher C (eds.), Mechanistic Modelling in Pig and Poultry Production. CABI International, Wallingford, Oxfordshire, UK. pp. 209–228.
- Mitchell MA and Kettlewell PJ, 2008. Engineering and design of vehicles for long distance road transport of livestock (ruminants, pigs and poultry). Veterinaria Italiana, 44, 201–213.
- Moak K, Bergeron R, Conte S, Bohrer BM, Arrazola A, Devillers N and Faucitano L, 2022a. Use of two novel trailer types for transportation of pigs to slaughter. 1. Effects on trailer microclimate, pig behaviour, physiological response, and meat quality under Canadian summer conditions. Canadian Journal of Animal Science (accepted for publication; https://doi.org/10.1139/CJAS-2022-0023).
- Moak K, Bergeron R, Conte S, Bohrer BM, Ferreira GA, Vero JG, Aboagye G, Arrazola A, Devillers N and Faucitano L, 2022b. Use of two novel trailer types for transportation of pigs to slaughter. 2. Effects on trailer microclimate, pig behaviour, physiological response, and meat quality under Canadian winter conditions. Canadian Journal of Animal Science (accepted for publication; https://doi.org/10.1139/CJAS-2022-0024).
- Monin G and Sellier P, 1985. Pork of low technological quality with a normal rate of muscle pH fall in the immediate post-mortem period: the case of the Hampshire breed. Meat Science, 13, 49–63.
- Morris BK, Davis RB, Brokesh E, Flippo DK, Houser TA, Najar-Villarreal F, Turner KK, Williams JG, Stelzleni AM and Gonzalez JM, 2021. Measurement of the three-axis vibration, temperature, and relative humidity profiles of commercial transport trailers for pigs. Journal of Animal Science, 99, p.skab027.
- Mount LE, 1974. The concept of Thermal Neutrality. In: Monteith JL and Mount LE (eds.), Heat Loss from Animals and Man: Assessment and Control. Elsevier.
- Mount NC and Seabrook MF, 1993. A study of aggression when group housed sows are mixed. Applied Animal Behaviou Science, 36, 377–383.
- Moustsen VA, Lahrmann HP and D'Eath RB, 2011. Relationship between size and age of hyperprolific modern cross-bred sows. Livestock Science, 141, 272–275.
- Mutua JY, Notenbaert AMO, Paul BK, Rahimi J and Butterbach-Bahl K, 2020. Heat Stress Assessment for Dairy Cattle and Pig in Uganda.
- Nalon E, Conte S, Maes D, Tuyttens FAM and Devillers N, 2013. Assessment of lameness and claw lesions in sows. Livestock Science, 156, 10–23.
- Nannoni E, Widowski T, Torrey S, Fox J, Rocha LM, Gonyou H and Faucitano L, 2014. Water sprinkling market pigs in a stationary trailer. 2. Effects on selected exsanguination blood parameters and carcass and meat quality variation. Livestock Science, 160, 124–131.
- Nannoni E, Liuzzo G, Serraino A, Giacometti F, Martelli G, Sardi L, Vitali M, Romagnoli L, Moscardini E and Ostanello F, 2017. Evaluation of pre-slaughter losses of Italian heavy pigs. Animal Production Science, 57, 2072–2081. (Abstract only presently)
- Neumann EJ and Hall WF, 2019. Disease control, prevention, and elimination. In: Zimmerman JJ, Karriker LA, Ramirez A, Schwartz KJ, Stevenson GW and Zhang J (eds.). Diseases of swine. John Wiley & Sons, Hoboken, NJ, USA, pp. 123–157.
- Nielsen BL, Dybkjær L and Herskin MS, 2011. Road transport of farm animals: effects of journey duration on animal welfare. Animal, 5, 415–427.
- Nielsen SS, Michelsen AM, Jensen HE, Barington K, Opstrup KV and Agger JF, 2014. The apparent prevalence of skin lesions suspected to be human-inflicted in Danish finishing pigs at slaughter. Preventive Veterinary Medicine, 117, 200–206.
- Noblet J, Dourmad JY, Le Dividich J and Dubois S, 1989. Effect of ambient temperature and addition of straw or alfalfa in the diet on energy metabolism in pregnant sows. Livestock Production Science, 21, 309–324.
- Norris AL and Kunz TH, 2012. Effects of solar radiation on animal thermoregulation. Solar Radiation, 195–220.
- Nowak R, Porter RH, Lévy F, Orgeur P and Schaal B, 2000. Role of mother-young interactions in the survival of offspring in domestic mammals. Reviews of Reproduction., 5, 153–163.
- Padalino B, Tullio D, Cannone S and Bozzo G, 2018. Road transport of farm animals: mortality, morbidity, species and country of origin at a Southern Italian control post. Animals, 8, 155.
- Parrott RF, Thornton N, Forsling ML and Delaney CE, 1987. Endocrine and behavioural factors affecting water balance in sheep subjected to isolation stress. Journal of Endocrinology, 112, 305–310.
- Pascual-Alonso M, Miranda-De la Lama GC, Aguayo-Ulloa L, Villarroel M, Mitchell M and María GA, 2017. Thermophysiological, haematological, biochemical and behavioural stress responses of sheep transported on road. Journal of Animal Physiology and Animal Nutrition, 101, 541–551.
- Passafaro TL, Van de Stroet D, Bello NM, Williams NH and Rosa GJ, 2019. Generalized additive mixed model on the analysis of total transport losses of market-weight pigs. Journal of Animal Science, 97, 2025–2034.
- Pastorelli G, Musella M, Zaninelli M, Tangorra F and Corino C, 2006. Static spatial requirements of growing-finishing and heavy pigs. Livestock Science, 105, 260–264.



- Peeters E, Deprez K, Beckers F, De Baerdemaeker J, Aubert AE and Geers R, 2008. Effect of driver and driving style on the stress responses of pigs during a short journey by trailer. Animal Welfare-Potters Bar then Wheathampstead, 17, 189.
- Pereira TL, Titto EAL, Conte S, Devillers N, Sommavilla R, Diesel T, Dalla Costa FA, Guay F, Friendship R, Crowe T and Faucitano L, 2018. Application of a ventilation fan-misting bank on pigs kept in a stationary trailer before unloading: effects on trailer microclimate, and pig behaviour and physiological response. Livestock Science, 216, 67–74. https://doi.org/10.1016/j.livsci.2018.07.013
- Peterson E, Zemmenga M, Hagerman A and Akkina J, 2017. Use of temperature, humidity and slaughter condemnation data to predict increases in transport losses in three classes of swine and resulting foregone revenue. Frontiers in Veterinary Science, 4, 67.
- Petherick JC, 1983a. A biological basis for the design of space in livestock housing. Farm Animal Housing and Welfare, 103–120.
- Petherick CJ, 1983b. A note on allometric relationships in large white x landrace pigs. Animal Science, 36, 497– 500.
- Petherick CJ and Philips CJC, 2009. Space allowances for confined livestock and their determination from allometric principles. Applied Animal Behaviour Science., 117, 1–2.
- Pilcher CM, Eliss M, Rojo-Gómez A, Curtis SE, Wolter BF, Peterson CM, Peterson BA, Ritter MJ and Brinkmann J, 2011. Effects of floor space during transport and journey time on indicators of stress and transport losses of market-weight pigs. Journal of Animal Science., 89, 3809–3818.
- Porcelluzzi A, 2013. Handbook for High Quality Control Posts for cattle, pigs and sheep. Available online: https://www.flipsnack.com/andpor/quality-control-post-handbook-english.html.
- Quiniou N and Noblet J, 1999. Influence of high ambient temperatures on performance of multiparous lactating sows. Journal of Animal Science, 77, 2124–2134.
- Raja SN, Carr DB, Cohen M, Finnerup NB, Flor H, Gibson S and Vader K, 2020. The revised IASP definition of pain: concepts, challenges, and compromises. Pain, 161, 1976.
- Randall JM, 1992. Human subjective response to lorry vibration: implications for farm animal transport. Journal of Agricultural Engineering Research, 52, 295–307.
- Randall JM, 1993. Environmental parameters necessary to define comfort for pigs, cattle and sheep in livestock transporters. Animal Science, 57, 299–307.
- Randall JM and Bradshaw RH, 1998. Vehicle motion and motion sickness in pigs. Animal Science, 66, 239–245.
- Randall JM, Cove MT and White RP, 1996. Resonant frequencies of broiler chickens. Animal Science, 62, 369–374.
- Rashamol VP, Sejian V, Pragna P, Lees AM, Bagath M, Krishnan G and Gaughan JB, 2019. Prediction models, assessment methodologies and biotechnological tools to quantify heat stress response in ruminant livestock. International journal of biometeorology, 63, 1265–1281.
- Renaudeau D, 2005. Effects of short-term exposure to high ambient temperature and relative humidity on thermoregulatory responses of European (Large White) and Carribbean (Creole) restrictively fed growing pigs. Animal Research, 54, 81–93.
- Renaudeau D and Dourmad JY, 2021. Future consequences of climate change for European Union pig production. Animal, 100372.
- Renaudeau D, Anais C, Tel L and Gourdine JL, 2010. Effect of temperature on thermal acclimation in growing pigs estimated using a nonlinear function. Journal of Animal Science, 88, 3715–3724.
- Renaudeau D, Gourdine JL and St-Pierre NR, 2011. A meta-analysis of the effects of high ambient temperature on growth performance of growing-finishing pigs. Journal of Animal Science., 89, 2220–2230.
- Renaudeau D, Collin A, Yahav S, De Basilio V, Gourdine JL and Collier RJ, 2012. Adaptation to hot climate and strategies to alleviate heat stress in livestock production. Animal, 6, 707–728.
- Reynolds R, Garner A and Norton J, 2019. Sound and vibration as research variables in terrestrial vertebrate models. ILAR Journal, 60, 159–174.
- Ribeiro BPVB, Lanferdini E, Palencia JYP, Lemes MAG, de Abreu MLT, de Souza CV and Ferreira RA, 2018. Heat negatively affects lactating swine: a meta-analysis. Journal of Thermal Biology, 74, 325–330.
- Riches HL and Guise HJ, 1997. Does the transport stocking density of 30-kg live weight pigs affect their heart rate and choice of posture? BSAP Occasional Publication, 20, 102–103.
- Riches HL, Guise HJ and Penny RHC, 1997. A national survey of transport conditions for pigs. Pig Journal, 38, 8–18.
- Ritter MJ, Ellis M, Bowman R, Brinkmann J, Curtis SE, DeDecker JM and Wolter BF, 2008. Effects of season and distance moved during loading on transport losses of market-weight pigs in two commercially available types of trailer. Journal of Animal Science, 86, 3137–3145.
- Ritter MJ, Yoder CL, Jones CL, Carr SN and Calvo-Lorenzo MS, 2020. Transport losses in market weight pigs: II. US incidence and economic impact. Translational Animal Science, 4, 1103–1112.
- Robbins LA, Green-Miller AR, Lay DC Jr, Schinckel AP, Johnson JS and Gaskill BN, 2021. Evaluation of sow thermal preference across three stages of reproduction. Journal of Animal Science, 99, skab202.
- Robinson JAB and Buhr MM, 2005. Impact of genetic selection on management of boar replacement. Therieogenology., 63, 668–678.



Rocha LM, Dionne A, Saucier L, Nannoni E and Faucitano L, 2015. Hand-held lactate analyzer as a tool for the real-time measurement of physical fatigue before slaughter and pork quality prediction. Animal, 9, 707–714.

Rocha LM, Velarde A, Dalmau A, Saucier L and Faucitano L, 2016. Can the monitoring of animal welfare parameters predict pork meat quality variation through the supply chain (from farm to slaughter). Journal of Animal Science, 94, 359–376.

Rodrigues VC, da Silva IJO, Vieira FMC and Nascimento ST, 2011. A correct enthalpy relationship as thermal comfort index for livestock. International Journal of Biometeorology, 55, 455–459.

- Rojek X, 2021. Protection of animals during transport: data on live animals transport. European Parliamentary Research Service report. Available online: https://www.europarl.europa.eu/RegData/etudes/BRIE/2021/690708/ EPRS_BRI(2021)690708_EN.pdf
- Romero MH, Sánchez JA and Hernandez RO, 2022. Field trial of factors associated with the presence of dead and non-ambulatory pigs during transport across three colombian slaughterhouses. Frontiers in Veterinary Science, 9. pp
- Romijn JA, Coyle EF, Sidossis LS, Zhang XJ and Wolfe R, 1995. Relationship between fatty acid delivery and fatty acid oxidation during strenuous exercise. Journal of Applied Physiology, 79, 1939–1945.
- Ross JW, Hale BJ, Gabler NK, Rhoads RP, Keating AF and Baumgard LH, 2015. Physiological consequences of heat stress in pigs. Animal Production Science, 55, 1381–1390.
- Rutherford KMD, Baxter EM, D'eath RB, Turner SP, Arnott G, Roehe R, Ask B, Sandøe P, Moustsen VA, Thorup F and Edwards SA, 2013. The welfare implications of large litter size in the domestic pig I: biological factors. Animal Welfare, 22, 199–218.
- Samad A, Obando Nuñez DR, Solis Castillo GC, Laquai B and Vogt U, 2020. Effect of relative humidity and air temperature on the results obtained from low-cost gas sensors for ambient air quality measurements. Sensors, 20, 5175. https://doi.org/10.3390/s20185175
- Santurtun E and Phillips CJC, 2015. The impact of vehicle motion during transport on animal welfare. Research Veterinary Science, 100, 303–308. https://doi.org/10.1016/j.rvsc.2015.03.018
- Sapolsky RM, 2002. Endocrinology of the stress-response. In: Becker JB, Breedlove SM, Crews D and McCarthy MM (eds.), Behavioral endocrinology (pp. 409–450). MIT Press.
- SCAHAW (Scientific Committee on Animal Health and Animal Welfare), 2002. The welfare of animals during transport (details for horses, pigs, sheep and cattle). European Commission, Report of the Scientific Committee on Animal Health and Animal Welfare. 130 pp.
- Schild SLA, Rousing T, Jensen HE, Barington K and Herskin MS, 2015a. Do umbilical outpouchings affect the behaviour or clinical condition of pigs during 6 h housing in a pre-transport pick-up facility? Research in Veterinary Science, 101, 126–131.
- Schild SLA, Brandt P, Rousing T and Herskin MS, 2015b. Does the presence of umbilical outpouchings affect the behaviour of pigs during the day of slaughter? Livestock Science, 176, 146–151.
- Schlader ZJ, Simmons SE, Stannard SR and Mündel T, 2011. The independent roles of temperature and thermal perception in the control of human thermoregulatory behavior. Physiology and Behavior, 103, 217–224.
- Schmid O and Kilchsperger R, 2010. Overview of animal welfare standards and initiatives in selected EU and third countries.
- Schwartzkopf-Genswein KS, Faucitano L, Dadgar S, Shand P, González LA and Crowe TG, 2012. Road transport of cattle, swine and poultry in North America and its impact on animal welfare, carcass and meat quality: a review. Meat Science, 92, 227–243. https://doi.org/10.1016/j.meatsci.2012.04.010
- Sejian V, Bhatta R, Gaughan JB, Dunshea FR and Lacetera N, 2018. Adaptation of animals to heat stress. Animal, 12, s431–s444.
- Semenova AA, Kuznetsova TG, Nasonova VV, Nekrasov RV and Bogolyubova NV, 2019. Myopathy as a destabilizing factor of meat quality formation. Theory and Practice of Meat Processing, 4, 24–31.
- Shaw FD and Tume RK, 1992. The assessment of pre-slaughter and slaughter treatments of livestock by measurement of plasma constituents—a review of recent work. Meat Science, 32, 311–329.
- Silanikove N, 2000. Effects of heat stress on the welfare of extensively managed domestic ruminants. Livestock Production Science, 67, 1–18.
- Smith RF, French NP, Saphier PW, Lowry PJ, Veldhuis JD and Dobson H, 2003. Identification of stimulatory and inhibitory inputs to the hypothalamic-pituitary-adrenal axis during hypoglycaemia or transport in ewes. Journal of Neuroendocrinology, 15, 572–585.
- Sommavilla R, Faucitano L, Gonyou H, Seddon Y, Bergeron R, Widowski T and Brown J, 2017. Season, transport duration and trailer compartment effects on blood stress indicators in pigs: relationship to environmental, behavioral and other physiological factors, and pork quality traits. Animals, 7, 8.
- Stephens DB, Sharman DF, Ingram DL and Bailey KJ, 1985. An analysis of some behavioural effects of the vibration and noise components of transport in pigs. Quarterly Journal of Experimental Physiology: Translation and Integration, 70, 211–217.
- Stinn JP and Xin H, 2014. Heat and moisture production rates of a modern U.S. swine breeding, gestation, and farrowing facility. Transactions of the ASABE., 57, 1517–1528. https://doi.org/10.13031/trans.57.10711



- Sutherland MA, McDonald A and McGlone JJ, 2009. Effects of variations in the environment, length of journey and type of trailer on the mortality and morbidity of pigs being transported to slaughter. Veterinary Record, 165, 13–18.
- Sutherland MA, Bryer PJ, Davis BL and McGlone J, 2010. A multidisciplinary approach to assess the welfare of weaned pigs during transport at three space allowances. Journal of Applied Animal Welfare Science., 13, 237–249.
- Talling JC, Waran NK, Wathes CM and Lines JA, 1998. Sound avoidance by domestic pigs depends upon characteristics of the signal. Applied Animal Behaviour Science, 58, 255–266.
- Tarrant PV, Kenny FJ, Harrington D and Murphy M, 1992. Long distance transportation of steers to slaughter: effect of stocking density on physiology, behaviour and carcass quality. Livestock Production Science, 30, 223–238.
- Tauson AH, Chwalibog A, Ludvigsen J, Jakobsen K and Thorbek G, 1998. Effect of short-term exposure to high ambient temperatures on gas exchange and heat production in boars of different breeds. Animal Science, 66, 431–440.
- Terlouw C and Bourguet C, 2022. Quantifying animal welfare preslaughter using behavioural, physiological and carcass and meat quality measures.
- Terlouw EMC, Arnould C, Aupérin B, Berri C, Le Bihan-Duval E, Deiss V and Mounier L, 2008. Pre-slaughter conditions, animal stress and welfare: current status and possible future research. Animal, 2, 1501–1517.
- Terlouw EMC, Picard B, Deiss V, Berri C, Hocquette JF, Lebret B and Gagaoua M, 2021. Understanding the determination of meat quality using biochemical characteristics of the muscle: stress at slaughter and other missing keys. Foods, 10, 84.
- Thodberg K, Fogsgaard KK and Herskin MS, 2019. Transportation of cull sows deterioration of clinical condition from departure and until arrival at the slaughter plant. Frontiers in Veterinary Science, 6, 28.
- Thodberg K, Gould LM, Støier S, Anneberg I, Thomsen PT and Herskin MS, 2020. Experiences and opinions of Danish livestock drivers transporting sows regarding fitness for transport and management choices relevant for animal welfare. Translational Animal Science, 4, 1070–1081.
- Thodberg K, Foldager L, Fogsgaard KK, Gaillard C and Herskin MS, 2022. Temperature conditions during commercial transportation of cull sows to slaughter. Computers and Electronics in Agriculture., 192, 106626.
- Thom EC, 1959. The discomfort index. Weatherwise, 12, 57–61.
- Torrey S, Bergeron R, Widowski T, Lewis N, Crowe T, Correa JA, Brown J, Gonyou HW and Faucitano L, 2013a. Transportation of market-weight pigs: I. Effect of season, truck type, and location within truck on behavior with a two-hour transport. Journal of Animal Science., 91, 2863–2871.
- Torrey S, Bergeron R, Faucitano L, Widowski T, Lewis N, Crowe T, Correa JA, Brown J, Hayne S and Gonyou HW, 2013b. Transportation of market-weight pigs: II. Effect of season and location within truck on behavior with an eight-hour transport. Journal of Animal Science, 91, 2872–2878.
- Turner AW and Hodgetts VE, 1959. The dynamic red cell storage function of the spleen in sheep I. Relationship to fluctuations of the jugular haematocrit. Australian Journal of Experimental Biology & Medical Science, 37.

Turner AI and Tilbrook AJ, 2006. Stress, cortisol and reproduction in female pigs. REPRODUCTION-CAMBRIDGE-SUPPLEMENT, 62, 191.

- Turner SP, Farnworth MJ, White IM, Brotherstone S, Mendl M, Knap P and Lawrence AB, 2006. The accumulation of skin lesions and their use as a predictor of individual aggressiveness in pigs. Applied Animal Behaviour Science, 96, 245–259.
- Urrea VM, Bridi AM, Camila Ceballos M, Paranhos da Costa MJR and Faucitano L, 2021. Behavior, blood stress indicators, skin lesions, and meat quality in pigs transported to slaughter at different loading densities. Journal of Animal Science, 99, 1–8.
- Valadez Noriega M and Miranda de la Lama GC, 2020. Implications, trends, and prospects for long-distance transport in cattle Review. Revista mexicana de ciencias pecuarias, 11, 517–538.
- Velarde A, Teixeira D, Devant M and Martí S, 2021. Research for ANIT Committee Particular welfare needs of unweaned animals and pregnant females, European Parliament, Policy Department for Structural and Cohesion Policies, Brussels.
- Visser K, 2014. Note on minimum space allowance and compartment height for cattle and pigs during transport. Wageningen UR Livestock Research. Report 764. ISSN 1570 – 8616.
- Vitali A, Lana E, Amadori M, Bernabucci U, Nardone A and Lacetera N, 2014. Analysis of factors associated with mortality of heavy slaughter pigs during transport and lairage. Journal of animal science, 92, 5134–5141.
- Vogel KD, Fabrega Rromans E, Llonch Obiols P and Velarde V, 2019. Stress physiology of animals during transport. Chapter, 3, 30–57.
- Von Borell EH, 2001. The biology of stress and its application to livestock housing and transportation assessment. Journal of Animal Science, 79(suppl_E), E260–E267.
- Walz PH, Taylor D and Pugh D, 2012. Fluid therapy and nutritional support. Sheep and Goats Medicine. 2nd edn. Saunders, Elsevier, Missouri. pp. 50–61.
- Warriss PD, Brown SN, Knowles TG, Edwards JE, Kettlewell PJ and Guise JH, 1998. The effect of stocking density in transit on carcass quality and welfare of slaughter pigs: 2. Results from the analysis of blood and meat samples. Meat Science., 50, 447–456.

- Warriss PD and Bevis EA, 1987. Liver glycogen in slaughtered pigs and estimated time of fasting before slaughter. British Veterinary Journal, 143, 354–360.
- Warriss PD and Brown SN, 1994. A survey of mortality in slaughter pigs during transport and lairage. The Veterinary Record, 134, 513–515.
- Warriss PD, Dudley CP and Brown SN, 1983. Reduction of carcass yield in transported pigs. Journal of the Science of Food and Agriculture, 34, 351–356.
- Warriss PD, Bevis EA, Edwards JE, Brown SN and Knowles TG, 1991. Effect of the angle of slope on the ease with which pigs negotiate loading ramps. Veterinary Record, 128, 419–421.
- Warriss PD, Brown SN, Barton Gade P, Santos C, Nanni Costa L, Lambooij E and Geers R, 1998. An analysis of data relating to pig carcass quality and indices of stress collected in the European union. Meat Science, 49, 137–144.
- Welfare Quality[®], 2009. Welfare Quality[®] assessment protocol for pigs (sows and piglets, growing and finishing pigs). Welfare Quality[®] Consortium. The Netherlands, Lelystad.
- Werner C, Reiners K and Wicke M, 2007. Short as well as long transport duration can affect the welfare of slaughter pigs. Animal Welfare, 16, 385–389.
- Weschenfelder AV, Torrey S, Devillers N, Crowe T, Bassols A, Saco Y and Faucitano L, 2012. Effects of trailer design on animal welfare parameters and carcass and meat quality of three Pietrain crosses being transported over a long distance. Journal of Animal Science, 90, 3220–3231.
- Weschenfelder AV, Saucier L, Maldague X, Rocha LM, Schaefer AL and Faucitano L, 2013. Use of infrared ocular thermography to assess physiological conditions of pigs prior to slaughter and predict pork quality variation. Meat Science, 95, 616–620.
- Wilhelmsson S, 2022. There's no time to rush!: pigs' and transport drivers' welfare and interactions during slaughter transport. Doctoral thesis. Sveriges lantbruksuniv. Acta Universitatis Agriculturae.
- Williams AM, Safranski TJ, Spiers DE, Eichen PA, Coate EA and Lucy MC, 2013. Effects of a controlled heat stress during late gestation, lactation, and after weaning on thermoregulation, metabolism, and reproduction of primiparous sows. Journal of Animal Science, 91, 2700–2714.
- WOAH (World Organisation for Animal Health), 2011. Terrestrial animal health code. Chapter 7.3. Transport of animals by land. Available online: https://www.WOAH.int/fileadmin/Home/eng/Health_standards/tahc/current/ chapitre_aw_land_transpt.pdf
- Xiong Y, Green A and Gates RS, 2015. Characteristics of trailer thermal environment during commercial swine transport managed under U.S. industry guidelines. Animals, 5, 226–244. https://doi.org/10.3390/ani5020226
- Yu J, Tang S, Bao E, Zhang M, Hao Q and Yue Z, 2009. The effect of transportation on the expression of heat shock proteins and meat quality of M. longissimus dorsi in pigs. Meat Science, 83, 474–478.
- Yundong Z, 2012. Pig fever [J]. Aquac. Technol. Consultant, 6, 172.
- Zappaterra M, Catillo G, Fiego DPL, Belmonte AM, Padalino B and Davoli R, 2022. Describing backfat and Semimembranosus muscle fatty acid variability in heavy pigs: analysis of non–genetic factors. Meat Science, 183, 108645.
- Zhao Y, Liu X, Mo D, Chen Q and Chen Y, 2015. Analysis of reasons for sow culling and seasonal effects on reproductive disorders in Southern China. Animal reproduction science, 159, 191–197.
- Zhao W, Liu F, Bell AW, Le HH, Cottrell JJ, Leury BJ, Green MP and Dunshea FR, 2020. Controlled elevated temperatures during early-mid gestation cause placental insufficiency and implications for fetal growth in pregnant pigs. Scientific Reports, 10, 20677. https://doi.org/10.1038/s41598-020-77647-1
- Zobel G, Leslie K, Weary DM and von Keyserlingk MA, 2013. Gradual cessation of milking reduces milk leakage and motivation to be milked in dairy cows at dry-off. Journal of Dairy Science, 96, 5064–5071.
- Zou B, Zhao D, He G, Nian Y, Da D, Yan J and Li C, 2020. Acetylation and phosphorylation of proteins affect energy metabolism and pork quality. Journal of Agricultural and Food Chemistry, 68, 7259–7268.
- Zurbrigg K, van Dreumel T, Rothschild M, Alves D, Friendship R and O'Sullivan T, 2017. Pig-level risk factors for intransit losses in swine: a review. Canadian Journal of Animal Science, 97, 339–346.

Abbreviations

- ABM animal-based measure
- AST aspartate aminotransferase
- bpm Breaths per minute
- CK creatine kinase
- EKE Expert Knowledge Ellicitation
- HR heart rate HRV heart rate variability LCT lower critical temperature
- PRE preventive measure
- RH relative humidity
- RR respiratory rate RT rectal temperature
- www.efsa.europa.eu/efsajournal



SAA	serum amyloid A
TCZ	thermal comfort zone
THI	temperature-humidity index
TNC	thermoneutral zone
TRACES	TRAde Control and Expert System
TRPB	transport-related problem behaviour
UCT	upper critical temperature
WC	welfare consequence



Appendix A – Template used during the selection of the highly relevant welfare consequences

Name Example Animal species Example Animal species Example Animal species Example Husbandery system Example Code 1 Clearly relevant 33 Not applicable VVV 2 to 31 Please rank the remaining WCs VVV ID Abbr Welfare Consequence Ranking Bindin B03 SCS Group stress 0 0 H03 HEA Heat stress 1 1 B02 RSP Resting problems 1 1 B04 GFS Moton Stress 1 1 B05 MNU Restriction of movement 1 1 B04 VAS Servery Overstimulation 1 1 B05 HNL Handling stress 1 1 B06 GSL Kull stress 1 1 B07 MOV Restriction of movement 1 1 B08 <th></th> <th></th> <th></th> <th></th> <th></th>					
Animal species Example Animal type Example Husbandery system Cearly relevant 32 Clearly not relevant 33 Not applicable 2 to 31 Please rank the remaining WCs VVV ID Abbr Welfare Consequence Ranking Bindin B03 SCS Group stress 0 1 H04 CLD Cold stress 1 1 B06 GFS Motion Stress 1 1 B07 RsP Resting problems 1 1 B06 GFS Motion Stress 1 1 B07 HNG Polonged hunger 1 1 B06 GFS Motion Stress 1 1 B07 HNL Handling stress 1 1 B08 GFS Inability to perform comfort behaviour 1 1 B07 HNL Handling stress 1 1 B08 SEL Skin lesions and wounds 2 2 B09 CMF Inability	me		Example		
Animal species Example Animal type Example Husbandery system Example Code 1 Clearly relevant 32 Clearly not relevant 33 Not applicable VVV 2 to 31 Please rank the remaining WCs VVV ID Abbr Welfare Consequence Ranking Bindin B03 SCS Group stress 0 0 H04 CLD Od stress 1 1 B06 GFS Meting problems 1 1 B06 GFS Meting problems 1 1 B07 HNQ Prolonged hunger 1 1 B08 MOV Restriction of movement 1 1 B04 VAS Sensory Overstimulation 1 1 B05 HNL Handling stress 1 1 B06 GSL Kin lesions and wounds 1 1 B05 HNL Handling stress 1 1 B06 INSL Skin lesions and wounds					
Animal type Example Husbandery system Image: Consequence Example Code 1 Clearly relevant Image: Consequence Ranking Bindin B03 SCS Group stress 0	Animal species		Example		
Husbandery system Image: Example Code 1 Clearly relevant 32 Clearly not relevant 33 Not applicable 2 to 31 Please rank the remaining WCs WV ID Abbr Welfare Consequence Ranking Bindin Bindin B03 SCS Group stress ID Abbr Welfare Consequence Ranking B03 SCS Group stress 0 H04 CLD Cold stress 1 B05 MEKA Hest stress 1 B06 GFS Motion Stress 1 B01 MOV testriction of movement 1 B04 VAS sensory Overstimulation 1 B05 HNL Handling stress 1 B09 CMF Inability to perform comfort behaviour 1 B12 EXP Inability to perform exploratory of foraging behaviour 1 H05 LOC Locomotory disord	imal type	e	Example		
Code 1 Clearly relevant 32 Clearly not relevant 33 Not applicable VVV 2 to 31 Please rank the remaining WCs VVV ID Abbr Welfare Consequence Ranking Bindin B03 SCS Group stress 0 0 H03 HEA Heat stress 1 1 B02 RSP Resting problems 1 1 B06 GFS Motion Stress 1 1 B01 MOV Restriction of movement 1 1 B04 VAS Sensory Overstmulation 1 1 B05 HNL Handling stress 1 1 B09 CMF Inability to perform exploratory or foraging behaviour 1 1 H02 L2O Prolonged thinst 1 1 H03 MOV Restriction of movement 1 1 B04 VAS Sensory Overstmulation 1 1 B05 L2D Inability to perform comfort behaviour <t< th=""><th>isbandery</th><th>y system</th><th>Example</th><th></th><th></th></t<>	isbandery	y system	Example		
Code 1 Clearly not relevant 32 Clearly not relevant 33 Not applicable 2 to 31 Please rank the remaining WCs WW				1	
32 Clearly not relevant 33 Not applicable 2 to 31 Please rank the remaining WCs VVV ID Abbr Welfare Consequence Ranking Bindin B03 SCS Group stress 0 0 H03 HEA Heat stress 1 1 B04 CLD Cold stress 1 1 B05 RSP Resting problems 1 1 B06 GFS Motion Stress 1 1 B01 MOV Restriction of movement 1 1 B04 VAS Sensory Overstimulation 1 1 B05 HNL Handling stress 1 1 B06 GFF Inability to perform comfort behaviour 1 1 B05 HNL Handling stress 1 1 B06 GFF Inability to perform comfort behaviour 1 1 B12 EXP Inability to perform exploratory or foraging behaviour 1 1 H02 H20 Prolonged thirst 1	Code	1	Clearly relevant		
33 Not applicable 2 to 31 Please rank the remaining WCs VVV ID Abbr Welfare Consequence Ranking Bindin B03 SCS Group stress 0 H03 HEA Heat stress 1 B04 CLD Cold stress 1 B05 GFS Motion Stress 1 B06 GFS Motion Stress 1 B01 MOV Restring problems 1 B04 VAS Sensory Overstimulation 1 B05 HNL Handling stress 1 B09 CMF Inability to perform comfort behaviour 1 B12 EXP Inability to perform comfort behaviour 1 H02 H20 Prolonged thirst 1 1 H05 LOC Locomotory disorders (including lameness) 1 1 H06 SKL Skin lesions and wounds 1 1 H10 RES Respiratory disorders 2 2 H11 MES Respiratory disorders 2	-	32	Clearly not relevant	-	
Z to 31 Please rank the remaining WCs WW ID Abbr Welfare Consequence Ranking Bindin B03 SCS Group stress 0 H03 HEA Heat stress 1 H04 CLD Cold stress 1 B06 GFS Motion Stress 1 H01 HNG Prolonged hunger 1 B04 VAS Sensory Overstimulation 1 B05 HNL Handling stress 1 B09 CMF Inability to perform comfort behaviour 1 B12 EXP Inability to perform exploratory or foraging behaviour 1 H02 H20 Prolonged thirst 1 1 H05 LOC Locomotory disorders (including lameness) 1 1 H06 SkL Skin lesions and wounds 1 1 H11 RES Respiratory disorders 1 1 H12 GED Gastro-enteric disorders 1 1 H11 RES Respiratory disorders 32 32	-	33	Not applicable		
IDAbbrWelfare ConsequenceRankingBindinB03SCSGroup stress0H03HEAHeat stress1H04CLDCold stress1B02RSPResting problems1B06GFSMotion Stress1H01HNGProlonged hunger1B01MOVRestriction of movement1B04VASSensory Overstimulation1B05HNLHandling stress2B09CMFInability to perform confort behaviour1H02H2QProlonged thirst1H05LOCLocomotory disorders (including lameness)1H06SKLSkin lesions and wounds1H10RESRespiratory disorders1H110RESRespiratory disorders1H12GEDGastro-enteric disorders1H147UMBUmbilical disorders32H17UMBUmbilical disorders32H07MANPain resulting from management procedures32H08BNLBone lesions (incl. fractures and dislocations)32H01EYEEye disorders and colorers32H03Skin disorders (other than pododermatitis or skin lesions)32H07IANPain resulting from management procedures32H07Skin disorders (other than pododermatitis or skin lesions)32H07ISOIsolation stress32B0		2 to 31	Please rank the remaining WCs	VVV	
IDAbbrWeitare ConsequenceRankingBindinB03SCSGroup stress0H03HEAHeat stress1H04CLDCold stress1B02RSPResting problems1B06GFSMotion Stress1H01HNGProlonged hunger1B01MOVRestriction of movement1B04VASSensory Overstimulation1B05HNLHandling stress1B09CMFInability to perform comfort behaviour1B12EXPInability to perform exploratory or foraging behaviour1H05LOCLocomotory disorders (including lameness)1H06SKLSkin lesions and wounds1H10RESRespiratory disorders1H11MUSMuscle disorders1H12GEDGastro-enteric disorders1H16MUSMuscle disorders32H17UMBUmbilical disorders32H07MANPain resulting from management procedures322H09SKDSkin disorders (other than pododermatitis or skin lesions)32H09SKDSkin disorders32B07ISOIsolation stress33B07ISOIsolation stress33					
B03SCSGroup stress0H03HEAHeat stress1H04CLDCold stress1B02RSPResting problems1B06GFSMotion Stress1H01HNGProlonged hunger1B01MOVRestriction of novement1B04VASSensory Overstimulation1B05HNLHandling stress1B09CMFInability to perform comfort behaviour1B12EXPInability to perform exploratory or foraging behaviour1H02H2OProlonged thirst1H05LOCLocomotory disorders (including lameness)1H06SKLSkin lesions and wounds1H110RESRespiratory disorders1H115MTBMetabolic disorders1H16MUSMuscle disorders32H17UMBUmbilical disorders32H16PinPain resulting from management procedures322H07MANPain resulting from management procedures322H09SKDSkin disorders (other than pododermatitis or skin lesions)32H11EYEEye disorders32H09SKDSkin disorders (other than pododermatitis or skin lesions)32H09SKDSkin disorders (other than pododermatitis or skin lesions)32H011EYEEye disorders32H03Sloation stress32	ID	Abbr	Welfare Consequence	Ranking	Binding
H03HEAHeat stress1H04CLDCold stress1B02RSPResting problems1B06GFSMotion Stress1H01HNGProlonged hunger1B01MOVRestriction of movement1B04VASSensory Overstimulation1B05HNLHandling stress1B09CMFInability to perform comfort behaviour1B12EXPInability to perform exploratory or foraging behaviour1H05LOCLocomotory disorders (including lameness)1H06SKLSkin lesions and wounds1H10RESRespiratory disorders1H11MUSMuscle disorders1H12GEDGastro-enteric disorders1H16MUSMuscle disorders32H17UMBUmbilical disorders32B08SEPSeparation stress32B16PLYInability to perform play behaviour32H07MANPain resulting from management procedures32H08BNLBone lesions (incl. fractures and dislocations)32H09SKDSkin disorders (other than pododermatitis or skin lesions)32B07ISOIsolation stress32B07ISOIsolation stress32	B03	SCS	Group stress	0	
H04CLDCold stress1B02RSPResting problems1B06GFSMotion Stress1H01HNGProlonged hunger1B01MOVRestriction of movement1B04VASSensory Overstimulation1B05HNLHandling stress1B09CMFInability to perform comfort behaviour1B12EXPInability to perform exploratory or foraging behaviour1H02H2OProlonged thirst1H05LOCLocomotory disorders (including lameness)1H06SKLSkin lesions and wounds1H11RESRespiratory disorders1H12GEDGastro-enteric disorders1H16MUSMuscle disorders32H17UMBUmbilical disorders32B16PLYInability to perform play behaviour32H08BNLBone lesions (incl. fractures and dislocations)32H09SKDSkin disorders (other than pododermatitis or skin lesions)32B07ISOIsolation stress33B07ISOIsolation stress33	HU3		Heat stress	1	
B02RSPResting problems1B06GFSMotion Stress1H01HNGProlonged hunger1B01MOVRestriction of movement1B04VASSensory Overstimulation1B05HNLHandling stress1B09CMFInability to perform comfort behaviour1B12EXPInability to perform exploratory or foraging behaviour1H02H2OProlonged thirst1H05LOCLocomotory disorders (including lameness)1H06SKLSkin lesions and wounds1H10RESRespiratory disorders1H11RESRespiratory disorders1H12GEDGastro-enteric disorders1H14MUSMuscle disorders1H15MTBMetabolic disorders1H16MUSMuscle disorders32B08SEPSeparation stress32B16PLYInability to perform play behaviour32H07MANPain resulting from management procedures32H08BNLBone lesions (incl. fractures and dislocations)32H09SKDSkin disorders (other than pododermatitis or skin lesions)32B07ISOIsolation stress33B07ISOIsolation stress33	H04		Cold stress	1	
B00GrSInitial Stress1H01HNGProlonged hunger1B01MOVRestriction of movement1B04VASSensory Overstmulation1B05HNLHandling stress1B09CMFInability to perform comfort behaviour1B12EXPInability to perform exploratory or foraging behaviour1H02H2OProlonged thirst1H05LOCLocomotory disorders (including lameness)1H06SKLSkin lesions and wounds1H10RESRespiratory disorders1H11GEDGastro-enteric disorders1H12GEDGastro-enteric disorders1H14MUSMuscle disorders1H17UMBUmbilical disorders32B08SEPSeparation stress32B16PLYInability to perform play behaviour32H07MANPain resulting from management procedures32H08BNLBone lesions (incl. fractures and dislocations)32H09SKDSkin disorders (other than pododermatitis or skin lesions)32H01EYEEye disorders32B07ISOIsolation stress33	BUZ	KSP CTS	Resting problems	1	
H01HNGFrobinged hunger1B01MOVRestriction of movement1B04VASSensory Overstimulation1B05HNLHandling stress1B09CMFInability to perform comfort behaviour1B12EXPInability to perform exploratory or foraging behaviour1H02H2OProlonged thirst1H05LOCLocomotory disorders (including lameness)1H06SKLSkin lesions and wounds1H10RESRespiratory disorders1H112GEDGastro-enteric disorders1H15MTBMetabolic disorders1H16MUSMuscle disorders2H17UMBUmbilical disorders32B08SEPSeparation stress32B16PLYInability to perform play behaviour32H07MANPain resulting from management procedures32H08BNLBone lesions (incl. fractures and dislocations)32H09SKDSkin disorders (other than pododermatitis or skin lesions)32B07ISOIsolation stress33	BU0		Motion Stress	1	
B01MOVNestriction of inverting1B04VASSensory Overstimulation1B05HNLHandling stress1B09CMFInability to perform comfort behaviour1B12EXPInability to perform exploratory or foraging behaviour1H02H2OProlonged thirst1H05LOCLocomotory disorders (including lameness)1H06SKLSkin lesions and wounds1H10RESRespiratory disorders1H12GEDGastro-enteric disorders1H16MUSMuscle disorders1H17UMBUmbilical disorders32B16PLYInability to perform play behaviour32H07MANPain resulting from management procedures32H08BNLBone lesions (incl. fractures and dislocations)32H09SKDSkin disorders (other than pododermatitis or skin lesions)32B07ISOIsolation stress33B07ISOIsolation stress33	HU1	HING	Prolonged nunger	1	
B04VASDensitive Overstminutation1B05HNLHandling stressB09CMFInability to perform comfort behaviourB12EXPInability to perform exploratory or foraging behaviourH02H20Prolonged thirst1H05LOCLocomotory disorders (including lameness)1H06SKLSkin lesions and woundsH10RESRespiratory disordersH112GEDGastro-enteric disordersH15MTBMetabolic disordersH16MUSMuscle disordersH17UMBUmbilical disorders32B08SEPSeparation stress32B16PLYInability to perform play behaviour32H07MANPain resulting from management procedures32H08BNLBone lesions (incl. fractures and dislocations)32H11EYEEye disorders32B07ISOIsolation stress33B07ISOIsolation stress33	BU1		Restriction of movement	1	
B03HNLHandling suessB09CMFInability to perform comfort behaviourB12EXPInability to perform exploratory or foraging behaviourH02H20Prolonged thirst1H05LOCLocomotory disorders (including lameness)H06SKLSkin lesions and woundsH10RESRespiratory disordersH11GEDGastro-enteric disordersH12GEDGastro-enteric disordersH15MTBMetabolic disordersH16MUSMuscle disordersH17UMBUmbilical disordersB08SEPSeparation stressB16PLYInability to perform play behaviourH07MANPain resulting from management proceduresH08BNLBone lesions (incl. fractures and dislocations)H09SKDSkin disordersH11EYEEye disordersB07ISOIsolation stressB07Isolation stressB07Isolation stressB07Isolation stressB07Isolation stressB07Isolation stressB07Isolation stressB07Isolation stress	DU4		Handling stress	1	
B03CivinInability to perform exploratory or foraging behaviourH02H20Prolonged thirst1H05LOCLocomotory disorders (including lameness)1H06SKLSkin lesions and wounds1H10RESRespiratory disorders1H12GEDGastro-enteric disorders1H15MTBMetabolic disorders1H16MUSMuscle disorders1H17UMBUmbilical disorders32B16PLYInability to perform play behaviour32H07MANPain resulting from management procedures32H08BNLBone lesions (incl. fractures and dislocations)32H11EYEEye disorders32B07ISOIsolation stress32B07ISOIsolation stress32B07ISOIsolation stress33B04MCVIsolation stress33	BOO		Handling scress		
D12EXPIntability to perform exploratory of roraging behaviourH02H20Prolonged thirst1H05LOCLocomotory disorders (including lameness)1H06SKLSkin lesions and wounds1H10RESRespiratory disorders1H12GEDGastro-enteric disorders1H15MTBMetabolic disorders1H16MUSMuscle disorders1H17UMBUmbilical disorders32B08SEPSeparation stress32B16PLYInability to perform play behaviour32H07MANPain resulting from management procedures32H08BNLBone lesions (incl. fractures and dislocations)32H11EYEEye disorders32B07ISOIsolation stress32B07ISOIsolation stress32	BU3	EVD	Inability to perform exploratory or foraging behaviour		
HOZHOXHOXIH05LOCLocomotory disorders (including lameness)	P12		Prolonged thirst	1	
H05ECCEccentratory disorders (including fameliess)H06SKLSkin lesions and woundsH10RESRespiratory disordersH12GEDGastro-enteric disordersH15MTBMetabolic disordersH16MUSMuscle disordersH17UMBUmbilical disordersB08SEPSeparation stressB16PLYInability to perform play behaviourH07MANPain resulting from management proceduresH08BNLBone lesions (incl. fractures and dislocations)H11EYEEye disordersH11EYEEye disordersB07ISOIsolation stressB07ISOIsolation stressB07Isolation stress33	H02 H05	100	Locomotory disorders (including lameness)	1	
HistorSiteSiteSiteSiteSiteH10RESRespiratory disordersH12GEDGastro-enteric disordersH15MTBMetabolic disordersH16MUSMuscle disordersH17UMBUmbilical disordersB08SEPSeparation stress32B16PLYInability to perform play behaviour32H07MANPain resulting from management procedures32H08BNLBone lesions (incl. fractures and dislocations)32H09SKDSkin disorders (other than pododermatitis or skin lesions)32H11EYEEye disorders32B07ISOIsolation stress33	H06	SKI	Skin Jesions and wounds		
H10H25H25H25H25H25H12GEDGastro-enteric disordersH15MTBMetabolic disordersH16MUSMuscle disordersH17UMBUmbilical disordersB08SEPSeparation stress32B16PLYInability to perform play behaviour32H07MANPain resulting from management procedures32H08BNLBone lesions (incl. fractures and dislocations)32H09SKDSkin disorders (other than pododermatitis or skin lesions)32H11EYEEye disorders32B07ISOIsolation stress33	H10	RES	Respiratory disorders		
H12GLDGuide Ciffere abordersH15MTBMetabolic disordersH16MUSMuscle disordersH17UMBUmbilical disordersB08SEPSeparation stressB08SEPSeparation stressB16PLYInability to perform play behaviour32H07MANPain resulting from management procedures32H08BNLBone lesions (incl. fractures and dislocations)32H09SKDSkin disorders (other than pododermatitis or skin lesions)32H11EYEEye disorders32B07Isolation stress33	H12	GED	Gastro-enteric disorders		C
H12H12H12H12H12H13MUSMuscle disordersH17UMBUmbilical disordersB08SEPSeparation stress32B16PLYInability to perform play behaviour32H07MANPain resulting from management procedures32H08BNLBone lesions (incl. fractures and dislocations)32H09SKDSkin disorders (other than pododermatitis or skin lesions)32H11EYEEye disorders32B07ISOIsolation stress33	H15	MTB	Metabolic disorders		0
H17UMBUmbilical disordersB08SEPSeparation stress32B16PLYInability to perform play behaviour32H07MANPain resulting from management procedures32H08BNLBone lesions (incl. fractures and dislocations)32H09SKDSkin disorders (other than pododermatitis or skin lesions)32H11EYEEye disorders32B07ISOIsolation stress33	H16	MUS	Muscle disorders		
BO8SEPSeparation stress32B08SEPSeparation stress32B16PLYInability to perform play behaviour32H07MANPain resulting from management procedures32H08BNLBone lesions (incl. fractures and dislocations)32H09SKDSkin disorders (other than pododermatitis or skin lesions)32H11EYEEye disorders32B07ISOIsolation stress33	H17	UMB	Umbilical disorders		0
B16PLYInability to perform play behaviour32H07MANPain resulting from management procedures32H08BNLBone lesions (incl. fractures and dislocations)32H09SKDSkin disorders (other than pododermatitis or skin lesions)32H11EYEEye disorders32B07ISOIsolation stress33	B08	SEP	Separation stress	32	0
H07MANPain resulting from management procedures32H08BNLBone lesions (incl. fractures and dislocations)32H09SKDSkin disorders (other than pododermatitis or skin lesions)32H11EYEEye disorders32B07ISOIsolation stress33	B16	PLY	Inability to perform play behaviour	32	C
H08BNLBone lesions (incl. fractures and dislocations)32H09SKDSkin disorders (other than pododermatitis or skin lesions)32H11EYEEye disorders32B07ISOIsolation stress33	H07	MAN	Pain resulting from management procedures	32	C
H09SKDSkin disorders (other than pododermatitis or skin lesions)32H11EYEEye disorders32B07ISOIsolation stress33D10MOXCompare the leader32	H08	BNL	Bone lesions (incl. fractures and dislocations)	32	C
H11 EYE Eye disorders 32 B07 ISO Isolation stress 33 P10 WOY Isolation stress 33	H09	SKD	Skin disorders (other than pododermatitis or skin lesions)	32	C
B07 ISO Isolation stress 33	H11	EYE	Eye disorders	32	C
	B07	ISO	Isolation stress	33	C
BID WSX Inability to perform sexual behaviour 33	B10	WSX	Inability to perform sexual behaviour	33	C
B11 USX Inability to avoid unwanted sexual behaviour 33	B11	USX	Inability to avoid unwanted sexual behaviour	33	C
B13 MAT Inability to express maternal behaviour 33	B13	MAT	Inability to express maternal behaviour	33	C
B14 SUC Inability to perform sucking behaviour 33	B14	SUC	Inability to perform sucking behaviour	33	C
B15 CHW Inability to chew and ruminate 33	B15	CHW	Inability to chew and ruminate	33	C
H13 RPD Reproductive disorders 33	H13	RPD	Reproductive disorders	33	C
H14 MAS Mastitis 33	H14	MAS	Mastitis	33	C

Please sort this table by the "Ranking" and complete the "Binding" (No. of cells with same rank)